The international tax law of Controlled Foreign Corporation rules and their influence on multinational companies' behaviour

Dissertation zur Erlangung des Doktorgrades der Wirtschafts- und Sozialwissenschaftlichen Fakultät der Eberhard Karls Universität Tübingen

> vorgelegt von Axel Prettl

> > Tübingen

2018

- 1. Betreuer:
- 2. Betreuer:

Prof. Dr. Martin Ruf Prof. Dr. Georg Wamser

Tag der mündlichen Prüfung: Dekan:

- 1. Gutachter:
- 2. Gutachter:
- 3. Gutachter:

06.06.2019 Prof. Dr. Josef Schmid Prof. Dr. Martin Ruf Prof. Dr. Georg Wamser Prof. Dr. Joachim Grammig

The international tax law of Controlled Foreign Corporation rules and their influence on Multinational companies' behaviour

A Dissertation Presented to the Faculty of Economics and Social Sciences of the University of Tübingen in Candidacy for the Doctoral Degree

> By Axel Maximilian Prettl from Düsseldorf

FIRST ADVISER: PROF. DR. MARTIN RUF SECOND ADVISOR: PROF. DR. GEORG WAMSER

Tübingen 2018

Acknowledgements

First and foremost, I would like to express my gratitude to my supervisors Martin Ruf and Georg Wamser for the guidance, great academic liberty, and unconditional support throughout my work on this dissertation. I thank Wilhelm Kohler and Frank Stähler for their kind willingness to take the chair in my examination committee.

I have learned a lot and truly enjoyed working with my friend and co-author Dominik von Hagen, who has especially contributed to Chapter 4 of this dissertation. Special thanks goes to my colleagues Jonathan Eklund, Peter Eppinger, Thomas Hettig, Sandra Kohler (!), Bohdan Kukharskyy, Michael Riedle, Elias Steinmüller, Susanne Wellmann and further friends in Tübingen, who have not only helped me with advice but where also up to non-econ or -business administration conversations.

I gratefully acknowledge the hospitality of the economics departments at the University of Michigan in Ann Arbor, the University of California in Berkeley and the University of Oxford, where parts of my work on this dissertation were conducted. In particular, I would like to thank Joel Slemrod, Alan Auerbach and Michael Devereux, who have made these research visits possible as my supervisors abroad. My work has also benefited enormously from the academic exchange with numerous colleagues who shared their knowledge and insights with me at seminars and conferences. Especially, I want to mention Daniel Reck and Martin Simmler, who became true friends over the past couple of years. But also Jenny Mayo, Ben Meiselmann, Daniela Morar, Max Risch, Daniel Schaffa, Enda Hargaden and others from my research stay in Ann Arbor, David Birke, Johannes Hermle, Maria Nowak and others from my research stay in Oxford, whom I had the pleasure of working, exchanging ideas and enjoying life with over the past years. Great thanks guys!

This dissertation was partly possible due to the financial support by the German Research Foundation (DFG), which enabled the Research School of International Taxation and the "Mini-Graduiertenkolleg", which I had the pleasure to be a part of.

I would not have been able to complete this dissertation without the strong support of my family during the hard time we all had to go through due to the unexpected loss of my mother. I am deeply grateful for my parents, siblings and all my friends for cheering me up and supporting my unconventional way of life.

To my family, friends, and the public.

Contents

	Ack	nowledg	gements		ii
1	Intr	oducti	ion		1
2	Wh	What are CFC rules and how do they work?			7
	2.1	Funda	mental fu	unctionality and the importance of CFC rules	8
	2.2	Expla	nation an	d comparison of CFC rule characteristics and requirements .	13
	2.3	Litera	ture on C	CFC rules	21
	2.4	CFC 1	ules per	country	24
3	Profit Shifting & CFC Rules			CFC Rules	53
	3.1	1 Introduction			
	3.2	2 Literature on profit shifting			60
	3.3	Conceptual framework			64
	3.4	 Empirical methodology A note on data and data sources Estimation results - Impact of CFC rule characteristics on profit shifting 			71
	3.5				78
	3.6				80
	3.6.1 How CFC rules limit profit shifting behaviour of multinationals		$^{\circ}\mathrm{C}$ rules limit profit shifting behaviour of multinationals $~$	81	
		3.6.2	How CF	°C rules are circumvented	99
			3.6.2.1	Graphical evidence for the influence of the EEA exemption .	100
			3.6.2.2	Impact of the passive-to-total income threshold on financial	
				income abroad	102

			3.6.2.3 Amount of shifted profits abroad)8
	3.7	Conclu	sion \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots 10)9
4	Mu	ltinatio	nal ownership structures and anti tax avoidance legislation 11	.2
	4.1	Introd	uction	13
	4.2	Litera	ure	18
	4.3	CFC r	$les and the acquisition of low-tax targets \ldots \ldots \ldots \ldots \ldots \ldots \ldots 12$	20
		4.3.1	Hypothesis development	20
		4.3.2	Empirical approach	21
			4.3.2.1 Acquirer perspective	21
			4.3.2.2 Target Perspective	27
		4.3.3	Data	28
		4.3.4	Results	30
			4.3.4.1 Graphical analysis	30
			4.3.4.2 Acquirer perspective	33
			4.3.4.3 Target perspective	11
			4.3.4.4 Comparison and further robustness of both perspectives 14	18
	4.4	CFC r	les and the direction of cross-border M&A $\ldots \ldots \ldots$	51
		4.4.1	Hypothesis development	51
		4.4.2	Empirical approach	51
		4.4.3	Data	55
		4.4.4	Results	55
	4.5	Conclu	$sion \ldots 15$	58
A	CFO	C rule	characteristics and changes per country 16	60
в	CFO	C rules	and Profit Shifting 16	;3
	B.1	3.1 Mathematical Proofs		
	B.2 Further bunching methodology elaboration			34

	B.3	Descriptives and more robustness checks	166		
	B.4	Further bunching insights	170		
С	CFC	C rules and M&A	176		
	C.1	Stylized identification variable example $\ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots$	176		
	C.2	Descriptives and robustness check tables on low-tax target acquisition	178		
	C.3	Descriptives tables on the direction of M&As	188		
Bi	Bibliography 19				

Chapter 1

Introduction

Most companies maximize their profits. Governments around the world, usually, tax these firm profits for various reasons. To maximize after tax profits, however, firms often try to circumvent this taxation by various means. The economic and juridical research presented in this dissertation is motivated by major empirical observations about international businesses and their investment behaviour affected by corporate taxation in the last decades.

As any other non-lump sum taxation, anti-tax avoidance laws create distortions and are market-interfering or a constraint for the reallocation of e.g. capital.¹ From a firm's perspective, international investment decisions can be affected by laws such as the three main anti-tax avoidance laws used globally: thin capitalization rules, transfer pricing rules and controlled foreign corporation (CFC) rules. Therefore, on the one hand, these laws could affect the amount of profits or specific capital invested and used in the headquarters country as well as in the countries of the foreign subsidiaries². On the other hand, from a country's perspective, such anti-tax avoidance laws are usually implemented to prevent profit shifting on the part of multinational companies (multinationals) and thus tax base erosion to secure tax revenue and equal treatment of firms. Thereto, multinationals are at best taxed the same as domestic firms to ensure capital export neutrality. However, it is

¹ e.g., Gordon (1986); Slemrod (1990); Nicodème (2009)

 $^{^{2}}$ e.g., Hines (1999); Huizinga, Laeven, and Nicodeme (2008); Fuest et al. (2013)

often not easily detectable whether or not investments are mostly taken for tax-driven profit shifting purposes, especially in an international context.

CFC rules - the laws of interest in this study - are an important group of these tax avoidance counter measures and, so far, only little is known about them. In general, CFC rules attempt to tax earnings of foreign subsidiaries, repatriated or not, under specific circumstances concerning both the headquarters and the subsidiary abroad, in case these earnings are shifted profits. These laws could become a potential alternative or rather a powerful addition to tackle raising profit shifting by multinationals. But are these legislations effective in counteracting the profit shifting behaviour of multinationals? To answer this question, and to better understand CFC rules on a global scale, this study sheds some light on these legislations in three ways: (1) The CFC laws of 27 countries are depicted and compared in detail in Chapter 2, (2), the influence of these laws on the profit shifting behaviour of multinationals is shown in Chapter 3.1, and (3), the effect of CFC rules on the ownership structure of multinationals is researched by analyzing cross-border M&A activity in Chapter 4. This introduction will continue with outlining the broader goal of my research, then I am highlighting my mutual theoretical and empirical base and used methodology, before providing a brief overview of each chapter, I conclude by noting implications for future work.

The overall goal of my research, initiated with this dissertation, is to contribute to a better understanding of the profit shifting and investment behaviour of multinationals in relation to governmental means of taxing economic performance in an increasingly globally connected business environment. Thereto, I aim at combining the interrelated research areas of economics, law and business administration. Designing a just and efficient corporate income tax system has become increasingly important in light of tightening governmental fiscal constraints and rising inequality. Recent books and studies have illustrated and proven the commonly felt, by academics and non-academics, increase of wealth and income inequality.³ Usually, businesses are ultimately held by individuals. Therefore, the taxation of corpo-

³ e.g., Piketty (2014); Zucman (2015); Alstadsæter, Johannesen, and Zucman (2019)

rations, especially in the international context, has a direct impact on the phenomenon of inequality and redistribution. If some corporations, or rather their owners, are capable of avoiding taxes while others are not, national as well as worldwide distortions are growing. Additionally, with regard to redistribution, the (non-) collected taxes can create a huge difference on the fiscal expenditure side as well. As our current world is developing with constantly increasing speed, accompanied by various concepts of taxing multinationals, the main open question is not how this development can be stopped, but how tax authorities can eventually tackle increasing issues like base erosion and profit shifting from high- to low-tax countries. With my research and the results of this dissertation, I provide deep insights on anti-tax avoidance laws and corporate tax avoidance behaviour on a global scale, which hopefully helps policy makers to gain a more solid information base for their decisions.

The underlying data base used in this dissertation is self-collected and comprises data about CFC rules for 61 countries, 27 of which do have these laws. For these 27 countries, I compiled a unique list of 18 distinct but comparable characteristics of CFC rules over a time frame of 11 years from 2004 to 2014. I combined this data set with further balance sheet data of multinational enterprises, provided by the data set ORBIS from Bureau van Dijk, to detect behavioural changes, provoked by the anti-tax avoidance laws. To ascertain these influences, I mostly used a mixture of quantitative, econometrical methods with various modifications: Simple comparison techniques, graphical analyses like event studies and distribution graphs, (fixed effect) ordinary least squared regressions, bunching observations, and (mixed) logit regressions. As all three main chapters of the work at hand are related to the CFC rule data set, the research found in this dissertation is not arranged in a standard cumulative dissertation design. Thus, Chapter 2, is rather written as a descriptive law part than an economic study and describes the data in great detail. The following chapters refer to these descriptions and explanations to prevent unnecessary repetition.

The single-authored Chapter 2, entitled *What are CFC rules and how do they work?*, provides a broad overview of the development and functionality of CFC rules worldwide,

based on the mentioned data set. First, the fundamental function and importance of CFC rules is described, and a brief historical overview is given. An explanation and comparison of the various law components follows in the next section. All important characteristics which are comparable and yet not too country-specific are compared across countries and over time. After the reader is familiar with the compositional details of these laws, a compilation of previous studies about CFC rules is provided to get an even better perspective on what is already known about this anti-tax avoidance legislation. The last section of this chapter shows the very details of the CFC rules as well as their development over time for every single one of the 27 countries.

Chapter 3, entitled Profit Shifting & Controlled Foreign Corporation Rules, is singleauthored and provides theoretical as well as empirical evidence of the influence of CFC rules on the profit shifting behaviour of multinational companies. So far, no published study exists, which shows the impacts of CFC rules based on such a rich worldwide panel data set on these laws. The literature part of this chapter focuses on profit shifting studies – within this dissertation – to limit repetition. A simple theory part of how multinationals' profit shifting behaviour could be influenced by CFC rules lays the groundwork for the further empirical analyses. Based on these theoretical considerations, the remainder of this chapter is split into two parts to examine two main components of CFC laws: a minimum low-tax threshold and a passive-to-total income threshold, which are part of many CFC laws. The mentioned CFC rule data set is combined with the ORBIS data base, which contains worldwide firm statistics. I estimate the response of financial profits abroad to the low-tax threshold of CFC rules with linear regression models in the first part. I find that CFC rules worldwide lead to a reduction of roughly 19% in financial income in or profit shifting into foreign low-tax subsidiaries. Then I provide graphical evidence of a court decision for European multinationals. The last part looks into bunching of foreign subsidiaries around the second threshold. This depicts specific tax avoidance tactics of multinationals to circumvent CFC rule taxation - at least to some extend.

An arising question is whether one can observe similar patterns of CFC rule influence on foreign direct investment decisions on an even broader scale. This question of deterring effects of the researched tax laws is answered in the following chapter in terms of cross-border mergers and acquisitions (M&A). Chapter 4, entitled Multinational Ownership Structures and Anti Tax Avoidance Legislation, includes joint work with Dominik von Hagen. In 2016, the value of cross-border M&A slightly exceeded the value of announced greenfield projects (UNCTAD (2017)), which emphasizes the global importance of M&As. It is known that tax laws in general influence M&A decisions, but we do not know much about the influence of CFC rules as an anti-tax avoidance measure so far. This chapter partly builds on former work by Feld, Ruf, et al. (2016); Arulampalam, Devereux, and Liberini (2019); Voget (2011) and uses, additionally to the CFC rule data set, the Thomson Financial SDC database, which contains worldwide M&A transactions. Our empirical analysis sheds light on the behaviour of acquiring foreign targets from two perspectives: Do CFC rules influence, (1) where the acquirer is located, and (2), where the target is located. In a third reassuring approach, we research the general impact of CFC rules on the direction of M&A. The result section begins with first descriptive graphical evidence. Then, we show that the probability of an acquirer being from a particular low-tax country is sensitive to the magnitude of potential taxes under CFC rules. Generally, our results seem to be larger and more robust with regard to the impact of CFC rules for the target location decision than for the question where the acquirer is located. Thus, these results point in the direction that the initial idea of CFC rules, to prevent multinational profit shifting behaviour, is indeed more affected than interfering real economic investments abroad. Third, we show that CFC rules alter the acquirer's choice of the targets' locations. Fourth, for a given pair of merging firms, the identity of the acquirer versus the target is sensitive to the existence and potential magnitude of taxes under CFC rules. Altogether, our study shows that for affected acquirer countries, CFC rules lead to less M&A activity in low-tax countries due to a reduced ability to shift income. The magnitudes of the effects we find are rather low, for example, a 10 percentage point increase in additional CFC rule taxation leads to a 0.5% lower likelihood that an acquisition takes place. This rather contradicts common lobby claims.

This dissertation analyzes the specifications and implications of CFC rules around the globe in great detail. Especially the influences of these laws on multinational firm behaviour. My findings depict the various CFC rule legislations and indicate that, in general, these antitax avoidance laws are capable of curbing profit shifting of multinationals globally. On the other hand, I provide findings on how these laws are circumvented to at least some extend. A next interesting step would be joint research about the three major anti-tax avoidance laws, potentially even include IP box regimes, or find ways to quantify law strength and make them comparable, to provide policy makers with even better tools for their work to ultimately curb the BEPS attitude of multinationals. Nonetheless, I hope that the novel insights and newly provided data sets contributed by this dissertation, will help policy makers and legislators in various countries to gain a deeper understanding of CFC rules and the influences of these laws on multinationals' behaviour to develop more suitable laws and curb profit shifting of multinationals for the influences of these laws on multinationals in the develop more suitable laws and curb profit shifting of multinationals for the set of these laws and curb profit shifting of multinationals is fliciently.

Chapter 2

What are CFC rules and how do they work?

International tax laws for corporations are quite complex in general and on top CFC rules are quite diverse between countries. So far, existing studies have researched one specific CFC rule or compared two countries' legislations only, except for one economic study by Voget (2011) about headquarters effects of CFC rules for a cross-country comparison for one year. Therefore, this chapter explains the diverse CFC legislations and compares CFC laws from 27 different countries worldwide, concerning their various characteristics over time. In Section 2.1, the fundamental functionality of CFC laws is explained. Section 2.2 gives a much more detailed description and comparison of the various characteristics of CFC laws, which is followed by an overview of published literature on CFC rules in Section 2.3. Lastly, Section 2.4 gives a detailed description of the CFC rule development for all 27 countries, which are researched in the dissertation at hand.

2.1 Fundamental functionality and the importance of CFC rules

New and old growing multinational company networks relocate activities across countries. Due to inter-company transactions in these networks across country borders, different laws and tax systems are strained. For smart firms this globally interconnected reality provides opportunities for legal tax planning.¹ After offshoring (e.g., Blinder (2006)), a further step in the profitability process is becoming more and more attractive, especially for very large corporations: Tax savings. However, these techniques are usually less complicated, as neither the full production process nor vast amounts of tangible assets have to be shifted to low-tax countries. Instead only the official income has to be shifted to benefit from tax incentives or lower effective income tax rates abroad (e.g., Huizinga and Laeven (2008)). The ways to do so are manifold whereas laws to counteract profit shifting behaviors are scarce (e.g., Gravelle (2009)). The line between fraud and technically legal operations are slim and for the public moral plays often a role (e.g.,

As countries observed that companies exploited these tax planning activities and tax avoidance became a large-scale problem, new laws were installed and old ones were adjusted. This process is still continuing.² Not only due to interest reasons can it be very profitable for multinationals to redistribute this income back to the mother country not immediately but later in time. However, at this point CFC rules are coming into action: CFC rules apply under specific conditions, explained in detail in this chapter, to at least some parts of a subsidiary's profits whether they are repatriated or not. In addition, CFC rules apply to income which failed to get taxed, for example, due to problems with arms-length pricing or other internal dept or profit shifting issues. CFC rules affect the whole firm structure,

¹ Not illegal tax evasion; Schäfer and Spengel (2010); Slemrod (2007); McGee (2012)

² For instance "thin capitalization rules" should regulate the dept-equity-ratio of multinationals or "transfer pricing regulations" try to regulate the intra-company prices, which both could be used for income shifting and therefore tax reductions. Firms not only use these but other methods and let the income arise in lowtax countries, for example by using financing methods, special services, royalty payments, or other capital movement methods in foreign subsidiaries to shelter income.

not only specific profit shifting methods. Therefore, and if they can accomplish what they are enacted for, CFC rules should be part of a country's tax legislation to counteract base erosion by profit and internal dept shifting of multinationals. However, implementing them is rather problematic.³ The first worldwide CFC rules were implemented after the Second World War instigated by the rising internationalization of companies and their evolution into multinationals. However, only in the last 20 years, CFC rules have become particularly important given the focus of multinationals to avoid high tax rates in their own home countries and how comparably easy it is nowadays to shift profits.⁴



Notes: Own source. This graph shows the increasing implementation of CFC rules in OECD countries.

Generally, CFC rules are receiving more and more attention. Many countries worldwide have implemented or strengthened them in recent years or plan to do so. One of the first multi-country attempts started by the OECD Commission in 1998 finally resulted in one of the main action points in the BEPS plan of the OECD (OECD (2013)). Additionally, the EU Commission recently started the Anti Tax Avoidance Directive (ATAD) which contains five legally-binding anti-abuse measures amongst others, CFC rules, which all EU Member

³ Ruf and Schindler (2015); Haufler, Mardan, and Schindler (2018); Kane (2013); Rust (2008); OECD (2015a); Weichenrieder (1996)

⁴ e.g. Dharmapala (2014) with an overview; Krautheim and Schmidt-Eisenlohr (2011); Huizinga and Laeven (2008); Redmiles and Wenrich (2007) for the USA p. 133; Schneider (2004)

States should apply against common forms of aggressive tax planning as of January 1st 2019.⁵ CFC rules are composed differently across countries even though they aim at the same goal: to prevent profit shifting in the first place. The various differences shown in this chapter point out the complexity of CFC laws and that, for broad comparison, it is not sufficient to focus on one aspect of these anti-tax avoidance legislations only.

In this chapter, explanations and descriptions of the various key requirements of CFC rules and their development are given. 6

One general problem of researching the influence of CFC taxation could be the existence of tax treaties between countries.⁷ If two countries have signed such a tax treaty, the taxation of the foreign country is influenced by that fact in most of the cases. CFC laws are usually enforced anyway⁸, and there are only a few examples where the CFC rule concerned subsidiaries' are explicitly excluded from tax treatments. Therefore, the following country comparison with the focus on the design of CFC rules does not take tax treaties into account. The banking and finance sector has done a good job in the last couple of decades to get special treatment within laws and now has the same stance in many CFC rules. Even though firms in this sector are also affected by different taxation and try to avoid higher taxes, they are often excluded from CFC rules and, therefore, are excluded in this study.⁹

From a comparative law perspective, one can generally distinguish between two main CFC treatment requirements, usually derived from historical factors of tax systems: First, concerning the country *where* the subsidiary and the profits are located - *jurisdictional approach* - and, second, regarding aspects of the subsidiary itself and specifically, the raise

⁵ Building on Directive (EU) $2016/\overline{1164}$.

⁶ In the following with "interests in a foreign entity" all possible interests like shares, voting rights, economic interests, influence trough specific persons, and more are meant. Unfortunately, there was no possibility to extract the detailed specification concerning this issue for all observed countries over time from the data sources and compare them on a decent level. Additionally, the later on used micro-level data does not offer detailed differentiation on that aspect.

 $^{^{7}}$ e.g., Oguttu (2009); Marques and Pinho (2014).

 $^{^{8}}$ More details about this topic has Kuźniacki (2015).

 $^{^{9}}$ e.g., Merz, Overesch, and Wamser (2017) have shown location choices of financial sector FDI due to tax and regulation policy or Andries, Gallemore, and Jacob (2017) the influence of corporate taxation on bank transparency.

or shift of *passive income abroad - transactional approach*. The rationale of the first system (jurisdictional approach) is to include income derived by foreign entities located in low- or non-taxation jurisdictions into taxation at the parent level to avoid the artificial transfer of income to tax havens or generally lower-taxed countries corporate-wise. While this approach is mainly an anti-abusive rule, the transactional approach has an anti-tax-deferral objective, aiming to achieve capital export neutrality. However, there is no pure form of just one approach in CFC laws anymore.

Therefore, it's not sufficient to focus on one of these characteristics alone if one wants to shed light on the world of different CFC rule settings: all characteristics and thresholds influence the behavior of multinationals concerning their profit shifting strategies substantially and in different ways. Coming from the law perspective, the two main important requirements, besides the control requirement, are the minimum tax threshold and the passive income threshold which are described below as well as in more detail in the following section.

Minimum Tax threshold

This is the most common form of the jurisdictional approach: Whether a foreign country is considered as a low-tax country is determined by a so-called minimum low-tax threshold. In many of the observed CFC rules, such a minimum tax threshold is specified. If the foreign (effective) corporate income tax rate is above this threshold, the country is counted as a safe harbor and not as a low-tax country. Therefore, the CFC law taxation consequences do not apply and vice versa. Some laws determine a fixed rate as the threshold, others bind these regulations to a percentage of their own corporate income tax (CIT) rate.

Passive income threshold

The transactional approach contains regulations on the relative amount of passive income of a firm: The passive or financial income in the foreign subsidiary has to be below a certain amount or percentage of the total income. If the passive income in a foreign subsidiary is below a specific percentage of total income, the CFC rules of the corresponding countries do not apply. Sometimes the passive income threshold is referred to as *free harbor rule* because it allows certain behavior, specifically accruing passive income in foreign low-taxed subsidiaries, up to a specific relative border.

Additionally to these two main characteristics, this study uses the following law distinctions as well: control of a foreign entity, blacklists, whitelists, special European Economic Area (EEA) exemptions, different tax bases, shares to hold to get affected as a single owner, use of effective tax rates, active business tests and special inclusions of trust, funds or other types of foreign entities. All CFC law characteristics are described in more detail in the following section.

Consequences of being a CFC

These two requirements, as well as the further requirements, have to be fulfilled so that an enacted CFC law takes effect for the owner(s) of a foreign entity. If that is the case, then at least parts of the foreign subsidiary's income will be taxed in the parent country, usually at the parent country's tax rate as *deemed redistributed income*. Thereby, the effective CIT rate for the overall company income increases.

By setting up such thresholds, these laws constitute more a boundary to alter the behavior of multinationals than a huge tax income garner for the governments. This follows from huge differences in taxation of foreign profits in the case where companies stay slightly above or below these limits and from the fact that companies try to stay on the right side of these thresholds. It is shown in many ways that thresholds, in general, alter the behavior of subjects which are influenced by them, but not necessarily in an intended way.¹⁰ Most of these studies are researching the behavior of individuals and the study at hand is one of the first ones to use these methods for large company conduct research. One can easily expect

 $^{^{10}}$ e.g. Kleven (2016) with a good overview.

that firms try to avoid taxation by CFC rules by using the given thresholds in the laws and requirements for their foreign entities to *not* count as CFC subsidiaries. But are they trying this on a global scale? This question and the questions if these laws work for the purpose they are designed for and if they have possible negative externalities are not answered yet and examined in the following chapters.

2.2 Explanation and comparison of CFC rule characteristics and requirements

In this chapter the different CFC rules of 61 countries in the time span between 2004 and 2014 are compared. To begin with, only the observed countries which do have a CFC rule - that are 27 - are compared 11

Minimum low-tax threshold - further details

	fix threshol	d countries	relative threshold countries		
	absolute	relative	absolute	relative	
percentages	18.49%	63.64%	18.48%	66.65%	
range	10% - $25%$	49% - $85%$	10.67% - $36%$	50% - $90%$	

Table 2.1: Minimal foreign CIT rate threshold comparison

For this comparison, statutory corporate tax rates are used only. There are two ways to identify such a low-tax threshold. The first group of countries determines a fixed percentage: Denmark until 2007, Germany, Hungary since 2010, Israel, Japan, South Korea and Turkey. The other larger group of remaining countries defines a relative percentage of their own CIT rate. As one can see in Table 2.1 the resulting absolute percentages set as thresholds are very similar in both groups, and amount on average to around 18.48%. In relative terms, the fixed-group is on average, with 63.64%, more than three percentage points below the group

¹¹ Excluded is Estonia as their CFC legislation affects only individuals and no corporations.

with relative thresholds. The range in both measures is broader in the relative-group. If the United States, as an extreme outlier, is excluded, the absolute rate for the relative group reduces to 17.01% and the relative percentage number levels to 64.49%. As one expects, the absolute numbers are considerably lower if the countries are sorted in low-taxed/high-taxed country groups. Countries which have absolute thresholds below or equal to 15% are: China, Greece, Hungary, Iceland, Israel since 2014, Korea, Lithuania, Portugal since 2007, Sweden since 2009, and Turkey. By contrast, no threshold is implemented in the CFC rules of Australia, Brazil, Canada, Denmark since 2007, Italy before 2009, New Zealand as well as the UK after 2012 (but quasi, in one of the "gate tests"), and CFC taxation is conditional on other terms.



Figure 2.2: Tax Thresholds Over Time

Figure 2.2 shows the development of the average tax thresholds over all countries in absolute and relative terms over time and, additionally, the average CIT rates per year. The results for the average tax rate are in line with Alexander, Vito, and Jacob (2016). Both threshold lines decline around three to four percentage points over the observed time span. This effect is mainly driven by the bigger group of relative tax threshold definers and their declining home CIT rates over time. But also some of the fix tax rate definers are lowering their thresholds in the observed time span, most likely due to the worldwide decline of corporate income tax rates and tax competitiveness as described above. The average statutory corporate income tax rate in all observed countries and the shown time span declines from 25.40% to 21.67% by nearly four percentage points.

Passive income threshold - further details

As noted above, many countries have passive-to-total income thresholds which are set differently. For the countries considered, these are: 5% for Australia, New Zealand and the USA; 10% for Germany; 15% for Spain; 20% for Mexico; 25% for Portugal and Turkey; 30% for Greece and 50% for China, Denmark since 2007, Estonia since 2011, Israel and Italy since 2009. Especially in the case of numbers below 15%, they are often referenced to as free harbor rule because they allow certain behavior up to a specific line. A comparable requirement are the so-called "active business exemptions", which exist in some countries additionally. However, these requirements are stated in words rather than numbers. They exempt certain CFC's if they fulfill some real economic activity requirements, which differ across countries and often request verifiable real economic activity. Such special exemptions exist in: Australia, Canada, Finland, France, Greece, Hungary, Italy, Japan, New Zealand since 2009, South Korea, Spain and the United Kingdom (UK). Usually these exemptions use words like "mainly active income" or something alike which then again is broadly understood and used as a 50% barrier even in many law guidelines. Due to these exemptions, it is possible for multinationals to have a subsidiary in a low-tax country and still shift profits to this entity without getting taxed by CFC rules. Thereto, they shift income until they are slightly below the mentioned thresholds. The underlying question if that is actually the case is further researched below. A crucial fact here is that these companies are easily able to manipulate their books in such a way as to shift as much profit as possible while staying below the relative thresholds.¹²

As a further specialty of this threshold, additionally, in nearly all EU countries' CFC rules do not apply to subsidiaries in the EEA/EU if some similar requirements ("mainly active income") of the subsidiary are fulfilled. Many of them were implemented after the Cadbury-Schweppes-Case of the European Court of Justice (ECJ) in 2006.¹³ These passages are officially implemented in the CFC legislation of: Finland since 2009, France, Germany since 2007, Greece, Hungary since 2012, Iceland, Italy since 2009, Norway since 2008, Portugal since 2012, Spain, Sweden since 2008 and UK since 2006. And even if the law was implemented later in most of the countries, companies were de lege forbidden but de facto allowed to use this exemption right after the ECJ case. These requirements usually include a generalized "mainly active activity" passage, which requires the foreign subsidiary to be mainly of active nature to not be affected by CFC rules.

Control threshold

As the name CFC legislation says, the foreign subsidiary has to be controlled by the owner in the home country. This defines the third main requirement for CFC rules. In most cases this control is defined by an ownership threshold of 50% or by an imprecise phrase like "exert influence due to contracts", "dominant influence" or "have to control".¹⁴ Deviant from this definition are eight countries: Canada where 10% of interests are deemed sufficient to control the foreign subsidiary; Denmark before 2008 where also 25% of shares are sufficient; France before 2005 where only 10% of control was enough; In Italy a rule for "related entities" is included in the CFC law where 20% of profit entitlements are enough to count as an owner;

¹² Anecdotal evidence: Many accountants and researchers in the accounting area persuaded me that firms or rather their accountants are easily capable of trimming the profits in subsidiaries to a desired outcome. Additionally there are studies which show exactly this capability: e.g., Collins and Shackelford (1997), Burgstahler and Dichev (1997), or Badertscher, Phillips, et al. (2009)

 $^{^{13}}$ European Court of Justice (2006)

¹⁴ As described above for this survey no distinctions are made between different possibilities of ownership like voting rights, shares, or interests.

The Mexican CFC rule does not foresee any precise percentages in terms of control but only states phrases about "linkage between parties" or "effective control"; In South Korea until 2006 also "controlling" or holding more than 20% of the voting power was sufficient. afterwards 20% of general controlling power and since 2012, 10% is enough to deem the foreign subsidiary as controlled; The Swedish CFC rules set the control threshold at 25%; Until 2010, a control of 1% was enough for an individual in Hungary, whereas now a company (1) has to control 25% or in the case of an individual (2), 10% control or a dominant influence or the majority of income of the CFC derives from a Hungarian source are required. These thresholds mostly contain indirect control through other related parties. In eight countries a threshold for single entity control exists in addition: If in Australia one resident entity controls a foreign subsidiary with at least 40% or "de facto", the subsidiary is deemed to be controlled; Until 2007, 25% control by one entity were enough for the subsidiary to count as a CFC in Denmark; In France 10% of control, held alone, are sufficient; The Hungarian CFC law contains an alone-control threshold of 25% since 2010; In Israel holding 40% of control by one entity is sufficient; The Portuguese CFC rules take effect if one resident entity controls 25% of the foreign subsidiary; In South Korea 20% of alone-control was sufficient until 2006; With some restrictions, 40% controlled power by one entity are sufficient in the UK.

In the CFC rules of nearly all countries affected, a minimum interest that an owner has to have in foreign subsidiaries to get affected by this law exists in addition. This is especially of interest if an entity controls a foreign subsidiary jointly with other companies. Therefore, in Canada and Germany, it is sufficient if one entity has at least 1% of interests in the foreign subsidiary; In France since 2005 and in Japan until 2010, 5% were enough to get texed on the bases of CFC rules; A resident entity in China, Estonia, Finland until 2009, Hungary since 2010, Israel, Japan since 2010, Lithuania, New Zealand, Portugal and the USA is affected by CFC legislation, if it has an interest of 10% or more in the controlled foreign entity; In Finland since 2009 and in the UK, this threshold is at 25%.

Tax base

Another important aspect of CFC rules - which differs across countries - is which part of the subsidiaries' income they affect. The income taxed if the CFC rule takes effect, that is the tax base, differs mainly in two regards. In six of the observed countries, only passive income of the foreign subsidiary is the tax base for CFC taxation: Australia, Canada until 2009, Germany, Israel, Lithuania, Spain and thr UK since 2013. The other group of 13 countries denotes the full (pro rata) income of the foreign subsidiary as tax base: Brazil, China, Denmark, Estonia, Finland, France, Iceland, Italy, Japan, Norway, South Korea, Sweden, Turkey, the UK until 2013 (but with many exemptions). The only countries which changed the tax base of their CFC laws significantly in the observed time span were Canada, Hungary and the UK. As a third category, one could think of countries determining the tax base in between these two definitions, that is all income with big exemptions or passive income with inclusions, i.e. Canada since 2009, Mexico, New Zealand and the USA. As a special case, Portugal and South Africa use the after tax income of the foreign subsidiary as CFC rule tax base. Until 2010, the Hungarian CFC rules denoted only dividends for individuals as tax base and added, under some restrictions, qualifying undistributed after-tax profits, capital gains, liquidation gains, consideration paid to a CFC and further measures as tax base for companies as well.

Effective tax rate

If a described minimum tax threshold is defined, it can make a huge difference if this rate is the statutory tax rate or an effective one. All countries prescribe the use of an effective tax rate except Denmark until 2008 and Lithuania. These two countries aim at the foreign statutory corporate income tax as written in the law. The usage of effective CIT rates was not common in every CFC law before the observed time span and was used by countries to attract firm investment - or more passive income - and by corporations as a tax loophole. How exactly "effective" is measured differs across countries

Taxing Individuals and Corporations

Nearly all of the countries considered include a taxation of individuals; only Turkey, the UK (however, another similar individual tax law is implemented) and until 2006 South Korea do not tax any income via CFC rules if the owner of the foreign subsidiary is a resident individual person. If the owner of the CFC is a company, Estonia does not tax any income of that foreign subsidiary via CFC rules. Yet, other anti-avoidance rules for companies are implemented in the Estonian tax law. In Hungary, only Hungarian individuals and no companies as owners were considered under the CFC regime until 2010.

Legal form of the foreign subsidiary

Every observed country includes the income of corporate foreign entities in the CFC rules of its tax laws. In many jurisdictions, however, hybrid or special types of legal business forms are available, which are not necessarily considered as companies in legal terms.¹⁵ Most of the observed countries try to catch these legal forms also in their CFC rules. In the following countries which do not or only to some extend include specific non-corporate legal forms into CFC laws are outlined. In the CFC rules of Australia, other than corporate foreign entities are not explicitly concerned but extra foreign investment funds (FIF) and deemed present entitlement rules existed until 2010. Since then these rules were planned to be replaced with new anti-avoidance rules which did not happen until the end of 2017. In addition, transferor trust rules are enacted in Australia. The Chinese CFC rules do only cover enterprises and do not specifically concern non-corporate legal forms of the foreign subsidiary. Denmark implemented an extension into the CFC rule so that Permanent Establishments are concerned too since 2007. Also, some special investment companies are taxed in another way. In Estonia, only corporate foreign entities are concerned with CFC rules. Some minor regulations catch a few other foreign investments to be taxed in Estonia. Since 2009, foreign

¹⁵ e.g., Permanent establishments, Partnerships or Trusts. In the following no further specification is made due to comparability reasons. Also, in the various laws the distinction is often not clear.

subsidiaries from a Finnish resident entity are concerned by CFC rules as well, if they have a non-corporate legal form and, in addition, general anti-avoidance rules apply. In Hungarys CFC rule, a foreign subsidiary is defined as a foreign person or its permanent establishment abroad until 2010, and afterwards, as a foreign person or foreign resident. In the CFC rules of New Zealand, non-corporate entities are not explicitly concerned but like in Australia, a FIF regime exists to cover other investments abroad. Turkey does not explicitly include non-corporate foreign entities in their CFC rules. All in all, one can say that most countries try to catch as much entity forms as possible and modify their CFC laws accordingly.

White list and Black list

Some of the observed countries enacted a black or white list, which contains foreign countries in which subsidiaries are *certainly* or *certainly not* affected by CFC rules. Italy was the only country with regulations based entirely on a black list and without any low-tax threshold until 2009. Other countries implemented a black list in addition to their tax thresholds, in particular in Finland since 2012, Greece with its CFC rule implementation in 2014, Lithuania, Mexico, Norway and Portugal. Contrary, a white list exists in the CFC laws of Australia, China, Estonia (mainly EU and tax treaty partners), Hungary, Lithuania, New Zealand (since 2009 only Australia), Norway and Sweden. Often the white list does not provide blanket protection for potential CFCs. Instead it is linked to some preconditions like real business presence or further subsidiaries in non-white list countries.

Tax credit

Generally, all countries provide a tax credit with different values for paid taxes from the foreign subsidiary. Sometimes little, and very specific restrictions exist. The only exemptions are Hungary and Iceland which do not have specific tax credit provisions, and Spain where the foreign tax is only deductible. All the mentioned characteristics are backed up by, substantiate or amend the recent work of the OECD (OECD (2015a) respectively. The policy considerations and objectives of the different CFC rules described, and the characteristics used in this work are consistent with those used by the OECD. As mentioned, due to data and practicability reasons not exhaustively all CFC rule characteristics discussed in the OECD work are captured in this comparison at hand.

2.3 Literature on CFC rules

All in all, only few studies have researched the impact of CFC rules, although these laws get increasingly noticed and more importantly evaluated. Studies which are not focused on CFC rules, use a simple CFC rule dummy for checking whether an enacted CFC rule in the parent country has an effect on their baseline regression results.¹⁶ Other work, which is focused on CFC legislation, let the CFC dummy variable depend on one characteristic like the low-tax threshold (Ruf and Weichenrieder (2012)) or two characteristics like the low-tax threshold and a de-minimis rule (Egger and Wamser (2015)). Ruf and Weichenrieder (2012) show that the German CFC legislation prevents passive investment in foreign low-tax countries and Egger and Wamser (2015) are able to show that the German CFC law does have a distorting effect on real foreign direct investments (FDI) if the foreign subsidiary is treated as a CFC. A third paper by Ruf and Weichenrieder (2013), shows that the Cadbury-Schweppes-Case of the ECJ (European Court of Justice (2006)) leads to more passive assets in foreign low-tax subsidiaries of German multinationals within the EEA. In a recent working paper Albertus (2018) shows that foreign CFC rules do have an influence on the real economic activity of foreign owned U.S. subsidiaries, especially the investment into low-tax subsidiaries further down the shareholder chain is analyzed. These few analyses and others observe the effects of one country legislation only and/or CFC rules serve as a robustness check with a

¹⁶ Lohse and Riedel (2013); Keller and Schanz (2013)

simple dummy.¹⁷ In addition, historical and law focused analyses of CFC rules are about one country or the comparison of two CFC legislations over time.¹⁸ Kane (2013) analyzes the foundations of CFC rules in comparison to transfer pricing rules, and suggests to define the core problem one is trying to fix first, and then examining the actual fit between the proposed solution and the purported problem afterwards, to gain better results and clearer legislations. Two law studies try to answer the question how "good" CFC legislation should be constructed from the juristic perspective.¹⁹ They show that it is possible to achieve a second best solution at least. In a study from Avi-yonah and Halabi (2012), the U.S. CFC rule, namely Subpart F income, is compared to other OECD CFC legislations and Aviyonah (2017) wrote recently about the original intend of the Subpart F income legislation. and how far the current version of the law is away from that intention. This became more interesting since the mentioned GITLI provision in the U.S. In a recent theoretical work, Haufler, Mardan, and Schindler (2018) show under which conditions CFC rules should play a role in an optimal tax mix including profit shifting behaviour of multinationals, also including thin capitalization rules. Further recent research, as the work from Dowd, Landefeld, and Moore (2017) about profit shifting in the U.S. including calculated elasticities, is supported by the results on CFC rules of the study at hand.

One study found in a major journal, which analyzes comparable CFC legislations of various countries empirically in detail is Voget (2011); using the fact of relocating headquarters. For this study, one of the main explanatory factors are CFC rules. Thereto, the author observes the CFC legislation for 22 countries in year 2008, including seven characteristics. In the baseline estimations, only a CFC dummy variable is used; if the parent country does have a CFC rule enacted or not. Later, four different characteristics are included and estimated in robustness checks. Significance is shown for one of the four CFC characteristics only, which are established as dummy variables. This significant variable is comparable

¹⁷ e.g., Altshuler and Hubbard (2003); Haberly and Wojcik (2015); Griffith, Miller, and O'Connell (2014)

¹⁸ Avi-yonah and Halabi (2012); Castro (2013); Redmiles and Wenrich (2007)

¹⁹ Pinto (2009); Rust (2008)

to the passive-to-total income threshold, analyzed in the study at hand. A second study on comparing CFC rules is from Bräutigam, Spengel, and Streif (2017), who show that the introduction of the mentioned Cadbury-Schweppes-Case of the ECJ European Court of Justice (2006) led to more profit shifting into low-tax countries within the EU due to more lenient CFC rules for EU member states. Further on, they provide evidence that the following 'IP box regime' introduction, which allows specific tax reliefs for research and development, patents and following royalty payments, led to even more tax distortions within the EU. Thereto, the authors used an extension of the Devereux-Griffith methodology (Devereux and Griffith (2003)) for the calculation of effective average tax rates and implemented the two varying laws. One further study from Markle and L. Robinson (2012), which is not published in a journal yet, shows the importance of CFC rules and their different impacts on multinational firm behaviour concerning the use of tax havens. Additionally, this study investigates different characteristics of CFC legislations in more detail. The authors develop a 9-point scale of inclusiveness of CFC rules, which is larger if the scope of income, caught by the law and taxed, is wider. As law data base, the CFC rules of 18 countries of year 2011 are used. One key finding is that effective CFC rules are a preferable instrument to prevent firms from using specific tax havens, than higher withholding taxes on interest or the taxation of dividends.

Weichenrieder (1996) uses a theoretical model to show the impact of anti-tax-avoidance provisions on the size and the growth of foreign subsidiaries. He extends the study of Hines (1994) and shows that such rules can, unexpectedly, lower the cost of capital of foreign subsidiaries, and - specific restrictions provided - can increase the subsidiaries' size. These anti-tax-avoidance provisions are basically CFC rules and their passive-to-total income threshold, which are empirically researched in the study at hand.

Another recent working paper from Clifford (2017) shows similar results as presented in the first part of Section 3.6. The second part of Clifford (2017) tries to answer the question about location choices of multinational greenfield investment, which are possibly influenced by CFC rules. Although, answered with a different data set and empirical method, the results are comparable to the ones shown in the fourth baseline regression of Table 3.1 in Section 3.6 and the whole Chapter 4.3.4. These results indicate, that multinationals with headquarters in a CFC rule country prefer to locate their subsidiaries in countries with CIT rates higher than the low-tax threshold of the according CFC rule but lower than their own home CIT rate. In this case they are still able to shift profits for a lower overall taxation.

This dissertation adds to the rare existing literature and knowledge about CFC legislation, by depicting different CFC rules from various countries all over the world and over time. Such a broad study and comparison of these increasingly important laws is not published yet, to the best of my knowledge. Using different data bases, various graphical and econometrical methods, I show the impact of CFC legislation on passive income abroad, and therefore the profit shifting behaviour of multinationals. Additionally, this dissertation provides theoretical thoughts of profit shifting behaviour of multinationals and its limitations by CFC legislation. These thoughts are researched empirically to compare the behaviour of firms on a tax notch, and show compelling evidence of behavioural responses of companies around the globe in Chapters 3 and 4.

2.4 CFC rules per country

This section presents a detailed description of CFC legislations and their development from 2004 to 2014 for every country researched in this dissertation. For every country a short and summarizing table of CFC law differences between 2004 and 2014 is provided.

CFC rules in Australia

The CFC rule in Australia (Part X of ITAA 1936) exists since 1990. No bigger CFC law amendments happened in the observed time span for the monitored characteristics. As of

the end of 2017, the Australian government assess' the OECD BEPS review to implement new recommendations.

	2004	2014
tax base	pass inc	pass inc
min. control	50%	50%
min tax absolute	none	none
min tax relative	none	none
black-/ whitelist	no/yes	no/yes
active business test	yes	yes

Table 2.2: CFC rules in Australia over time

An Australian shareholder may be liable to tax the passive income ("attributable income") of a foreign subsidiary if, generally, one requirement is fulfilled. The subsidiary has to be controlled by Australian attributable taxpayers. To control a foreign subsidiary it is necessary to hold, directly or indirectly, together with four or fewer Australian residents more than 50% of interests in the CFC, or control effectively the CFC respectively. The requirement is also fulfilled if one Australian entity holds 40% or more of interests. Also, income can only be attributed if the Australian resident is an attributable taxpayer (interest of at least 10%). No low-taxation threshold exists. The attributable income may vary, depending on the country of the CFC (whether it is listed or not). Generally, active income is not attributed but only certain passive income. Nevertheless, if more than 5% of a subsidiarys turnover is of passive nature, a save haven active income test is failed and results in the attribution of "adjusted tainted income"/passive income to the home tax base, as long as the CFC is located in an unlisted country. If the subsidiary is located in a listed country, the attributable income is smaller. A tax credit for foreign paid taxes and minor other tax decreasing specifics may be granted. No proper black- or whitelist exists. In addition to the CFC rule, a Foreign Investment Fund rule applies to capture special dividends and other payments.

CFC rules in Austria

In Austria, no actual CFC rule exists. However, there are rules (Art. 10(4)-(6) of the CIT Code) under which the undistributed income of a foreign company may be taxed at the Austrian shareholder level on a pro rata basis, if the subsidiary is characterized as a foreign real estate or investment fund. These rules prohibit the tax-free repatriation of shifted profits from the foreign subsidiary back to an Austrian resident and are in the spirit of CFC rules.

CFC rules in Brazil

The CFC legislation in Brazil (Provisional Measure No. 2,158-35/2001 and Normative Instruction No. 213/2002) does exist since 1995 but became effective since 2001. No bigger CFC law amendment happened in the observed time span for the monitored characteristics but the Brazilian government changed the law after 2014.

2004 2014 tax base inc inc 20%20%min. control min tax absolute none none min tax relative none none black-/ whitelist no/no no/no active business test no no

Table 2.3: CFC rules in Brazil over time

The Brazilian CFC rules apply to companies subject to the equity pick-up method for accounting purposes and, basically, include income from foreign affiliates very broadly, compared to other CFC regimes. If a Brazilian corporation (or individual) has an affiliated company abroad, which is the case if, (i) the investor holds significant influence, i.e. the right to participate in the decisions of the invested company; or (ii) the investor holds more than 20% of the invested company without controlling it, the income of this subsidiary is taxed at the parent level. The control interest of 20% or more can be hold directly or indirectly. No low-taxation or passive-income threshold exists in Brazil. The only explicit exemption exists for companies which are active in the oil and gas sector, which are exempt from CFC enforcement. A tax credit for paid taxes in the foreign country exists. Nevertheless, there is no tax credit carryforward unless the Brazilian company was in an NOL position (net operating loss limitations (i.e. 30%)) at the time the foreign income was taxed. No black- or whitelist exists. Nonetheless, the Brazilian CFC law had many loopholes in the details, which partly got corrected with the new law changes in 2014/15.

CFC rules in Canada

Canada was one of the first countries which implemented a CFC legislation (Sec. 90-95 and 112/113 of Income Tax Act and Part LIX of Income Tax Regulations) worldwide in 1972. The only amendment in the considered time span for the regarded characterizations was in 2009.

Table 2.4: CFC rules in Canada over time 2004 + 2014

	2004	2014
		quasi
tax base	pass inc	pass inc
min. control	10%	10%
min tax absolute	none	none
min tax relative	none	none
black-/ whitelist	none	none
active business test	yes	yes

At first glance, Canada seems to have a comparably strong CFC law within the OECD countries. A foreign entity is considered to be a CFC ("Foreign Affiliate") if a Canadian resident, individual or company, does have interests, directly or indirectly, in any class of shares of at least 1% and together with related persons of at least 10%. The "foreign accrual property income" (FAPI) consists of passive income and further designated income which, pro rata, is taxed in Canada with the respective corporate or personal income tax of the Canadian taxpayer. Before 2009, the considered and taxed income was passive income only, without the additionally deemed passive income. A participation exemption for active income for subsidiaries in designated treaty countries exists. No low-tax threshold and no black- or whitelist exists. A tax credit for foreign paid taxes on the FAPI will be granted. Canada

also have some further foreign investment entity rules to catch broader foreign investments. Nonetheless, it was rather simple to circumvent Canadian CFC rules with so called "tracking arrangement" structures, which should be harder to facilitate after a recent law change in early 2018.²⁰

CFC rules in China

The CFC rule (Enterprise Income Tax Law (Order of the President [2007] No. 63, EIT Law), Article 45 and its implementation rules) in China is in force since 2008. In the observed time span no elementary amendments in the observed characteristics occurred.

Ta	ble 2.5: CFC rules i	n Chin	a over time
		2004	2014
	tax base	none	full inc
	min. control	none	50%
	min tax absolute	none	12.50%
	min tax relative	none	50%
	black-/ whitelist	none	no/yes
	active business test	none	none

A Chinese resident (company or individual) who is a shareholder of a foreign entity may be taxed on the undistributed pro rata income of this entity if three requirements are fulfilled: (1) the foreign entity is controlled by Chinese residents (including other Chinese residents, they have more than 50% of interests in the subsidiary or effective control), and (2), the concerned Chinese shareholders hold at least 10% of interests of the subsidiary (directly or indirectly), and (3), the effective tax rate for the CFC in the foreign country is less than 50% of the current Chinese income tax rate. If these requirements are fulfilled the pro rata CFC income must be included into the gross income of the Chinese resident. Some taxes payed by the subsidiary in the foreign country can be credited against the "CFC taxes" in China. Besides some very special exemptions, a de minimis rule exists, which exempts foreign income if the profits are below 5 million RMB or if the income is mainly active. No

²⁰ e.g., DLA piper (2018); EY (2018a); EY (2018b)
blacklist exists but a whitelist (including Australia, Canada, France, Germany, India, Italy, Japan, New Zealand, Norway, South Africa, the UK or the US) is in force. Even though the Chinese CFC rules where legally in force since 2008, these laws were rarely if ever enforced until a public case in the year 2014 (e.g., Qio (2017); AmiNews (2017)). This, in reverse, does not necessarily mean, that CFC rules in China were not influencing firm behaviour by deterring effects.

CFC rules in Denmark

The CFC rule in Denmark (Art. 32 of Corporate Tax Code and Art. 16H in the Tax Assessment Code for individuals) do exist since 1995. In the observed time span one bigger law amendment happened in 2007.

	2004	2014
tax base	full inc	full inc
min. control	50%	50%
min tax absolute	22.50%	none
min tax relative	80%	none
black-/ whitelist	no	none
active business test	none	none

Table 2.6: CFC rules in Denmark over time

A Danish parent company, or individual, may be liable to tax the income of a foreign subsidiary if two requirements are fulfilled. First the subsidiary has to be controlled by the Danish parent company and, second, the subsidiary's business is mainly of a financial nature. Latter is fulfilled if more than 50% of its total income is passive income and more than 10% of its assets are passive assets/of a financial nature. To control a foreign subsidiary it is necessary to hold, directly or indirectly with related groups, more than 50% of the voting power. If these conditions are met the Danish parent is taxed on a pro rata share of the foreign passive income with the relevant Danish CIT. This also applies to individuals. A tax credit for taxes paid by the subsidiary is granted. Before 2008 the CFC rules were slightly different: To control a subsidiary it was sufficient to hold at least 25% of the shares and the threshold of passive income was not 50% but two thirds of the total income of the subsidiary. In the observed years from 2004 to 2007 also a foreign low-tax threshold was installed, varying over years from 22,5% to 21%. No black- or whitelist exists in the Danish CFC rule.

CFC rules in Estonia

Estonia has no common CFC rule. In Estonia, resident individuals only are affected by the CFC regulations since 2000. For resident companies a comparable but weak anti-avoidance rule applies. If an Estonian individual, alone or with associated persons, controls (at least 50% of the capital of the foreign company are held by Estonian residents) a subsidiary in a foreign low-tax country (effective tax rate is less than one third of the Estonian income tax) and holds at least 10% of the shares of the CFC (directly or accompanied), the pro rata income of this subsidiary is attributed to the income tax and charged in Estonia. Estonia has an official whitelist of countries that are not considered as low-tax jurisdictions, comprised mainly of tax treaty partners, EU member states and certain other OECD countries. No blacklist exists. A tax credit for the paid foreign taxes is granted. If more than 50% of the income of the foreign subsidiary is derived from genuine business activities the foreign entity will be tax exempt from CFC regulations. Within the anti-avoidance regulations for companies, more payments to foreign companies in low tax jurisdictions are non-deductible for corporate income tax purposes since 2012.

	2004	2014
tax base	full inc	full inc
control	50%	50%
min tax abs	7%	7%
min tax rel	33%	33%
black-/ whitelist	no/yes	no/yes
active business test	none	none

Table 2.7: CFC rules in Estonia over time

CFC rules in Finland

The Finnish CFC rule (Act on the Taxation of Shareholders in CFCs) is in force since 1995. One bigger tax reform happened in 2009. An entity of Finland (company or individual) who is a shareholder of a foreign company may be taxed on the undistributed pro rata income if three requirements are fulfilled: (1) the foreign entity is controlled by the Finnish resident, (2) the Finnish shareholder holds at least 25% of interests in the subsidiary (directly or indirectly), and (3) the effective tax rate in the foreign country is less than 60% of the current Finnish tax rate. The foreign subsidiary is deemed to be controlled by a Finnish resident (or jointly with other Finnish residents) if she holds at least 50% of interests in the subsidiary. Taxes payed by the subsidiary in the foreign country are credited against the CFC law inflicted taxes in Finland. Two explicit exemptions exist: (1) for exempt activities (if the subsidiary runs on a genuine business activity base), and (2) a tax treaty exemption (if a tax treaty with the corresponding country is in force, provided the foreign CIT is not lower than 75% of the Finnish income tax). Since 2009, subsidiaries in an EEA country (excluding Lichtenstein) and countries with whom Finland does have a tax information exchange agreement receive a special treatment. They are generally exempt from the CFC rule, provided they carry on genuine and actual economic activities. In 2012 the Finnish tax authorities introduced a greylist of countries (containing 13 countries) for which these exemptions are not applicable. No actual white- or blacklist exists for CFC rule purposes in Finland.

	2004	2014
tax base	full inc	full inc
control	50%	50%
min tax abs	17.40%	12%
min tax rel	60%	60%
black-/ whitelist	no	yes/no
active business test	none	yes

Table 2.8: CFC rules in Finland over time

CFC rules in France

Since 1980, CFC rules in France (Art. 209-B of the French Tax Code) exist. This specific CFC law is for companies only but a complementary one persists for individuals. The CFC legislation applies to an entity subject to French CIT, provided the following two facts. First, it has to hold, directly or indirectly, at least 50% of the shares, financial rights or voting rights of a foreign legal entity (or permanent establishment)²¹ which, second, is established in a country with an effective taxation of 50% or lower than in France. If the control threshold is reached by a group it is reduced to 5% for every single entity as an anti-abuse provision. In the considered time period these two facts changed in 2005. Before that year the law was a bit stricter, as follows: a CFC was regarded as such if the held control share was at least 10% and the foreign CIT threshold was at two thirds of the French CIT rate. If the law applies, the received, or deemed received, pro rata income of the foreign subsidiary is taxed with the CIT in France. The paid tax on that income in the foreign country can be credited against the tax amount in France. Since 2011 in France exists a list of countries that are deemed to be low tax countries but no actual legal black- or whitelist exists. France has a lot of tax treaties with other countries and they have to be considered in the specific case because there are possible tax reliefs. Also, a "save harbour clause" exists: if a French company can prove that the localization of the foreign entity is not motivated by tax reasons, it may avoid the CFC rule and, additionally, within the EU only artificial structures abroad are affected by the French CFC rule.

	2004	2014
tax base	full inc	full inc
control	10%	50%
min tax abs	22%	16.67%
min tax rel	66%	50%
black-/ whitelist	no	yes/no
active business test	yes	yes

Table 2.9: CFC rules in France over time

 $^{^{21}}$ There are differences between these distinctions, in other countries as well but they are too specific for this general comparison. If the differences are important for treatment I will distinguish them anyway.

CFC rules in Germany

The CFC rules in Germany (Art. 7-14 AStG) were implemented in 1972. After a few changes the law was reformed again in 2003 to, inter alia, implement intra-company lending provisions. In the observed time period, minor CFC law changes occurred only. In Germany both, individuals and companies, are effected by the CFC rules if they fulfill the following requirements. The German parent, or together with associated individuals, have to have a control share of at least 50% of the foreign subsidiary. As an anti-abuse provision, this threshold is reduced to 1% for each direct shareholder if the 50% are reached by a group. The foreign country is considered to be a low-tax country if the CIT rate of the foreign subsidiary is below 25%. The tax base is the passive income of the foreign subsidiary. If the CFC rule applies, the pro rata passive income of the subsidiary is taxed in Germany with the German CIT rate, credited by a tax deferral of the payed CIT by the subsidiary in the foreign country. The law takes no effect if the passive income is smaller than 10% of the total subsidiary income, or the applicable income is below the de-minimis threshold of 80000 EUR. Germany does not have a white- or blacklist of countries for CFC rule purposes. In certain cases, the CFC income attribution does not apply, if the subsidiary is located in an EU/EEA country and the taxpayer can demonstrate that the CFC carries out genuine business activities (since 2007). There are more special treatments for special cases provided in the law.

	2004	2014
tax base	pass inc	pass inc
control	50%	50%
min tax abs	25%	25%
min tax rel	65%	85%
black-/ whitelist	none	none
active business test	yes	yes

Table 2.10: CFC rules in Germany over time

CFC rules in Greece

Greece implemented a CFC rule (Art. 66 of Income Tax Code) recently in 2013 which is applicable since 2014. This rule concerns both, individuals and companies. The (un-)distributed income of a foreign subsidiary is taxed by a Greek entity, with the appropriate Greek CIT, if: (1) the Greek resident (together with related entities) holds more than 50% of interests in the foreign subsidiary, and (2), the subsidiary is situated in a country that has a preferential tax regime or is on a blacklist, and (3), more than 30% of the subsidiaries net profits are passive, and (4), the principal class of shares of the foreign subsidiary are not traded on a regulated stock market. A foreign country does have a preferential tax regime if its CIT rate is lower than 50% of the Greek ones. No tax credit for foreign taxes is provided. The only exemption given is for subsidiaries in an EU/EEA member state that has negotiated a tax information exchange agreement, but that exemption falls if the subsidiary has an artificial tax avoiding nature.

bie 2.11. Of C fulles 1	2004	2014
tax base	none	full inc
control	none	50%
min tax abs	none	none
min tax rel	none	none
black-/ whitelist	none	yes
active business test	none	yes

Table 2.11: CFC rules in Greece over time

CFC rules in Hungary

In 1997 a CFC rule (Art. 4, 7, 8, 18, 29/Q and Annex 3 of the CIT Code) was implemented in the Hungarian law. One of the biggest changes in this rule occurred in 2010. Before that (1) any interest in a foreign entity by a Hungarian individual, and (2) a foreign CIT rate lower than two-thirds of the Hungarian CIT rate qualified the foreign entity as a CFC. The dividends received by, and pro rata expenses or losses booked for, that entity are taxed in Hungary with the CIT rate. In 2007 additional entities in countries of the EU, OECD or tax treaty partners of Hungary were exempt from the CFC provision. No rules affected the undistributed or deemed profits of a CFC. Nonetheless, all this rules affected Hungarian resident individuals only. In 2010 the CFC system changed: a foreign country deemed to be a low tax country if the effective tax rate for the foreign entity is below two-thirds of the Hungarian one; a Hungarian entity, now also companies were considered, has to hold, directly or indirectly, at least 25% of interest in the foreign company, and the pro rata after tax passive income²² is considered to be taxed if the CFC rules apply. Additionally, a real economic activity exemption exists since than. One year later, in 2011, the foreign tax threshold was lowered and decoupled from the home CIT rate to a fix percentage of 10%. In 2012, the real economic proof was shifted to the taxpayer. No actual white- or blacklist exists and no special tax credit provisions apply. In 2008 and 2009 a huge tax amnesty applied for the repatriated income of CFCs (not for CFCs in Andorra, Lichtenstein and Monaco) if at least 50% of the repatriated funds are invested in treasury bonds and this investment is held for at least 2 years. Although, the CFC rules in Hungary have become more complex, they generally are not relevant for non-Hungarian controlled multinationals.

	2004	2014	
tax base	dividends	dividends and more	
control	1%	25%	
min tax abs	10.67%	10%	
min tax rel	67%	53%	
black-/ whitelist	none	none	
active business test	yes	yes	

Table 2.12: CFC rules in Hungary over time

CFC rules in Iceland

Iceland implemented a CFC rule (Art. 57 a) of Income Tax Act) in 2010. According to this rule a CFC is a foreign entity, (1) located in a low-tax country, with an effective income tax lower than two thirds of the Icelandic rate, (2) which is owned or controlled (directly or

²² Actually, there is a list of concerned income but it is more or less passive income.

indirectly) by at least 50% by a resident taxpayer, corporate or individual. The profits of this entity, distributed or not, are then pro rata attributed to its resident shareholders and respectively taxed in Iceland. No tax credit is granted but two exemptions exist when no CFC rule taxation takes place: If the CFC is located (1) in a treaty country and the income of the CFC is not mainly of financial income or with the country exists an information exchange provision, or (2), in an EEA member state and is engaged in real business operations there and the Icelandic tax authorities are able to obtain all necessary information on the basis of an international tax treaty. No black- or whitelist exists.

	2004	2014
tax base	none	full inc
control	none	50%
min tax abs	none	13.33%
min tax rel	none	67%
black-/ whitelist	none	none
active business test	none	none

Table 2.13: CFC rules in Iceland over time

CFC rules in Israel

The CFC rules (Section 75 B of the Income Tax Ordinance) of Israel were implemented in 2003. In the observed time frame three bigger law amendments occurred in 2006, 2009 and 2014. Before that, the pro rata "deemed dividends" of a foreign company or any other body of persons has to be implemented in the residents, company or individual, tax base if the following requirements are fulfilled: (1) the foreign entity's shares are not, or less than 30% publicly traded, and (2), the profits or the income of the foreign entity are mainly (more than 50%) of passive nature, and (3), the income is effectively taxed at a tax rate lower than 20% in the foreign country, and (4), the foreign entity has to be controlled by: (4a) Israeli residents, directly or indirectly, holding more than 50% of the foreign entity, or (4b), one single Israeli resident holds more than 40% of interests and together with close relatives more than 50%. These rules were triggered if an Israeli residents holds at least 10% of the

means of control²³. A tax credit for the foreign paid taxes is granted. No black- or whitelist exists. Since 2006, not only companies but also trusts and other firm structures are affected by the CFC legislation. Three years later, in 2009, the CFC law became more diluted e.g., new immigrants were enabled to use privileged benefits of the CFC rule. According to the law amendment in 2014, along other changes, the foreign tax threshold dropped from 20% to 15% effective foreign income tax rate, the passive income is defined slightly different and the foreign tax credit changed. To catch further specific foreign investments, other anti tax avoidance laws as the rule for Foreign Professional/Occupational Companies exist concurrently to CFC rules.

	2004	2014
tax base	pass inc	pass inc
control	50%	50%
min tax abs	20%	15%
min tax rel	56%	57%
black-/ whitelist	none	none
active business test	none	none

Table 2.14: CFC rules in Israel over time

CFC rules in Italy

The Italian CFC law (Art. 167/168 of Income Tax Code) is in force since 2001/02. In the observed time frame some changes occurred to reshape the law. Since 2002, profits of a non-resident entity are deemed to be profits of an Italian resident if two main requirements are fulfilled. First, the resident controls, directly or indirectly, the foreign entity, and second, the foreign entity is residing in a tax haven, as defined in a blacklist containing around 70 countries. This applies for companies and individuals. A foreign company is controlled if the Italian resident holds the majority of the votes at the shareholder meeting, or sufficient votes to exert a decisive influence, or the foreign entity is under a dominant influence of another, due to a special contractual relationship. The pro rata full income of the CFC will be taxed

 $^{^{23}}$ For the mentioned unity reasons above I will continue using the word "interests". Due to specialty reasons, the so called means of control are mentioned here, which are further detailed in the law.

in Italy by the Italian CIT rate, or the specific residents average tax rate, but no lower than 27%. A tax credit for the taxes payed in the foreign country exists. It is possible to prove that the foreign subsidiary carries on a genuine business to avoid the CFC rules. Since 2004, the CFC rules apply if the Italian resident holds a profit entitlement of at least 20% in a foreign related entity as well. In this case a relief exists, if the profits are derived through permanent establishments of the foreign low-taxed company. After 2008, the blacklist was officially abolished and should be replaced by a whitelist but has been considered for a longer time. At the same time, the CFC rule is applicable for a foreign company if the foreign CIT rate is lower than 50% of the Italian effective tax rate, and the passive income of the CFC is more than 50% of the overall income of the subsidiary. By this change, the active business test was strengthened a bit compared to the rather subjective active business legislation before. Since then, the Italian resident companies are able to avoid CFC rule results only if they are able prove that the foreign company is not a wholly artificial arrangement for tax purpose advantages only, which requires a requested ruling from the tax authorities.

	2004	2014
tax base	full inc	full inc
control	50%	50%
min tax abs	none	15.70%
min tax rel	none	50%
black-/ whitelist	yes/no	yes/no
active business test	yes	yes

Table 2.15: CFC rules in Italy over time

CFC rules in Japan

In 1978 a CFC rule (Art. 66-6 - 66-9 and Art. 40-4 - 40-6 of Special Taxation Measures Law) was implemented in the Japanese law. One bigger amendment was implemented in the CFC legislation in April 2010, when Japan switched its underlying international tax systems to a territorial one. Before that, the pro rata profits (distributed or not) of a foreign entity may be attributed to the Japanese residents tax duty if it holds substantial interests therein, and this subsidiary is located in a low-tax jurisdiction. This rule applied and still applies to individuals and companies nearly similarly. A Japanese entity does have substantial interests if it (directly or indirectly) has at least 5% of interests where more than 50% of the overall interest is owned (directly or indirectly) by Japanese resident entities together. A country is deemed as a low-tax jurisdiction if the effective payed tax by the subsidiary in the foreign country is 25% or less. A tax credit for the foreign payed taxes is available. Before 2010 an active business exemption existed. After the law amendment in 2010, the threshold percentage of a single shareholder person was raised from 5% to 10% and the low-tax threshold dropped to 20% for the effective foreign CIT rate. Additionally, the following conditions were added to the reshaped law: (1) A clearer active business income test concerning the main business of the subsidiary, (2) a substance test (fixed place of business with a head office), (3) a local management and control test, and (4) an unrelated party transaction test or local business test. If all theses requirements are satisfied, certain passive income is affected by CFC taxation only. Otherwise, the full (pro rata) income of the foreign entity has to be included in the parent tax base. No white- or blacklist exists.

	2004	2014
tax base	full inc	full inc
control	50%	50%
min tax abs	25%	20%
min tax rel	61%	56%
black-/ whitelist	none	none
active business test	yes	yes

Table 2.16: CFC rules in Japan over time

CFC rules in Republic of Korea

The Korean CFC law (Art. 17 International Tax Coordination Law) is in force since 1997. In the observed time frame some changes occurred to reshape the law. In 2004 profits of a non-resident entity are deemed to be profits of an Korean resident if two requirements are fulfilled: (1) the resident controls, directly or indirectly, the foreign entity and, (2) the foreign entity is situated in a low-tax country (average foreign income tax of the last three years is lower than 15%). This CFC rule applied only to companies. A foreign company counts as controlled if the Korean resident holds the majority of the shares or "de facto" controls the foreign entity and holds more than 20% of voting rights. The pro rata full income of the CFC will then be taxed in Korea by the Korean CIT rules. A tax credit for the taxes payed in the foreign country exists. The company has the possibility to prove that the subsidiary carries out an active business operation to avoid CFC rule taxation. Since 2006, the Korean CFC rules apply if the Korean resident holds simply "interests" of at least 20% in the foreign related entity only, and also individuals are affected. Additionally, a de-minimis rule was implemented such that CFC income below 100 million WON were not concerned. In 2010, the de-minimis threshold was doubled to a minimum CFC income of 200 million WON. The law amendment in 2012 reduced the control threshold to only 10% of interests again. Along with this change, the active business test was strengthen a little and the requirements got stronger in juridical wording.

	2004	2014
tax base	full inc	full inc
control	50%	50%
min tax abs	15%	15%
min tax rel	51%	62%
black-/ whitelist	none	none
active business test	yes	yes

Table 2.17: CFC rules in South Korea over time

CFC rules in Latvia

No proper CFC rule does exist in Latvia. Nevertheless, some payments made by Latvian residents to foreign low-taxed companies or individuals are subject to a special withholding tax of 15%, due to law specifications (Art. 17.17 in the Personal Income Tax Code and Art. 3.8 in the CIT Code). A country is deemed to be a low-tax country if it is on the blacklist of Latvia.

CFC rules in Lithuania

Lithuania implemented its CFC rule (No. IX-675 in CIT Code) in 2002. No changes occurred to the law in the observed time frame. The CFC rules apply if a Lithuanian entity, company or individual, (1a) controls (hold more than 50% of interests (directly or indirectly)) a foreign entity, or (1b) holds at least 10% of interests, and together with related persons hold more than 50% of that foreign entity, and (2a), the foreign entity is not located in a country on the whitelist (or is located in a whitelist country but is organized in a special treated business form), or (2b), the foreign country is on the blacklist or the applicable CIT for the subsidiary is below 75% of the Lithuanian CIT. If the CFC rule applies for a CFC-construct the Lithuanian resident entity has to ad the positive passive (active income is not attributed if it satisfies establishment requirements), proportionally income into its income tax base. The foreign paid tax by the subsidiary may be credited against the tax due in Lithuania if a tax treaty is enacted with that country. The regime does not apply if the subsidiarys income comprises less than 5% of the total income of the Lithuanian entity.

	2004	2014
tax base	pass inc	pass inc
control	50%	50%
min tax abs	11.25%	11.25%
min tax rel	75%	75%
black-/ whitelist	none	none
active business test	none	none

Table 2.18: CFC rules in Lithuania over time

CFC rules in Malta

There is no actual CFC rule in Malta. Nevertheless, an investment of a Maltese resident in a non-EU passive entity is considered as a low taxed investment if the following requirements are fulfilled: (1) the foreign CIT is lower than 15%, and (2), more than 50% of the income is passive, and (3), more than 50% of income is from portfolio investments, while the foreign tax is not at least 5%. This is an anti-avoidance rule of the full participation exemption regime of Malta.

CFC rules in Mexico

Mexico implemented a CFC rule (Title VI of Mexicos Income Tax Law) in 1997 and amended it two times from 2004 until the end of 2014, that was in 2005 and 2013 to specify the rules. If a Mexican company or individual obtain income from foreign entities, distributed or not, it may have to include the pro rata part of this full income into its own income tax base if this entity is a CFC. A foreign company is deemed to be a CFC if: (1) the resident entity controls the subsidiary (namely "have to control"; since 2005 a taxpayer is presumed to control foreign legal entities, unless demonstrated otherwise), and (2), since 2005 the foreign effective tax paid by the subsidiary is less than 75% of the own Mexican tax rate, and (3), at least 20% of the CFC gross income is passive income. In addition, many specific rules exists which permit the non-taxation of active income. The paid tax in Mexico can be credited with the already paid tax in the foreign country by the subsidiary. In the Mexican CFC law no whitelist of countries exists but a black list is enacted. Especially before 2005 this black list was important as no foreign tax threshold existed, currently this black list is secondary. Taxpayers are required to provide information on investments in enlisted entities on this black list. The amendment in 2013 especially implemented an exemption for specific royalties.

	2004	2014
tax base	full inc	full inc
control	50%	50%
min tax abs	none	22.50%
min tax rel	none	75%
black-/ whitelist	yes/no	yes/no
active business test	none	none

Table 2.19: CFC rules in Mexico over time

CFC rules in Netherlands

There is no actual CFC rule in Netherlands. However, the valuation of participation in foreign companies are included in the taxable income of the resident company. To be affected, the participation has to be at least 25%, and 90% or more of the foreign assets have to be passive assets.

CFC rules in New Zealand

New Zealand implemented a CFC legislation (Subparts CQ, DN and EX of Income Tax Act 2007) in 1988. In Juli 2009 a bigger law amendment in the considered time span for the regarded characterisations happened. A foreign entity is considered to be a CFC if a group of five or fewer New Zealand residents, individuals or companies, do have interests, directly or indirectly, of at least 50%, or this group effectively controls the subsidiary, or a single resident has at least 40% of interests in that entity (and no other non-resident bigger shareholder exists), or de-facto control exists. Respectively, an income interest of 10% or more in a CFC from a New Zealand resident is required to be subject to the CFC rules. The attributable income consists mainly of passive income and further designated income, which - pro rata - will be taxed in New Zealand with the respectively tax by the resident taxpayer. Since 2009, an active business income test or rather a save harbour threshold applies to all CFC's (like in Australia). Therefore, the income of a foreign company will not be taxed if it consists of less than 5% of passive income. No actual whitelist exists but before 2009, a special exemption rule existed for greylist countries, provided some specified relief criteria are met. This greylist exemption was abolished, with exception to Australian CFCs. No low-taxation threshold and no blacklist exists. A tax credit for foreign paid taxes on the attributed income will be granted. As in Australia, additionally to the CFC rule, a "Foreign Investment Fund" rule applies to capture special dividends and other payments, not from funds only but further special company structures as well.

14,510 2.20. 01 0 10	2004	2014
tax base	quasi pass inc	quasi pass inc
control	50%	50%
min tax abs	none	none
min tax rel	none	none
black-/ whitelist	no/yes	no/yes
active business test	yes	yes

Table 2.20: CFC rules in New Zealand over time

CFC rules in Norway

CFC rules in Norway (Art. 10-68 of Norwegian Tax Code) exist since 1992. In the observed time period one bigger change in 2008 happened. A Norwegian entity, company or individual, holding interests in a foreign subsidiary may be taxed on the according share of the subsidiary's worldwide net profits by the Norwegian Income tax if the following requirements are fulfilled: (1) alone or together with other associated persons the Norwegian resident controls, directly or indirectly, at least 50% of the shares of the CFC, and (2), the income of the CFC is effectively low-taxed in the subsidiarys country (lower than two thirds of the Norwegian CIT). A special rule for crediting the paid foreign taxes against the taxes in Norway for the same profits does exist. Norway have a whitelist of countries all over the world which are generally excluded from the CFC rule, nevertheless, some types of income may be taxed by the CFC rule anyway. Additionally, a blacklist of deemed low-tax countries applies. Since 2008, the income of subsidiaries not in treaty countries only, but also in EEA countries may be exempt if a proof of real economic activity of this subsidiary can be provided.

	2004	2014
tax base	full inc	full inc
control	50%	50%
min tax abs	18.67%	18%
min tax rel	67%	67%
black-/ whitelist	yes	yes
active business test	none	yes

Table 2.21: CFC rules in Norway over time

CFC rules in Poland

There was no CFC rule in Poland in the observed time frame but since 2015.

CFC rules in Portugal

In 1995 a CFC rule (Art. 66 of the CIT code) was implemented in the Portuguese law. Pro rata after tax profits of a foreign subsidiary (distributed or not) may be attributed to Portuguese resident tax duties, holding substantial interests therein if this subsidiary is located in a low-tax jurisdiction. This rule applies for individuals and for companies similarly. A Portuguese entity has substantial interests if it (directly or indirectly) owns, (1) at least 25% of the subsidiarys capital, or (2), at least 10% of its capital where more than 50% of the overall capital is owned (directly or indirectly) by Portuguese resident entities. A country is deemed as a low-tax jurisdiction if, (1) it is included in the provided blacklist (consisting of 83 to 81 countries over the focused time frame), or (2), the CIT in the foreign country is 60% or lower than the Portuguese equivalent. In the CFC rule a tax credit for foreign payed taxes exists but not necessarily for all taxes paid by the CFC in the foreign country. If the following requirements are met the CFC rule will be exempt: (1) at least 75% of the profits are from local business activities, and (2), the main activity of the subsidiary is not of a financial nature. No actual whitelist of countries exists but, since 2012, the CFC rule is not applicable if (1) the subsidiary is located within the EU or EEA (with whom an information exchange agreement has been concluded), and (2), the subsidiary performs genuine economic activity.

	2004	2014
tax base	after tax inc	after tax inc
control	50%	50%
min tax abs	15%	15%
min tax rel	60%	60%
black-/ whitelist	yes/no	yes
active business test	yes	yes

Table 2.22: CFC rules in Portugal over time

CFC rules in Russia

There was no CFC rule in Russia during the observed time frame but since 2015.

CFC rules in Slovenia

No actual CFC rule exists in Slovenia. But some provisions in the CIT Code force to tax payments to persons in foreign low-tax countries for certain services like consulting, marketing, market research, human resources, administration, information and legal services. This applies if the effective tax rate in the foreign country is lower than 12.5% or the country is on a provided blacklist. In addition, payments of interests on loans are no deductible expenses for CIT purposes if the granted person is in a blacklist country.

CFC rules in South Africa

In South Africa a partial residence based tax system was effective as of 1998 and the CFC rule (section 9D of the South African Income Tax Act) exists in its current general form with full and not only partial income inclusion since 2001. In the observed time period one bigger change in the CFC legislation occurred in 2008. If a South African entity, individual or company, obtains income by foreign entities it may have to include the after tax pro rata part of this income into its own South African income tax base. Before 2008, the owner was affected if he held at least 10% of interests, and (together with others) controlled the foreign entity which was presumed if together 50% of shares where held. Additionally, the passive income of the foreign entity had to be more than 5% (save harbour threshold). If these requirements were fulfilled the full after tax income of the foreign entity was included in the domestic parent tax base. Since 2008, the foreign CIT rate has to be lower than 75% of the South African CIT rate which introduced the anti-abusive jurisdictional approach. In addition, since 2008 a South African resident would also be affected by the CFC rule if she holds 20% and more interests in the foreign entity alone, and the base of what was

counted as a foreign entity was extended to capture e.g. PEs as well. In the South African CFC legislation no black or whitelist exists but an active income exemption and some other specific exemptions more, mostly to exclude active income from CFC rule taxation. One main exemption used to circumvent the South African CFC legislation is the Foreign Business Entity exemption which has a complex structure itself.

	2004	2014
tax base	after tax inc	after tax inc
control	50%	50%
min tax abs	none	21%
min tax rel	none	75%
black-/ whitelist	no/yes	no/yes
active business test	none	yes

Table 2.23: CFC rules in South Africa over time

CFC rules in Spain

The CFC rule (Art 107 of CIT Code and Art. 91 of Personal Income Tax Code) in Spain exists since 1995. In the observed time period no bigger changes of the law occurred. If a Spanish entity, individual or company, obtains income by foreign entities it may have to include the passive pro rata part of this income into their own income tax base. A foreign company is deemed to be a CFC if: (1) the resident entity holds an interest of at least 50% in the subsidiary, and (2), the foreign effective tax paid by the subsidiary is less than 75% of the Spanish equivalent. The paid tax in Spain can be credited with the already paid tax in the foreign country by the subsidiary. Some exemptions are implemented for: (1) specific regulations for gains from a participation in another company of the CFC, or (2), passive income of the subsidiary lower than 15% of the total net profits, or 4% of the total turnover of the subsidiary, or (3), specific intracompany lending, or (4), income of a subsidiary in an EU country, provided genuine business activities. From this latter list in point (4) Luxembourg since 2006, and Cyprus and Gibraltar since 2007 are specifically excluded. In Spain no actual white- or blacklist of countries exists for CFC rule purposes.

1		
	2004	2014
tax base	pass inc	pass inc
control	50%	50%
min tax abs	26.25%	22.50%
min tax rel	75%	75%
black-/ whitelist	no/yes	no/yes
active business test	none	yes

Table 2.24: CFC rules in Spain over time

CFC rules in Sweden

CFC rules in Sweden (Art. 39a of Swedish Income Tax Code) exist since 1990. In the observed time period one bigger change happened in 2008. A Swedish entity, company or individual, holding interests in a foreign subsidiary may be taxed on its share of the subsidiary's worldwide net profits by the Swedish CIT if the following requirements are fulfilled: (1) low income taxation in the subsidiarys country (lower than 55% of the Swedish CIT), and (2), alone or together with other associated persons the Swedish resident controls, directly or indirectly, over at least 25% of the capital or voting rights. A special rule for crediting the paid foreign taxes against the taxes in Sweden for the same profits do exist. Sweden has a whitelist of countries all over the world which are generally excluded from the CFC rule, nevertheless some types of income may be taxed through the CFC rule anyway. No blacklist exists. Since 2008, the income from subsidiaries in EEA countries may be exempt if a proof of real economic activity of this subsidiary can be provided. The countries Belgium, Estonia, Ireland, Luxembourg and Netherlands are expressly excluded from this regulation. In the observed time period Cyprus was excluded from the whitelist between 2005 and 2007 as well.

	2004	2014
tax base	pass inc	pass inc
control	25%	25%
min tax abs	15.40%	12.10%
min tax rel	55%	55%
black-/ whitelist	none	none
active business test	none	yes

Table 2.25: CFC rules in Sweden over time

CFC rules in Turkey

Turkey implemented a CFC rule (Art. 7 of Corporate Tax Law) in 2007 without fundamental changes until the end of 2014. If a Turkish company obtains income by foreign entities, distributed or not, it may have to include the pro rata part of this income into its own CIT base. A foreign company is deemed to be a CFC if: (1) the resident entity holds an interest of at least 50% in the subsidiary, and (2), the foreign effective paid tax by the subsidiary is less than 10%, and (3), at least 25% of the CFC gross income is passive income. The paid tax in Turkey can be credited with the already paid tax in the foreign country by the subsidiary. For CFC total income below 100000 TRL (former YTL) a de-minimis exemption exists. These CFC rules do not count for individuals. In Turkey no white- or blacklist of countries exists.

20042014tax base none full inc control 50%none min tax abs 10%none min tax rel none 50%black-/ whitelist none none active business test none none

Table 2.26: CFC rules in Turkey over time

CFC rules in United Kingdom (UK)

The CFC legislation in UK (Art. 9A of Taxation Act 2010 (TIOPA)) was implemented first in 1984. A likewise taxation of individuals is included in another law. Due to a different structure than the other CFC laws and many special rules the British CFC law is one of the most complex ones (104 pages of tax law). The biggest change in the observed time frame was introduced with the Finance Act 2012 and the new rules apply since 2013. Before that the CFC rule applied if a shareholder of the UK entity (together with associated or connected persons) held an interest of at least 25% in the foreign subsidiary. If UK shareholders have an interest of more than 50% in the foreign subsidiary it counts as controlled.²⁴ As a second requirement, the subsidiary must be located in a low-tax country, meaning that the effective tax rate of the subsidiary is 25% lower than it would have been in the UK. If the rules apply to a company the resident shareholders have to pay a charge on the pro rata income of the subsidiary times the appropriate UK CIT rate. A few exemption methods are implemented. The CFC regime does not apply if: (1) at least 35% of the subsidiary are controlled by public, or (2), at least 90% of chargeable profits are distributed to UK shareholders, or (3), only real commercial transactions are executed, or (4), the subsidiary is in a foreign country for other reasons than tax reduction, or (5), the profits of the subsidiary are lower than 50,000 GBP (de-minimis exemption). The first mentioned exemption (public control) was canceled in 2007 and the second one (distribution to UK shareholders) in 2009. In 2007 an EEA exemption for subsidiaries in EEA countries was implemented due to the Cadberry-Schwepps-Case (European Court of Justice (2006)). Since 2013, the law is different at least on the practical law side, the given numbers from above are hardly different. A general foreign low-tax threshold does no longer exist, and various new exemptions assembled in a "Gateway Test" were introduced. This gateway test aims to filter out the non-artificial profits of business actions diverted from the UK. For simplification a pre-gateway test exists, to preserve obvious non-CFC rule affected subsidiaries from detailed reminder requirements. The new exemption rules are: (1) temporary exemption (12 month after becoming controlled from the UK for the first time), (2) low profits exemption (profits lower than 50000 GBP, (3) low profit margin exemption (accounting profits are lower than 10% of operating expenditure, (4 and 5) excluded territories exemption and tax exemption (as with the old low-tax threshold subsidiaries located in countries with a CIT rate higher than 75% of the UK's CIT rate are not concerned), or (6) some special finance company exemptions. Now, mainly passive income instead of all foreign income - in terms of this comparison - is taxed. A separated

 $^{^{24}}$ It also counts as controlled if two persons together control the subsidiary, provided that the non-UK resident controls at least 40% but not more than 55% and the UK resident controls at least 40%.

credit for foreign taxes exists for the different types of income. In the UK no actual blackor whitelist exists but a list of "no low-tax countries".

	2004	2014
tax base	full inc	pass inc
control	50%	50%
min tax abs	22.50%	none
min tax rel	75%	none
black-/ whitelist	no/yes	none
active business test	yes	yes

Table 2.27: CFC rules in United Kingdom over time

CFC rules in United States of America

The CFC rule ("subpart F" of Subtitle A chapter 1 subchapter N part III of USA Internal Revenue Code (title 26)) from the United States (U.S.) is the oldest so called CFC law worldwide and in force since 1962. In the observed time span no elementary amendment in the observed characteristics occurred. An entity of the U.S. (company or individual) who is a shareholder of a foreign entity may be taxed on specific undistributed pro rata income if three requirements are fulfilled: (1) the foreign entity is controlled by U.S. residents (together with other U.S. persons they have more than 50% of interests in the CFC), and (2), the concerned USA shareholders hold at least 10% of interests of the subsidiary (directly or indirectly), and (3), the effective tax rate for the CFC in the foreign country is less than 90% of the current highest U.S. corporate income tax rate. If these requirements are fulfilled, parts of the pro rata CFC income, namely "Subpart F Income" consisting of quasi passive income and some extras, must be included into the gross income of the U.S. resident. Various taxes payed by the subsidiary in the foreign country can be credited against the "Subpart F" taxes in the USA. Besides some very special exemptions, a de-minimis rule exists which exempts income below 1 million USD. A save harbour rule exempts the CFC legislation if the foreign passive income is below 5% of the foreign total turnover. If the passive income instead is above 70% of gross income a full inclusion of income results. No actual white- or blacklist exists. A rather bothersome issue are the so called 'check the box rules' which weaken the impact of the U.S. CFC rules quite heavily.²⁵ In November 2017 the 'Tax Cuts and Jobs Act' implemented the global intangible low-taxed income (GILTI) tax component which brought back the idea of CFC rules in the U.S. law.²⁶ No further discussion of this issue will be found here as this is outside the time scope of this study.

	2004	2014
tax base	quasi pass inc	quasi pass inc
control	50%	50%
min tax abs	36%	36%
min tax rel	90%	90%
black-/ whitelist	none	none
active business test	none	none

Table 2.28: CFC rules in United Kingdom over time

Some economical important countries without CFC rules until 2014

To highlight the fact that not all economical important countries, or countries concerning multinational taxation or profit shifting effects, have a law like a CFC rule enacted, a small list of these countries is given, which are included in the following empirical sections as well.

As of 2014 there are no CFC rules in: Belgium, Belarus, Bermuda, British Virgin Islands, Bulgaria, Cayman Islands, Chile, Croatia, Curacao, Cyprus, Czech Republic, Hong Kong, India, Ireland, Liechtenstein, Luxembourg, Malaysia, Netherlands, Netherlands Antilles, Panama, Poland, Romania, Russia, Seychelles, Singapore, Slovakia, Switzerland, Taiwan, Ukraine. This list is not exhaustive.

 $^{^{25}}$ e.g., Mutti and Grubert (2009); Field (2008-2009); Grubert and Altshuler (2006)

²⁶ e.g., Goldman, Lisa (2018); Avi-yonah and Mazzoni (2017)

Chapter 3

Profit Shifting & CFC Rules

At present, controlled foreign corporation (CFC) rules are one of the three main anti tax avoidance laws in developed countries. This paper examines the different CFC rule settings in the OECD as well as further countries to show their effects on profit shifting of multinational companies. Using an unique CFC law panel data set for 61 parent countries over 11 years and micro level firm data for more than 260,000 companies worldwide, I show that CFC rules lead to up to 19.1% less financial income in foreign low-tax subsidiaries. CFC rules comprise many distinct characteristics, upon which this study sheds light in order to better understand these laws and their influence on multinational's behaviour.

Using bunching at one of the CFC law given thresholds, I calculate shifted profits into foreign low-tax subsidiaries of around 11.67 billion USD due to tax avoidance as a conservative estimate over the observed time frame. Based on my findings, CFC rules seem to (1) constitute a bridge for territorial tax system countries to counteract profit shifting of multinationals, (2) draw a bottom line to these firm behaviors, and (3) tax shifted income parts in a purpose-built way in the home country.

This chapter is partly online available as the paper: Prettl (2018), Profit Shifting & Controlled Foreign Corporation Rules - The thin bridge between corporate tax systems, Working Paper (SSRN)

3.1 Introduction

Twenty-five years ago the profit shifting behavior of international companies was not a commonly discussed issue, while nowadays one can read every other week that some large corporation hid millions of profits somewhere to pay less taxes, in a newspaper (e.g., The Economist (2013)). The scandals of known tax havens and their networks of corporations in recent years have made clear that multinationals around the globe use varying techniques to circumvent stricter taxation laws whether to hide or shift profits to a foreign low-tax country.¹ It is known that this international firm behavior led to a race to the bottom for corporate income tax rates worldwide and made it harder for governments to gain revenue from these taxes (e.g., Overesch and Rincke (2011)). The perspective in general - for the country and the firms - could be different if the headquarters of the multinationals were operating in countries which generally tax worldwide profits or territorial profits only and exempts foreign ones from taxation.² In a very recent study by the European Commission, the losses are estimated at billions of euros or dollars per year respectively, which accounts for more than 10% of the corporate tax revenues of the EU, same goes for the US and Japan (Alvarez-Martinez et al. (2018)). To deal with these issues, many countries use different forms and scopes of antitax avoidance rules. Most-studied tools, like thin capitalization rules and transfer pricing rules, do not present an entirely satisfactory solution.³ Controlled foreign corporation (CFC) rules - the laws of interest in this work - are another important group of these counter measures, and so far only little is known about them. They could become a potential alternative, or rather a powerful addition, to tackle raising profit shifting. CFC rules attempt to tax earnings from foreign subsidiaries, repatriated or not, under specific circumstances, concerning both the owner and the subsidiary, if these earnings are likely shifted profits. But do these laws actually counteract the profit shifting behavior of multinationals?

¹ e.g., Jalan and Vaidyanathan (2017); Brown and Drake (2013); Faulkender and Smith (2016); Dyreng, Lindsey, et al. (2015); Gumpert, Hines, and Schnitzer (2016)

 $^{^{2}}$ e.g., Voget (2011); Feld, Ruf, et al. (2016)

³ e.g., Haufler and Runkel (2012); Buettner, Overesch, Schreiber, et al. (2012); Wamser (2014); Beer and Loeprick (2015); Ruf and Schindler (2015); Cristea and Nguyen (2016); Mardan (2017); Davies et al. (2018)

To answer this question, one has to understand the basics of these complex legislations first. CFC laws are, unlike thin capitalization laws or transfer pricing rules, more concerned about the whole structure of the multinational corporation rather than the specific methods of profit shifting. Therefore, they can be understood as a profit shifting barrier in the first place. And secondly, as a vessel to collect profits shifted abroad to tax emerging gains out of an ethically dubious practice to tackle base erosion and profit shifting (BEPS). Therefore, CFC laws usually aim at the so called 'passive income in foreign low-tax country subsidiaries', and especially at earnings which are raised out of patents, trademarks, royalties, interests or other financial incomes — the types of profits that are most easily shifted abroad to benefit from lower taxation rates. Therefore, a CFC rule generally works in the following way: If a multinational's foreign subsidiary fulfills certain requirements, then at least parts of its income are taxed in the multinationals parent country where the CFC rule is enacted, even if no repatriation takes place. In most cases, three factors are crucial: Low taxation in the foreign country, ownership requirements, and unusual financial income derived, for example, through unusual firm construction, mostly denoted as 'passive or tainted income'.⁴

On the one hand, by implementing CFC rules in a tax system, the parent country's government is able to tax income of the company's foreign subsidiaries. This constitutes a big exemption in a territorial tax system, which usually exempts foreign profits. Worldwide tax systems generally tax foreign profits on repatriation but provide credit for paid taxes.⁵ Nowadays, there are no longer real pure approaches of this tax system distinction.⁶ On the other hand, CFC rules are often intentionally constructed to set a lower bound of quasi allowed profit shifting by setting up "save harbour" thresholds. Through my research, I set out

⁴ Passive income has different definitions in different tax laws but all in all, and especially for the further pages of this survey, with "passive income" financial income derived from dividends, interest, rents, royalties and other capital gains is meant. The definitions of passive income of the different countries are not the same but rather similar, and for this comparison the specific details are not crucial.

⁵ Ironically, the only remaining big country using a worldwide tax system, the US, implemented a new law in 1996, which diminished nearly all effects of their CFC rules (Grubert and Altshuler (2006)); Resulting in huge accumulations of profits abroad that do not get repatriated and alter investments of US multinationals (Hanlon, Lester, and Verdi (2015) or Feld, Ruf, et al. (2016)).

⁶ e.g., Clausing (2015); Altshuler, Shay, and Toder (2015)

to show that CFC rules are the slim bridge between the approaches of taxing multinationals, worldwide or territorial tax systems, in our globalized world. Due to the aforementioned profit shifting actions of multinationals, implementing or strengthening such CFC rules can be one way for governments to react to BEPS. In recent years, the OECD implemented CFC rules as one big issue in their BEPS plan for important international taxation issues (OECD (2013)). Additionally, the EU Commission started the Anti Tax Avoidance Directive (ATAD)⁷ which, inter alia, impels members to constitute a CFC rule in their respective tax legislations by 2019.

At OECD's public consultation of strengthening CFC rules, it was lamented about the lack of research on these laws (OECD (2015b)). This is likely due to two main reasons: First, as one can see in this study and other work discussed below, these laws are very complex and, second, they currently lack importance in the United States mainly due to the so called "check the box rule", which allows American multinationals fairly easy to circumvent CFC rules.⁸ The latter one changed radically with the implementation of the 'Tax Cuts and Jobs Act' in November 2017, which introduced a global intangible low-taxed income (GILTI) tax component, by which multinationals are taxed (again) on specific foreign lowtax income.⁹ All in all, a broad study of these laws covering multiple countries over time is missing, yet crucial for a better understanding and policy recommendations. Therefore, this chapter examines CFC rules across all OECD as well as 25 additional countries of relevance in international business taxation from 2004 to 2014 to shed more light on these critical aspects. Using different estimation methods based on the micro-level panel data set ORBIS from Bureau van Dijk and a self-conducted unique panel data set on CFC rules for 61 countries (27 of them do have a CFC rule in at least one of the observed years), I explore the impact of CFC legislation on passive income abroad.

 $^{^7}$ Based on Directive (EU) 2016/1164. A recent critical review from the law perspective is provided by Hentschel and Moser (2017) or Koerver (2016).

⁸ Mutti and Grubert (2004); Grubert and Altshuler (2006)

⁹ e.g., Goldman, Lisa (2018); Avi-yonah and Mazzoni (2017)

This chapter presents new evidence that CFC rules curb profit shifting behavior significantly on a global scale. By compiling an extensive list of various CFC rules and their changes over time by hand, I am able to explore the influence of CFC laws and their characteristics in detail, and share more insightful results than the existing literature so far. Therefore, I can demonstrate the importance of these anti-tax avoidance measures to counteract profit shifting behavior, substantiate my findings with detailed firm-level observations, and increase the scarce empirical knowledge about these increasingly important laws. Besides graphical evidence for the behavior change of multinationals, I explore both between subsidiary cross-sectional variation and within-subsidiary time variation around two given thresholds by using a newly created theoretical framework and data of more than 350,000 subsidiary-year observations.

Due to the complexity of CFC legislation, this study aims to provide a better understanding of the different influences these laws can have on decisions of multinationals, theoretically and empirically. Put differently, to answer questions about CFC rule specifics, it is important to get first a bigger picture of how these laws influence corporate behavior. Generally, CFC rules could have three main influences: (1) on profit shifting behavior, (2) on location choices of subsidiaries, and (3) on investment behavior. These three are, to some extent, interrelated. The primary focus of this study is the first issue, as the motivations behind profit shifting of multinationals are the behavior CFC rules are aiming at, but I provide insights into location choices and general investment behavior influences as well.

Building on a new and simple theoretical framework, the results of this study are presented in two settings. The first group of results suggests that a CFC legislation leads to at least around 19.1% less profit shifting behavior into foreign low-tax subsidiaries, which constitutes a relatively conservative lower bound. These results originate from fixed effects panel regressions, which are identified by many changes in quasi parent-subsidiary pair relations due to changes in corporate income tax rates and the researched low-tax threshold. This threshold is included in nearly all CFC rules and changes differently across countries over the time frame of 11 years. The results are robust to variation tests using a large range of different specifications. Additionally, I show that corporations react differently to various CFC rule characteristics like the specified tax base of the foreign subsidiary. In a next step it is shown that financial profits abroad are decreasing if the subsidiary is located in a low-tax country, determined by the CFC rules in the parent country, but increasing if the subsidiary is located in a country with a corporate income tax rate above this threshold. This indicates a clear influence of CFC rules on profit shifting behavior as well as on location choice, which could lead to more corporate tax revenue from a global perspective. Building on this evidence and the following findings, I conclude that CFC rules constitute a functional anti-tax avoidance law that counteracts profit shifting behavior of multinationals in a certain way. This substantiates the theoretical findings of Haufler, Mardan, and Schindler (2018), who show that CFC rules and thin capitalization rules should both play a role in an optimal tax mix, and the empirical work by Ruf and Weichenrieder (2012), which shows an actual influence of German CFC rules on foreign passive assets of multinationals based in Germany.

As shown in the underlying theory, the low-tax threshold only explains parts of the CFC rule influence on firm behavior and decision choices. Therefore, in the second part of this study, I examine how multinationals circumvent these laws. First, I present graphical evidence by an event study of the mentioned Cadburry-Schweppes-Case (European Court of Justice (2006)) decision influence on foreign low-tax profits. Second, I examine bunching at a so called passive-to-total income threshold, which can annul the low-tax threshold consequences for some firms, with strong identification. If this specification is part of a CFC rule, a company could still generate financial profits in a low-taxed subsidiary without being affected by the anti-tax avoidance legislation, as long as it is below this second threshold. As this results in large average tax rate differences on the two sides of this limit for the multinational, one can observe clear bunching. Therefore, the last part of this study shows details about the location- and time-varying effects of financial profits in foreign low-tax countries explained by this threshold. The results are generally in line with Egger and Wamser (2015)

who show FDI decision influences of German CFC rules on German multinationals. The study at hand uses the same kind of threshold but for various countries and profit shifting influences rather than foreign real investment decisions. The results from Ruf and Weichenrieder (2013) about the influences of the Cadbury-Schweppes-Case decision of the European Court of Justice (European Court of Justice (2006)) on German multinationals and foreign intangible assets are substantiated and discussed in a multi-country setting.

By researching these two thresholds, the aforementioned two main critical factors of CFC rules, that is low-taxed location and rise of passive income, are examined in detail, and a more holistic picture on CFC rules is provided. As explained, if the foreign low-tax subsidiary is generating enough real or active income, multinationals can still shift some profits into these subsidiaries if the passive-to-total income threshold of CFC rules is high enough. In a last step, by using the bunching results and a simple back of the envelope estimation, I am able to show that, during the observed time frame, at least 11.67 billion USD are shifted from CFC rule countries to lower taxed subsidiaries at the passive income threshold alone. This indicates that this passive-to-total income threshold is one important key, which currently seems to be set too high, at least for some constellations, to achieve better anti-BEPS results by the international government community. All together this demonstrates the complexity of CFC laws and emphasizes that, in order to show their impact on corporate behavior, multiple factors need to be taken into consideration. Thereby, this study shows the behavioral response of multinationals due to tax laws in many countries worldwide and extends the sparse literature in this area. It is also the first study which shows corporate tax related bunching for large companies on a global scale using uncommon, non-administrative data. Recently, Zwick and Mahon (2017) show bunching of national companies in equipment investment due to tax incentives, and Almunia and Lopez-Rodriguez (2018) show that sizedependent tax enforcement in Spain leads to clear bunching below the given threshold to avoid stricter tax enforcement.

These results could have important implications for policy, not only for the details of the legislation but also for the very functionality of these laws. If, for example, the United States implemened stricter CFC rules again or got rid of the 'check-the-box' system, what they are currently working towards with the GILTI rules, they would set a boundary and become capital export neutral again,¹⁰ and thus be able to tax millions of shifted profits abroad without disturbing the welfare increasing active investment behavior in foreign countries. Thus, the incentives of American companies to shift profits abroad instead of investing them in their own country would be reduced. My research also shows compelling evidence that CFC rules seem to constitute a proper way to set a boundary for profit shifting behavior of multinationals and ultimately their owners. The ATAD of the European Commission from June 2016 led to a guideline that induces member states to apply CFC rules and other measures beginning on January 1st 2019. Given the results of this study, countries should be persuaded that decent CFC rules are a functional way to tackle profit shifting around the globe.

After an overview of additional literature on profit shifting issues in Section 3.2, Section 3.3 explains the theoretical framework, while Section 3.4 describes the empirical methodology and Section 3.5 provides information about data and data sources. Section 3.6 shows the results and Section 3.7 concludes.

3.2 Literature on profit shifting

As Section 2.3 already provided published literature mainly about CFC rules, this section will provide insights into the literature and the more general discourse in the multinational profit shifting area. An often cited literature review on this topic is the one by Dharmapala (2014), which highlights and overviews literature about base erosion and profit shifting until 2014. Not only since then hundreds of studies are written about multinationals that shift profits

 $^{^{10}}$ For the background and importance of capital export neutrality see e.g., Richman (1963); Musgrave (1969); Grubert and Mutti (1995); Egger, Merlo, et al. (2015)

to circumvent higher taxation. It is obvious that this section will provide a glimpse into this field only. First, I want to mention that I will not comment on the huge discussion on the alternative taxing method of destination based cash flow tax, (re-)initiated by studies from A. Auerbach and Devereux (2015); A. J. Auerbach, Keen, and Vella (2017).

In a recent working paper, Alvarez-Martinez et al. (2018) estimate the profit shifting and base erosion amount of lost corporate tax revenue from a macro perspective of around 36 billion annually or 7.7% of total corporate tax revenues of the EU. They also provide high numbers for the US and Japan. Maybe even more important for the study at hand, their results suggest that eliminating profit shifting opportunities would slightly reduce investment and GDP but would raise corporate tax revenues due to enhanced domestic production. This in turn could reduce other taxes and increase welfare.

There are various channels through which multinationals shift their earned profits as well as services into low-tax countries. One main channel is through transfer mis-pricing of goods sold cross-border, but intra-company-wise. If multinationals set prices which are not following the arms-length principle, which specifies that goods should be priced the same as between unrelated companies, they can use these differences for pre-tax income shifting. Using the same data base as the study at hand, the work by Beer and Loeprick (2015) shows that the heterogeneity between multinationals is not negligible, and that specifically complex and intangible asset rich multinationals are utilizing tax rate differentials for profit shifting. Additionally, the study shows that transfer pricing documentation requirements reduce profit shifting, however they are not able to find significant results for the former identified high intangible asset multinationals. Using a self-created index of transfer pricing regulations, Marques and Pinho (2016) show that stricter laws lead to less CIT rate differential sensitivity. Cristea and Nguyen (2016) provide evidence of profit shifting via transfer pricing using Danish firm-level data with reduced unit values between 5.7 to 9.1 percent. Davies et al. (2018) instead, using French firm-level data on arms-length and intra-firm export prices, provide evidence on the key role of tax heavens for tax avoiding profit shifting behaviour via transfer mis-pricing.

Another way of shifting profits is through intra-firm debt financing or lending. The transfer pricing channel and the debt financing channel are both researched by Saunders-Scott (2015), who finds that multinationals treat the profit shifting channels of intra-company debt and transfer mis-pricing rather as substitutes. Also researching both, transfer pricing rules and thin-capitalization rules, the study from Buettner, Overesch, and Wamser (2018) shows influences of these anti tax avoidance rules on real FDI and employment in foreign subsidiaries. They find negative effects of thin-capitalization rules for employment and real FDI for high tax country subsidiaries but no effects for transfer pricing rules on these variables.

Solely focusing on thin capitalization rules, the counter measures against profit shifting via debt financing by setting limits of the amount that a company can claim as a tax deduction on interest, are the following studies. Buettner, Overesch, Schreiber, et al. (2012) and Buettner, Overesch, and Wamser (2016) present evidence for the effectiveness of thin capitalization rules against internal debt shifting in the international profit shifting context based on firm-level data about German multinationals. The study from Blouin et al. (2014), creating an index of thin capitalization rules, shows that subsidiaries of US multinationals tend to rely less on internal debt financing in host countries with tighter thin-capitalization rules. Merlo, Riedel, and Wamser (2015) provide various results for the influence of thin capitalization rules on location choices of multinationals. Ruf and Schindler (2015) summarizes economic effects of thin capitalization rules including literature, theoretical and empirical, about the influence on German multinationals. Later the authors discuss their findings and experience of Nordic countries with thin capitalization rules and conclude that these laws or rather the underlying arms length principle is administratively too costly and impracticable. As a potential alternative, the authors mention CFC rules. The study at hand confirms their thoughts about CFC rules as a promising avenue for limiting internal debt shifting, as I show that CFC rules limit foreign financial income and provide empirical evidence.

One implication of profit shifting is that multinationals tend to choose different locations than they would have done otherwise. Buettner and Ruf (2007) show German multinationals' location decision influences caused by cross border CIT rate differences using fixed-effects logit models for probability estimation. Similarly, the research by Barrios et al. (2012) explains the location choice in a more general setting across countries and, specifically, shows that the decision of an incorporation responds to host and parent country taxes. Both taxes do have a negative impact on the location choice of new foreign subsidiaries.

Various other perspectives of profit shifting are taken by the following studies. Beuselinck, Deloof, and Vanstraelen (2015) show profit shifting of multinationals in general and the intensifying role of weak tax enforcement by countries' governments. A different technique of profit shifting is examined by Simone and Sansing (2018). They analyze income shifting by means of cost sharing agreements with foreign affiliates and explain the major effects when this technique is used. A further perspective is taken by Simone, Klassen, and Seidman (2017), who show that under certain circumstances specifically loss-making firms abroad are used to shift profits. Accordingly, these firms change the reported profits to pay less taxes in total. J. H. Heckemeyer and Overesch (2017) provide a quantitative review of the empirical literature on profit-shifting behaviour of multinational firms. The result of their meta analysis is a tax semi-elasticity of subsidiary pretax profits of about 0.8, which translates into an 0.8% increase in subsidiary pre-tax profits if the tax rate differential is one percentage point smaller. Another literature review is taken by Riedel (2018), which analyses and summarizes literature that deals with the quantitative importance of tax avoidance behaviour of multinationals using income shifting from high- to low-tax countries. Many of the mentioned studies exploit CIT rate differentials. Alexander, Vito, and Jacob (2016) show that these tax differences are indeed causal for profit-shifting behaviour of multinationals, but that recent tax base broadening reforms like the analyzed CFC rules in the study at hand, counteracted this attitude to some extend.

There is already a lot of research on profit shifting, multinationals and the influence of taxes, yet many more studies will follow due to the reasons given above and below. The dissertation at hand is a part of this ongoing research and answers a few questions about the influence of CFC rules on the behaviour of multinationals.

3.3 Conceptual framework

As laid out in the first chapter of this work, CFC rules are mostly constructed around two different important characteristics: the minimum tax threshold as a sign for being located in a low-tax country, and the passive-to-total income threshold for questioning if the profits raised abroad are out of active business or primarily of passive nature. Also, from the law perspective, this is the main distinction - jurisdictional vs. transactional approach.¹¹ Of course, the foreign subsidiary has to be controlled by the parent, which usually is the case if more than 50% of interests in the foreign entity are held by the parent. The ownership requirement is not researched in this study, except in some empirical robustness checks. Therefore, this requirement is excluded in the following.

Profit shifting decision of multinational



Figure 3.1: Multinational's choice set of shifting profits abroad

As seen in Figure 3.1 the multinational firm has, in general, four different options to use foreign subsidiaries as profit shifting vehicles.¹² At first, the firm can choose between (a) using already set-up foreign subsidiaries, or (b) building/buying new foreign subsidiaries.

 $^{^{11}}$ Further details are shown in Section 2.1

 $^{^{12}}$ Obviously, a so far national firm can only choose between the two options of new foreign subsidiaries on the right arm (b) of Figure 3.1.
The right arm of the decision tree is, most likely, associated with higher fixed costs to set up the new foreign entity. At the second level, the multinational can again choose between two options: (1) shifting profits into a subsidiary located in a country with a CIT rate below the minimum low-tax rate threshold, or (2) shifting into a country above that low-tax threshold. Option (1) means a potentially lower tax rate and, therefore, offers more profit shifting options. This technique is, however, limited. As explained above, the passive-to-total income threshold allows passive profits and therefore potential profit shifting, up to a certain ratio of the total or real income in the same subsidiary. Therefore, in this case, additionally to the shifting process, real income has to be generated abroad. For profit shifting option (2), shifting into a country with a CIT rate higher than the low-tax threshold, none of this has to be considered, but tax savings can still be generated. The results in Section 3.6.1 give empirical evidence for decision (2) and the influence of the low-tax threshold on multinational profit shifting behaviour. Section 3.6.2.2 is looking into details about choice (1), shifting into countries with CIT rate below the low-tax threshold and the influence of the passive-to-total income threshold.

For theoretical reflections, consider a high-tax country A and two lower-tax countries Band C with CIT rates $\tau_A > \tau_C > \tau_{low-tax-threshold} > \tau_B$, so that a subsidiary in B would be option (1), i.e. below the threshold, and a subsidiary in C option (2), i.e. above the threshold. For now, focusing on the left arm (a) of Figure 3.1 a net of after-tax profit maximizing company has its headquarters in country A with enacted CFC rules and owns a subsidiary in country B and C. All three of them have standardized production and cost functions, depending on the capital used in the respective firms.

$$A: f(K_A) - c_A(K_A) \tag{3.1}$$

$$B: f(K_B) - c_B(K_B) \tag{3.2}$$

$$C: f(K_C) - c_C(K_C) \tag{3.3}$$

As described above, the multinational is able to shift a portion of its passive profits Φ from country A into its foreign subsidiaries in B and/or C. Therefore the profits without taxation are calculated as:

$$A: \ \pi_A = [f(K_A) - c_A(K_A)] \cdot \ (1 - \Phi)$$
(3.4)

$$B: \ \pi_B = f(K_B) - c_B(K_B) + \ \alpha \ \Phi \left[f(K_A) - c_A(K_A) \right]$$
(3.5)

$$C: \ \pi_C = f(K_C) - c_C(K_C) + \ (1 - \alpha) \ \Phi \left[f(K_A) - c_A(K_A) \right]$$
(3.6)

with the first part in B, $f(K_B) - c_B(K_B)$, as real income R and the last part, $\Phi[f(K_A) - c_A(K_A)]$, as the shifted income S, and α as the shifting fraction of income which is shifted into country B. The reader should keep in mind, that the mean of shifting is not relevant for CFC rules. The crucial requirement is the occurrence of (too much) passive income abroad and not how these resources got there. It is assumed that $\Phi < 1$ as at least some profits have to be active in the home country, and the 'passive' nature is determined by the treatment in the according subsidiary corresponding to the specific CFC law. Due to the passive-to-total income threshold in CFC rules, the effective resulting tax rates for the profits in B depend on the ratio between real income and shifted passive profits Φ into that subsidiary. The tax rate in B is given by:

$$Tax \ rate = \begin{cases} \tau_B & if \quad R \ge \gamma S \\ \tau_B^* & if \quad R < \gamma S \end{cases}$$
(3.7)

with $\tau_B^* > \tau_B$, and usually even $\tau_B^* = \tau_A > \tau_B$. γ stands for the actual legal passive-tototal income threshold rate. Besides reasons for an administrative overload of including all foreign passive incomes or very small fractions of passive incomes into the home tax base, and resulting de-minimis or save harbour rulings, there is no other obvious economical reason why γ should not be zero or very small.¹³ The capital a multinational can use to allocate between the two facilities is exogenously given and sums up to: $\bar{K} = K_A + K_B + K_C$.

If the multinational decides to shift profits to a foreign low-tax subsidiary some convex shifting costs $C_{\Phi K_A}$ occur, which one could think of as costs for setting up the paper work for hiding or explaining the ongoing shifting to the local government. For simplicity, it is assumed that these costs are not deductible, reflecting for example non-deductible efforts of the owner, and are the same for shifting into both country B and C. Additionally, one should think of these costs as (1) increasing in, and (2) depending on the heterogeneous ability to shift financial profits out of the home country capital K_A (with $C'_{\Phi K_A}, C''_{\Phi K_A} > 0$). Formalized, including taxes and the costs of shifting $C_{\Phi K_A}$, the maximization of the profit shifting problem for the left choice arm (a) of Figure 3.1 is:¹⁴

$$max \Pi_0 = \Pi_A + \Pi_B + \Pi_C - T_A - T_B - T_C - C_{\Phi K_A} , \qquad (3.8)$$

where T_A , T_B and T_C are the CIT amounts paid in countries A, B and C, respectively. The tax base solely depends on the invested (and shifted) capital K_A , K_B and K_C .¹⁵ As one can see, even the solution of only one choice arm of the optimization choice problem – depicted in Figure 3.1 – becomes quite complex. To solve the right arm from Figure 3.1 instead, the solution would have to include fixed costs to account for the new subsidiary abroad. Bringing them both together would give a theoretical ground for profound empirical location choice results. Therefore, this study does not try to explain the general location choice question, but rather shows the different influences of the characteristics of CFC rules on profit shifting behaviour of multinationals, in order to get a better and broader understanding

 $^{^{13}}$ In fact, it seems that after the Cadbury-Schweppes-Case of the ECJ many countries in the European Union, and later others more, increased their passive-to-total income thresholds and/or active business test guidelines up to 50% from smaller ratios before. This is also shown by Bräutigam, Spengel, and Streif (2017).

¹⁴ Note that these considerations still hold if there is a chain of subsidiaries located in different countries because CFC rules are usually enforced through this kind of ownership chains. $C_{\Phi K_A}$ is attributed to the home entity in country A, which is in practice not necessarily the case. As they are assumed to be not tax deductible, this does not matter in this simplified setting.

¹⁵ This concludes in a tax on capital rather than a tax on profits. But in the case of shifted profits, which are determined by the used capital, this results in the same conclusion.

of these laws and their specifics concerning profit shifting.¹⁶ Nevertheless, from this first theoretical exercise, it becomes already clear why the following of this chapter is split into two perspectives, depending on the two main thresholds of CFC laws. Therefore, in the following, these two aspects are mainly handled separately to show their implications in a feasible way, but brought together selectively to see overall conclusions as well.

Minimum low-tax threshold considerations

For the first part of the analysis about the low-tax threshold, no further detailed theoretical framework is needed. For now, only profit shifting options into country C, a foreign country with a CIT rate above the low-tax threshold but below the own home CIT rate (option (2) from above so that $\alpha = 0$), exist. Without further math one can easily see the multinational's decision of shifting profits into foreign subsidiaries. If the profits occur in a subsidiary located in a foreign country with a CIT rate above the minimum low-tax threshold, no additional taxation results from CFC legislation. In contrast, if the foreign CIT rate is below the threshold of the according CFC rule, at least some of the profits abroad would be taxed at the home country CIT rate and the overall effective tax rate of the multinational would be higher, provided that the other CFC rule requirements are fulfilled as well, as mentioned above.¹⁷ It is easy to see, that it could be of advantage from a business tax accountant perspective to shift profits into a subsidiary which is located (slightly) above the tax threshold, as the overall average tax rate for that multinational would be consequently lower. This leads to:

¹⁶ One could think of the decision of opening a new subsidiary in a low-tax country as a first step in a two-step decision process, where the second step is determined by the outcome of the calculations around the passive-to-total income threshold. In further research, it would be interesting to see where additional theoretical results, building on this base, would lead to. The same would be true for low-tax threshold effects from the first distinction in the decision tree above: if an already existing or a new subsidiary is used as a profit shifting vehicle.

¹⁷ In the following regressions in Section 3.6.1 these additional requirements are taken into account gradually, and are commented in the explanation of the empirical part.

Conjecture 1. Financial profits of multinational companies are higher in foreign subsidiaries which are located in countries with CIT rates above the low-tax threshold of CFC rules, as profits can be shifted there without further taxation by CFC laws.

Passive-to-total income threshold considerations

The impact of the second important threshold analyzed in this study, the passive-to-total income threshold, i.e. option (1) from Figure 3.1, is shown by utilizing bunching methods and building on, or rather extending, other recent work from the bunching literature.¹⁸ In contrast to these studies, I do not present any tax elasticities, for various reasons. First, the meaning and interpretation of the resulting elasticities in this context are questionable as, e.g.: the strict assumptions, like equal marginal tax rates for all firms or the questionable functional form for the heterogeneity density, are not satisfied in this setting; I have nonadministrative data only which lacks of accuracy, and these elasticities would be on a very aggregated level, but stem from rather country specific and widely varying thresholds, which are effecting firms differently due to further conditions. Second, as the observed firms can easily manipulate their books, the so far used methods for local treatments would not work properly.¹⁹ Third, in very recent studies, it is shown that the used methods of elasticity calculation from bunching have their flaws, and that the conclusive effect of the results becomes less clear (e.g., Blomquist and Newey (2017)). The study at hand rather uses bunching to identify the impact of CFC rules and shows their causality for multinational profit shifting behavior at the given discontinuities in the choice sets of firms. Additionally, the bunching mass is used to calculate most likely shifted profits into foreign low-tax countries.

Building on the theoretical thoughts from above, if one ignores for the next exercise the option to shift profits into a country with a tax rate above the low-tax threshold, i.e. option

 $^{^{18}}$ Saez (2010); Chetty et al. (2011); Kleven and Waseem (2013); Slemrod (2013); Devereux, Liu, and Loretz (2014)

¹⁹ In fact, subsidiaries which bunch could come from every point of the relative income distribution.

(2), country C drops out and Equation 3.8 gets simplified. Accordingly, the overall marginal tax rate for the financial (and possible to shift) profits is:

$$\tau = (1 - \Phi)\tau_A + \Phi \tau_B \tag{3.9}$$

Now the multinational chooses both K_A and Φ to maximize the profit: $\pi = \pi_A + \pi_B - T_A - T_B - C_{\Phi K_A}$, where T_A is the CIT amount paid in A and vice versa for B. The first order conditions are:²⁰

$$f_A' = c_A' + \tau + C'_{\Phi K_A} , \qquad (3.10)$$

$$C'_{\Phi K_A} = \tau_A - \tau_B \tag{3.11}$$

The first expression is the normal marginal condition - marginal output value equals marginal cost - incorporating the cost of shifting income. Expression 3.11 indicates that the firm will shift profits up to the point when the gain from that technique – i.e. the tax benefits – equals the marginal costs of shifting. Therefore, by choosing the level of passive profits that occur abroad, the multinational implicitly chooses the level of overall firm profit taxation. Thus, one could think of these thresholds as notches rather than kinks as not the marginal but the overall average tax rate is changing discontinuously.

For thoroughness reasons, the elasticity of taxable income, in this case corporate income, could be derived as in Appendix B.2.

In the context of this subsection, the shifting decision of the tax minimizing multinational depends on γ from Equation 3.7. The multinational would try to shift as many profits abroad as possible to maximize the tax differential savings. If the CFC law allows passive income in the foreign low-tax subsidiary up to a certain ratio of the total foreign income, one would expect bunching at this threshold as the effective tax rate difference at this point can be

 $^{^{20}}$ Mathematical proofs are in Appendix B.1. The CIT payed abroad by the subsidiary are usually credited in the CFC rule country and, therefore, cancel out.

substantial. The costs of shifting could include further activities abroad to generate more active income to, ultimately, generate more foreign passive income or rather shift more profits abroad.²¹ Therefore, I propose:

Conjecture 2. Foreign subsidiaries of multinational companies with positive financial income are bunching at the given passive-to-total income thresholds of CFC laws if the tax benefits are high enough, as the effective CIT rate for the shifted profits might be much lower for income ratios right below the threshold.

3.4 Empirical methodology

Minimum low-tax threshold - fixed effects panel regressions

For this part of the result section, I utilize panel fixed effects estimation methods, including subsidiary firm and parent firm fixed effects, to identify the average causal effect of CFC rules enacted in the parent country on profit shifting behavior of multinationals into foreign (low-tax) subsidiaries. Due to the rich informed panel data set ORBIS, which is discussed further below, I observe within-firm over-time and cross-country effects. I provide evidence in favor of all theoretical predictions made as well as several robustness checks.

All underlying theoretical assumptions seem to be fulfilled and are tested in robustness checks. Due to the firm and parent fixed effects modelling, specific firm and/or parent firm conditions are controlled for. To analyze the effects of the minimum tax threshold of CFC rules on multinational profit shifting behavior, I use a semi-log linearised model of the following form for subsidiary s, in foreign country f, with an owner o in its home country h,

 $^{^{21}}$ Further theoretical thoughts in this direction are shown by Weichenrieder (1996).

and (later) an intermediate firm located in a different country i, in year t:

$$ln(financial profits)_{sfoht} = \alpha + \beta_1 CFC_{fht} + \beta_2 X_{ft} + \beta_3 X_{ht} + \beta_4 X_{st} + \beta_5 X_{it} + \beta_6 \Gamma_t + \beta_7 \Gamma_i + \epsilon_{sfoht}.$$
(3.12)

The main variable of interest CFC, an indicator which is equal to one if a subsidiary is located in a low-tax country concerning the CFC rules of the owner country h and its influence β_1 , are explained further below. In both countries, I control for different factors with appropriate and frequently used control variables. I include X_{ft} to control for timevarying effects in the *foreign subsidiary country* such as the CIT rate²², GDP per capita, unemployment rate and corruption, and X_{ht} to control for the corresponding effects in the *home country of the owner*. Also, to control for subsidiary specific time-varying effects like the variability of financial reports, I use X_{st} for tangible assets, sales, operating profits or others. Γ_t controls for time fixed effects and Γ_i for fixed effects of intermediate countries between the owner and subsidiary in the robustness checks, as X_{it} does for time varying effects. The selection of control variables is based on plausibility considerations and previous mentioned studies on international taxation and investments or profit shifting abroad.²³

There might be other variables which potentially influence the dependent variables used in foreign subsidiaries. As other eminent literature, I assume that these are either controlled for (for instance in the fixed effects) or are uncorrelated with – in my case – CFC law regulations and their specific changes over time and are therefore included in the error term. By using this method, I identify my results by cross-section and cross-time differences in a

 $^{^{22}}$ I do not use effective average tax rates of firms at the beginning, as intra-firm behavior and the taxation of profits is the main purpose this study is interested in. However, in different regressions later on I use effective average (country, not firm) tax rates to show certain aspects of CFC laws as these rates can differ significantly from statutory rates (e.g. White (2013)).

 $^{^{23}}$ Additionally, I tested for various other parameters like common language, distance or more, and the results stayed more or less the same. The results of these regressions can be delivered on request. One should keep in mind that specific interdependent factors between countries are controlled for in nearly all regressions by the used fixed effects already.

panel data set. As described above and in Section 2.2 there are many changes in the absolute and relative low-tax thresholds and country CIT rates in the various country pairs over time. A stylized variation overview of possible changes in the CFC indicator variable for a single subsidiary-parent-year observation is given in Figure C.1 in Appendix C. All together, this results in a valid identification strategy for the purpose of this study. Additionally, and to verify my results, I add two more CFC rule characteristics – effective average CIT rates instead of statutory CIT rates, and the distinction between different tax bases²⁴ – to further baseline regressions, to narrow down the identified subsidiaries by their different CFC rule effects and show the distinct influence of these characteristics.

Throughout the main regressions, CFC rules are expected to influence the profit shifting behaviour of multinationals if the foreign subsidiary is located in a low-tax country, constituted by the relevant CFC law. Specifically, the foreign subsidiary is deemed to be a CFC if the statutory CIT rate in foreign country f is below the minimum low-tax rate threshold of the specific CFC rule of the parent in home country h in that year. Hence, the main variable of interest is constructed as

$$CFC = \begin{cases} 1 & \text{if } \tau_{ht}^{threshold} > \tau_{ft} \text{ or country } h \text{ applies CFC} \\ & \text{rules without a tax rate threshold} \\ 0 & \text{otherwise} \end{cases}$$
(3.13)

where $\tau_{ht}^{threshold}$ is the tax rate threshold of the CFC rule of the parent home country hand τ_{ft} is the statutory CIT rate in the foreign subsidiary country f in year t. The $\tau_{ht}^{threshold}$ of one parent country is the same for all foreign countries, at least for one year. As stated, in some regressions average effective CIT rates are used for refinement instead of statutory ones.

In further baseline results, as well as in robustness checks, this specification is refined as follows. To account for affected foreign tax bases, another CFC rule characteristic, the

 $^{^{24}}$ In Section 2.2 is a short description of the differences.

variable of interest is split up into three different variables to show the distinct influences: The first specific variable only equals 1 if the owner country CFC rules account for all (active and passive) income as tax base. The same applies, vice versa, for CFC rules of parent countries, which only account for passive income or some income measures in between these two. To show further differences of profit shifting multinationals and specific CFC rule influences on profit shifting location choices, the group of subsidiaries located in foreign countries with CIT rate above the CFC rule low-tax threshold is split up in two groups: one accounts for subsidiaries above the threshold but below the home CIT rate of the parent and the second group accounts for subsidiaries located in foreign countries with CIT rates. This constructs as:

$$CFC \ Below \ threshold = \begin{cases} 1 & \text{if } \tau_{ht}^{threshold} > \tau_{ft} \text{ or country } h \text{ applies CFC} \\ & \text{rules without a tax rate threshold AND } \tau_{ht} > \tau_{ft} \\ 0 & \text{otherwise} \end{cases}$$

$$Above \ threshold = \begin{cases} 1 & \text{if } \tau_{ht}^{threshold} < \tau_{ft} \text{ AND } \tau_{ht} > \tau_{ft} \\ 0 & \text{otherwise} \end{cases}$$

$$Higher \ than \ own \ CIT = \begin{cases} 1 & \text{if } \tau_{ht} > \tau_{ft} \text{ and country } h \text{ applies CFC rules} \\ 0 & \text{otherwise} \end{cases}$$

$$(3.16)$$

Building on the former labeling, τ_{ht} stands for the statutory CIT rate in the owner's home country h. As a logical implication, this approach includes parent countries with existing CFC rules only. In a further robustness check the variable of interest can only become 1 if a thin capitalization rule exists in addition to the CFC rules in the owner's home country h. The variable is constructed as:

$$CFC \ \mathcal{C} Thin \ Cap = \begin{cases} 1 & \text{if } \tau_{ht}^{threshold} > \tau_{ft} \text{ or country } h \text{ applies CFC} \\ & \text{rules without a tax rate threshold} \\ & \text{AND thin capitalization rules are enacted in country } h \\ & 0 & \text{otherwise} \end{cases}$$
(3.17)

In a last distinction, I accounted for the size of companies. I split up the subsidiary group of probably affected CFC subsidiaries (former baseline group CFC) into three groups according to size: very large, large and medium. The variables are constructed as:

$$CFC \ very \ large = \begin{cases} 1 & \text{if } \tau_{ht}^{threshold} > \tau_{ft} \text{ or country } h \text{ applies CFC} \\ & \text{rules without a tax rate threshold AND } o \text{ is very large} & (3.18) \\ 0 & \text{otherwise} \\ \end{cases} \\ CFC \ large = \begin{cases} 1 & \text{if } \tau_{ht}^{threshold} > \tau_{ft} \text{ or country } h \text{ applies CFC} \\ & \text{rules without a tax rate threshold AND } o \text{ is large} & (3.19) \\ 0 & \text{otherwise} \\ \end{cases} \\ 0 & \text{otherwise} \end{cases} \\ CFC \ medium = \begin{cases} 1 & \text{if } \tau_{ht}^{threshold} > \tau_{ft} \text{ or country } h \text{ applies CFC} \\ & \text{rules without a tax rate threshold AND } o \text{ is large} & (3.19) \\ 0 & \text{otherwise} \\ \end{cases} \\ 0 & \text{otherwise} \end{cases}$$

In another regression, this holds for the size of subsidiary s instead of the owner's o size which gives further insights from a different perspective. In further robustness checks, no variation of the constructed variable of interest takes place and the specifications are explained in the according Section 3.6.1.

In this setting, one could think of three main selection into treatment or omitted variable bias problems. First, the parent headquarters could be moved in order to not be affected by CFC rules. Voget (2011) shows that CFC rules do have a positive effect on the probability of reallocation, comparing 140 headquarters that have been moved with 2000 that have not been moved over a period of 10 years. As described above, I am using parent fixed effects to controls for this potential bias. Additionally, this bias would drive my results towards zero. Second, subsidiaries which are above the low-tax threshold could be used as a new profit shifting vehicle if another subsidiary from the same parent would be affected by changed conditions. This could lead to an overestimation of the treatment effect. Therefore, I test for these issues in robustness checks, additionally to first results of this behaviour in the fourth baseline regression. Third, as shown in the theoretical framework section, the passive-to-total income threshold could be utilized as well. A deemed to be treated foreign subsidiary could be in fact unaffected by CFC taxation if it is below these thresholds. This specific issue is dealt with in robustness checks as well. As already mentioned, due to these and other common econometric caveats, the results of this part should be understood as a *lower bound* of the influence effect of CFC rules on profit shifting behaviour in low-tax subsidiaries.

Passive-to-Total income threshold - Bunching descriptives

In this part of the analysis, by applying the theoretical framework from above and using the quasi-experimental bunching approach, I first show clear bunching at the passive-tototal income threshold. This identifies a further important impact of CFC rules on profit shifting behavior of multinationals and gives reassuring results for Conjecture 2. At first, it is important to mention that there is no other obvious reason for subsidiaries to stay just below or exactly at the passive-to-total income threshold than tax avoidance incentives inflicted by CFC rules.²⁵ Building on the framework described above, I use a variation of the bunching estimation method proposed in Saez (2010); Chetty et al. (2011); Kleven (2016) to (1) show the influences of these laws in a non-parametrical way on an aggregated level and (2), calculate the amount of shifted profits into foreign low-tax countries.

In this context, the interesting relative threshold point follows from the increase in tax duties that occur if a multinational shifts "too much" profits into its foreign low-tax subsidiary, so that $R < \gamma S$ in Equation 3.7, meaning that the passive income abroad is above the allowed passive-to-total income threshold. In this case CFC rules are getting enforced, which leads to a tax rate increase from τ_1 to τ_2 , with $\tau_1 < \tau_2$ at the threshold K. Following the standard approach taken in the literature, I fit a flexible polynomial to the observed distribution, excluding data in a range around the threshold π^* , in this case mainly the 50% passive-to-total income ratio, and then extrapolate the fitted distribution to the threshold (see e.g., Kleven (2016)). In addition to this basic approach, I use alternatives with differences over time and between countries at the threshold level to identify the clear behavioral response of multinationals to these tax laws.²⁶ The results of these considerations are used to explain and discuss first descriptives and non-parametrical results. Additionally, I use the counterfactual calculation of the polynomial to explain the shifted amount of profits at this specific threshold. This is done by a simple back-of-the-envelope calculation explained in the according Section 3.6.2.3.

I do not observe a clear kink or notch, as for various CFC rules the according CIT rate jumps from the foreign CIT rate to the home CIT rate for *all foreign income* and for other CFC rules *for parts* of the foreign income only. Additionally, the firms which bunch could come from a broad range of the observed *income ratio*, as it is much easier for firms to modify these ratios than for individuals – see Section 3.3 for further explanations. Nevertheless, in general, one could think of this passive-to-total income ratio threshold as a notch setup, as

 $^{^{25}}$ As an ecdotal evidence I spoke with a lot of accountants at different universities and conferences and no one could think of a real other reason why someone should decide to aim for a specific income ratio of passive to total earnings. No one has shown contrary opinions.

²⁶ For completeness reasons further mathematical proofs are shown in Appendix B.2.

usually the full foreign (passive) income would be included in the tax base and, therefore, the discontinuity occurs at the average tax rate level, not the marginal one. In addition to the already mentioned reasons of aggregation and the simple book trimming ability of firms, no clear missing mass as in other bunching studies concerning notches is expected, due to the usage of non-administrative data.

3.5 A note on data and data sources

To obtain detailed information about CFC legislations worldwide, I screened the explicit laws of 27 countries and filtered out the main important CFC rule characteristics, as described in detail in Section 2.1. To broaden the base for this empirical analysis, I added 34 countries, which are important for economical and international business taxation related reasons, especially related to profit shifting behavior. These 61 countries (OECD and 26 more) are home to the parent companies considered in this study, and differ with regard to whether or not CFC rules are installed. In addition to the self-conducted data set on CFC rule characteristics, I used the ORBIS firm-level data by Bureau van Djik ranging from 2004 to 2014 to show the economic relations on a micro level. This database not only provides balance sheet data about firms around the world but also links their ownership status. I used all unconsolidated firms contained in this data set, which have a subsidiary in a foreign country that is owned with at least 50% of shares by a parent company in one of the 61 countries. There are certain caveats using the ORBIS data set. First, it is known that the ownership links in ORBIS are only reported for the last year of information - 2014 in this case. This means that potentially misidentified links could occur if the ownership changed in the observed time frame. As other studies, I still use the ORBIS database due to the fact that this problem should bias my results towards zero or insignificance only. Second, as this study researches profit shifting behavior, it is important to have a sufficient number of observations in so called 'tax havens'. The importance of this is shown for example by Desai, Foley, and Hines (2006); Gumpert, Hines, and Schnitzer (2016); Laffitte and Farid Toubal (2018). As one can see in Table B.1 in Appendix B the study at hand includes observations from tax havens like Bermuda, British Virgin Islands and Cayman Islands, but also from larger European tax havens like Luxembourg or the Netherlands (used e.g., by Laffitte and Farid Toubal (2018)). The reader should keep in mind that the included firms in the ORBIS database from these countries are scarce and, most probably, do not account for the full sample of firms located in these countries. The author is not aware of, or does not have access to, a data base which captures more comprehensive corporate data. Nonetheless, the large panel data in this study seems to be valid for the conducted research.

Based on the ORBIS data, the variable 'financial profits' abroad is the best proxy for passive income and for showing profit shifting behavior of multinationals. Additionally, it represents the specific target variable CFC rules are usually aiming for.²⁷ This study uses unconsolidated firm data only. As other mentioned studies, most regressions are using logarithmic variables and, therefore, in most cases only firms with positive financial profits are observed. This is relaxed in robustness checks. As in most of the observed CFC rules, banks and financial industry parents are excluded, they are excluded in the following calculations as well (further explanations are given in Section 2.1). In robustness checks, I used intangible assets as dependent variable as well, which gives more or less similar results. After matching parent companies²⁸ to their foreign subsidiaries²⁹ and merging the CFC legislation data and further control variables for the 61 parent countries, I end up with a total of 1,815,624 affiliateyear observations for 265,802 subsidiaries in foreign countries. Table B.1 in Appendix B lists subsidiary-year observations of the variables used by parent country. The control variables are chosen to fit other eminent literature in multinational taxation and profit shifting as e.g., Becker and Riedel (2012); Ruf and Weichenrieder (2012). Table B.3 in the appendix shows some descriptive statistics for the underlying data and variables used. In the baseline

 $^{^{27}}$ e.g., also used by Ruf and Weichenrieder (2012).

 $^{^{28}}$ With parent companies the global ultimate owner from the ORBIS data base are meant.

²⁹ Not only direct subsidiaries but also chain sub-subsidiaries are possible.

regressions for around 80,000 foreign subsidiaries, more than 5,000 changes in the variable used for identification occur, meaning that subsidiaries potentially gain or loose CFC status that often. As explained in Section 2.1, tax treaty effects and firms from the banking and finance sector are not included in the regressions.

CFC rules have not always been of as much interest as they are nowadays. Therefore, there are not as many "high quality data sources" available for those laws as in other areas. As the laws are rather complex, the gathering of trustable information was challenging. To get the best data available, I used different sources for one country-year. As sources, I mostly used the European Tax Handbooks from IBFD (2002-2016), various corporate tax guides (Ernst & Young (2004-2016); Deloitte (2015); KPMG (2003-2018)) and the specific legal texts of the countries as provided. For translation and help with the country-specific laws, I worked together with the subsidiaries of the German "Industrie- und Handelskammer" and law attorneys in the appropriate countries. In addition, I used various other annual tax bulletins and tax-change publications. For the before-mentioned data reasons, this analysis focuses on the 11 years from 2004 to 2014. The different characteristics of the various laws are described in Section 2.1 and Section 2.2.

3.6 Estimation results - Impact of CFC rule characteristics on profit shifting

As laid out above, CFC rules are quite complex. The literature expects that CFC rules, to some extend, have an impact on investment decisions of multinationals. As shown in this study, there is not *one* simple impact of a generally formulated CFC rule due to the complexity and levels of firm choices involved. Therefore, in this section, I show the impact of the different characteristics of CFC rules on profit shifting by, mostly, differences in foreign subsidiary's financial profits, the variable most CFC rules are aiming at, as it most clearly and detectable indicates profit shifting behaviour.

3.6.1 How CFC rules limit profit shifting behaviour of multinationals

As seen in Sections 3.3 and 3.4 one would expect a generally negative impact of CFC rules on financial profits abroad for specific subsidiaries. To examine this empirically, I run different regressions. All regressions are panel fixed effects estimations and, if not stated otherwise, include subsidiary, parent and year fixed effects to account for the underlying specifics.³⁰ All regressions are using winsorized observations at the 1% and 99% level of financial profits to exclude outliers. Besides some robustness checks, all regressions are clustered at the parentyear level and standard errors are robust to heteroskedasticity. Baseline results are presented in Table 3.1. One can see an expected significant negative impact of CFC rules on financial profits in low-tax subsidiaries. In Specification (1), the statutory CIT rate in the subsidiary country is used for determining this low-tax threshold, and the CFC indicator variable is 1 if this rate is below the allowed threshold in the corresponding CFC law in this year. This suggests a decrease in financial profits in these subsidiaries of around 15.2% compared to subsidiaries which are probably not affected by potential CFC legislations of their parents. Usually, it is not the statutory CIT rate, which determines whether a country is a low tax country, but rather the effectively paid CIT (rate). Therefore, from Specification (2) onward the effective average CIT rates of the subsidiary countries is used.³¹ Again, the indicator variable is 1 if this effective CIT rate is below the minimum low tax rate threshold of the corresponding CFC law and zero otherwise. Both indicators are highly significant but the effect in Specification (2) is larger and the standard error is smaller.

³⁰ These parent and subsidiary fixed effects automatically include industry specifics and, therefore, it is not necessary to provide industry fixed effects results. The results are showing the same significances. After submission of this dissertation the work on the single papers continued. Therefore, improvements were made, e.g. regarding the used fixed effects. To be more accurate, this amendment is included for the baseline regressions in Table 3.1 only. The other regressions contain old fixed effects as submitted and the improvement can be found in later versions of this part of the dissertation as single paper.

 $^{^{31}}$ As a proxy for firm specific effective CIT rates, the effective average (country not firm specific) CIT rate is used from the Oxford Centre of Business Taxation.

Dependent Variable			Log Financia	al Profits	
	(1) statutory CIT	(2) effective CIT	(3) different tax base	(4) excl. non-CFC sister subs	(5) split of non treated
$\begin{array}{c} \text{Cfc rule +} \\ \text{sub in low tax country } (CFC) \\ CFC \text{ but EATR} \end{array}$	-0.152^{***} (0.0310)	-0.260^{***} (0.0260)		-0.270^{***} (0.0387)	
CFC + All income base			-0.234^{***} (0.0307)		
CFC + Between all, and Passive income base			-0.314^{**} (0.152)		
CFC + Passive income base			-0.293^{***}		
CFC (Below threshold)			(0.0002)		-0.202***
Above threshold					(0.0504) 0.0732^{**} (0.0358)
Higher than own CIT					0.0418 (0.0269)
ln Tangible Fixed Assets	0.115^{***}	0.112^{***}	0.112^{***}	0.113^{***}	(0.0205) 0.113^{***} (0.00306)
CIT sub	(0.00302) 0.261 (0.177)	(0.00300)	(0.00300)	(0.00390)	(0.00390)
EATR sub	· · ·	-3.196^{***} (0.205)	-3.213^{***} (0.205)	-3.276^{***} (0.239)	-3.336^{***} (0.250)
ln Corruption sub	0.403^{***}	0.609^{***}	(0.609^{***})	(0.593^{***})	0.594^{***}
ln gdp sub	(0.0555) 2.386^{***} (0.254)	(0.0548) 2.560^{***} (0.272)	(0.0540) 2.585^{***} (0.272)	(0.0013) 2.799^{***} (0.315)	(0.0014) 2.819^{***} (0.317)
ln gdppc sub	(0.254) -1.108*** (0.258)	(0.272) -1.407*** (0.278)	(0.272) -1.430*** (0.270)	(0.310) -1.906*** (0.321)	(0.317) -1.927^{***} (0.323)
ln unemployment sub	(0.238) - 0.0864^{***} (0.0203)	(0.278) -0.0726^{***} (0.0208)	(0.279) -0.0724*** (0.0208)	(0.321) - 0.0535^{**} (0.0243)	(0.323) -0.0530^{**} (0.0243)
Observations # of potential CFC subsidiary obs Adjusted R-squared	$\begin{array}{r} 359,243 \\ 50,045 \\ 0.772 \end{array}$	$\begin{array}{r} 350,\!647 \\ 62,\!234 \\ 0.774 \end{array}$	$\begin{array}{r} 350,\!647 \\ 62,\!234 \\ 0.774 \end{array}$	$288,490 \\ 61,349 \\ 0.764$	$288,490 \\ 61,349 \\ 0.764$
Firm FE & Parent & Parent country x Year FE	YES	YES	YES	YES	YES

Table 3.1: Influence of enacted CFC rules and their characteristics on profit shifting

Clustered standard errors on parent-year level in parentheses *** p<0.01, ** p<0.05, * p<0.1Note: 'CIT' stands for statutory corporate income tax rate, 'EATR' for the effective average tax rate. From Specification (2) onward, effective average CIT rates are used to determine whether or not a subsidiary is located in a foreign low-tax country (by the CFC law of the parent country). In Specification (3), the probably treated subsidiaries are differentiated by the concerned tax base of their parents' CFC rules. In Regression (4), all high-tax country subsidiaries from CFC rule affected parents with subsidiaries in low-tax countries are excluded. In the last regression, (5), the non-treated subsidiaries of CFC rule parents are split into two groups additionally, i.e. higher or lower than the CIT rate of the parent country. All regressions are panel fixed effects estimations and winsorized at the observation levels of 1% and 99% of Financial Profits.

As the tax base of CFC laws differs between countries and over time, I distinguished the laws between 'passive income', 'all income' or 'something between passive and all income' with regard to the inclusion of foreign profits. This simple distinction stems from the various CFC laws. In Specification (3), the formerly observed low-tax country subsidiaries are split up into these three groups, according to the CFC rules enacted in their corresponding parent country. As one can see, the quasi aggregated value splits up quite differently. All three distinctions show significant negative effect but with varying size. If all income of the foreign subsidiary could be affected by the CFC rule, one can observe clear negative effects on financial profits abroad in Specification (3), with a significant decrease of 23.4% in foreign financial profits. If the passive income is concerned by CFC laws, the effect stays significantly negative for the subsidiaries in Specification (3), with a 29.3% decrease in financial profits abroad. For the incomes in between total or passive income only, one can observe a significant negative effect of 31.4% in Specification (3), however, the statistical significance is slightly less. It seems that subsidiaries, which are potentially effected by their passive income only, are more aggressively used for shifting or, put differently, parents from countries with CFC rules that would affect all income abroad are less risky and, therefore, less responsive to law changes.

So far, the control group consisted of all subsidiaries in the sample which are not directly affected by CFC rules. But one could think of the possibility that multinationals search for other ways to shift their profits abroad and to decrease overall tax payments. As pointed out above, one possibility to circumvent the CFC taxation into a low-tax country subsidiary could be to shift the profits in another foreign non-low tax subsidiary of the same parent instead. Therefore, statistically, this group of subsidiaries would also be considered as treated but in the opposite direction. As this could lead to an overstatement of the results, in Regression (4) subsidiaries which have a treated CFC affiliate are excluded. One can see that the coefficient is still significant and increased to 27% as expected. This observation leads to the following consideration.

In Specification (5), I show that financial profits of CFC rule affected multinationals accrue predominantly in foreign subsidiaries, which are located in countries with CIT rates higher than the low-tax threshold but lower than the own CIT rate of the parent company. Therefore, the subsidiaries located in countries with CIT rates above the low-tax threshold are divided into two groups: (i) above the low-tax threshold but below the parent country CIT rate, and (ii) above the parent's country CIT rate. One can still see the negative effect of CFC rules on financial profits in foreign low-tax countries of -20.2%, but the effects of the remaining two groups differ significantly. Subsidiaries of group (i) have 7.3% higher financial profits as they are not affected by CFC rule taxation. If the CIT rate of the subsidiary country is higher than the parent country CIT rate, group (ii), the financial profits are higher as well but statistically insignificant as shifting profits there does not lead to tax savings.³² Therefore, it seems that parents affected by CFC rules shift their profits as well, but into subsidiaries located in countries above the low-tax threshold. This behaviour still lowers the average effective tax rates companies would have to pay for these shifted profits and is legitimate. These baseline results proof Conjecture 1 of the theoretical part in Section 3.3 correct, as they show that multinationals choose to shift their profits preferably not into low-tax countries but into countries with CIT rates above the low-tax threshold, which accounts for option (2) in the decision tree in Figure 3.1. Decision choice (1) from the theoretical framework will be researched in more detail in Subsection 3.6.2.2.

The used control variables generally show the expected signs and are significant. On the firm level, tangible fixed assets in the subsidiary have a significant positive effect on financial profits, which seems plausible. Robustness tests in Table 3.3 for varying firm controls do not alter this result. The probably only unexpected sign is displayed by the statutory tax rate in the foreign subsidiary country. Here I observe more financial profits with higher subsidiary tax rates but statistical insignificance. As CFC rules concern effective taxes the used EATR measure from Specification (2) onward seems to provide a better control variable for this

 $^{^{32}}$ If the excluded affiliates from Regression (4), the 'CFC sister subsidiaries' located in countries above the low-tax threshold, are included again, the results stay nearly the same.

effect, it shows the expected sign and is highly statistically significant. Other reasons for this positive sign for CIT my be the ones observed in Specification (4) and (5). As expected, one can see higher financial profits with higher corruption and higher gdp, both with statistical significance. One can see less financial profits, with higher gdp per capita and with higher unemployment rates. In the context of profit shifting behavior of multinationals, there is no clear expected sign for these two control variables, as both ways could be plausible. In robustness checks one will see that the signs of these two coefficients will change or they become insignificant.

The same Regressions (1)-(3) & (4) are undertaken again but with time-variant parent country controls in Table B.4 in Appendix B.3. The overall picture stays very much the same and the added control variables are showing the expected signs with significant coefficients: Higher CIT rates in the parent countries lead to more profit shifting or more financial profits abroad, respectively, and higher corruption in the parent country leads to more profit shifting as well. No clear sign is expected for the *GDP*, *GDP per capita* or *unemployment* neither for profit shifting nor for foreign financial income. The results of the study at hand are in line with former findings from Ruf and Weichenrieder (2012) about the influence of German CFC rules on foreign investment behaviour, specifically, concerning their results about lower passive investments in foreign low-tax countries of CFC rule affected German multinationals.

Robustness checks and further insights

To show the robustness and causality of my results, variations of the baseline regressions are presented in this subsection. At first, I examined the used fixed effects and clustering conditions. All results in Table 3.2 proof the robustness of my baseline results, as the coefficient on the variable of interest stays negatively significant, and more or less at the same size: in Regression (1) a simple OLS regression is conducted, and in (2) no firm fixed effects are used, both resulting in negatively significant larger coefficients. In Regression (3), I used the most strict fixed effects possible and included parent-subsidiary fixed effects, which cancel out all time-invariant, link-specific effects within multinational companies, as well as home country-year fixed effects to eliminate all variation that could influence the results on the yearly parent country level. For these reasons, only observations of parents with at least two subsidiaries and two observations can be included, other observations are getting dropped automatically due to these strict conditions. Therefore, the observed number of subsidiaries abroad drops to around 156,000 observations. These fixed effects include any country by country fixed effects as well.³³

In the following regressions, I change the clustering from parent-year in the baseline results to (4) home country-year, (5) host country-year, and (6) parent level to observe if these variations of perspective drive my results. The significances of some control variables change but the variable of interest, CFC, stays significant, which provides trust for the causality of the effect. Even though I include parent fixed effects in most regressions, there could exist some other interfering effects from the parent home country. To check for this possibility, I included four macro control variables for the parent countries in Regression (7) to account for these possible influences, but the results stay the same. The statutory CIT rate in the parent country seems to affect financial profits abroad positively. With more corruption in the parent country, again, more financial profits accrue abroad, which seem to be the logically expected signs for both. Additionally, with increasing investment freedom, the foreign subsidiaries show more financial profits, which seems plausible as well. There is no clear expected sign for GDP in the parent country, but it seems that multinationals from smaller countries shift more profits or have more financial income abroad.³⁴ Other mentioned studies show that the CIT rate of the parent country could play an important role and alter behaviour in this setting. Therefore, I rerun all regressions presented in this

³³ As subsidiary firm, parent and also the used subsidiary-parent fixed effects include inherent industry fixed effects and this stricter fixed effects version shows no influence, no regression results for industry fixed effects are shown but are available upon request.

 $^{^{34}}$ The exact same regressions are undertaken again but with time variant parent country controls in Table B.4 in Appendix B.3. The overall picture stays very much the same and the further control variables are showing the expected signs with significant coefficients: Higher CIT rates in the parent countries lead to more profit shifting/ more financial profits abroad and higher corruption in the parent country leads to more profit shifting too.

study including this variable as well. The results stay nearly the same and are therefore not shown in the study at hand, but are available on request.

As shown in the theoretical part in Section 3.3, one could think of the possible selfselection of very subtle multinationals utilizing subsidiaries in countries with lower CIT rates than the allowed low-tax threshold in the according CFC rule, but with a lower passiveto-total income ratio than the second threshold, as a profit shifting vehicle. This could result in multinationals, that still shift their profits into foreign low-tax countries even if they have enough active income in these subsidiaries to stay below the prescribed ratio threshold.³⁵ Therefore, in Regression (8) of Table 3.2, I included another indicator variable for the second mentioned important 'passive-to-total income' threshold to account for this selection problem. This indicator variable is 1 if the parent country has a CFC rule enacted, both, parent and subsidiary, are located in the EEA, the passive-to-total income ratio is above 50%, the observed year is after the mentioned Cadbury-Schweppes-Case in 2006, or the passive-to-total income ratio is above the permitted threshold in the specific CFC rule. By construction, this indicator variable depends indirectly on the ratio of financial-to-total income. This is not a forbidden exclusion criteria for a control variable per se, but it can fuel doubts about the causality of the result of Regression (8). As this is a robustness check which simply confirms the baseline regression, the result is still presented in this study but the reader should interpret this regression result with caution in terms of causality. Nonetheless, the coefficients show the expected signs and the interaction effect of both variables shows a larger negative correlation between the expected effect of CFC rules on financial income in foreign low-tax subsidiaries than in the baseline regression. Another approach to tackle this selection problem is undertaken in Regression (6) of Table 3.4 below. In both regressions, the results only vary a bit in magnitude, however, stay in the expected direction, which leads to the presumption that – for the underlying ORBIS data set – the multinationals that select themselves into this specific low-tax country group are rather small in terms of total

 $^{^{35}}$ Further theoretical thoughts about this issue are undertaken by the mentioned study from Weichen-rieder (1996).

financial profits compared to the non-selecting group. Therefore, the selection doubts, which are generally intractable in the underlying setting, can be satisfied to some extend at least. In addition, these results support the claim that the presented baseline regression results are a lower bound in terms of size rather than an exact estimate, which would be hard to extract in this aggregated setting anyway.

Table	e 3.2: Ro	bustness of	f fixed effects	, clustering or par Log Fin.	ent country con ancial Profits (5)	trols of be	aseline results	8)
	(1)	(Z) 	(3) 	(4) cluster at	(5) cluster at	(b) cluster at	$\operatorname{Parent}_{\cdot}$	(8) Pass Inc
	OLS	no Firm FE	most strict FE	Parent_Country*Year	Sub_Country*Year	Parent	country controls	threshold
-	-0.178^{***} (0.0194)	-0.344*** (0.0183)	-0.167*** (0.0406)	-0.177*** (0.0506)	-0.177^{***}	-0.177^{***} (0.0194)	-0.178^{***} (0.0162)	-0.530^{***} (0.0199)
angible Fixed Assets				0.118^{***}	0.118^{***}	0.118^{***}	0.115^{***}	0.122^{***}
sub				(0.00440) 0.102	(0.00666) 0.102	(0.00441) 0.102	(0.00382) 0.0384	$(0.00351) -0.404^{**}$
-				(0.418)	(1.072)	(0.196)	(0.175)	(0.159)
rruption sub				0.700*** (0.126)	(0.292)	(0.0525)	(0.0526)	(0.0447)
lp sub				2.183***	2.183**	2.183***	2.352***	1.714***
lppc sub				(0.554) -1.326**	(1.028)-1.326	(0.274) -1.326***	(0.219)-1.545***	(10.201)
temployment sub				(0.575) 0.0426	(1.076) 0.0426	(0.275) 0.0426^{*}	(0.219) 0.0265	(0.204)- 0.0241
GUO				(0.0531)	(0.0888)	(0.0229)	(0.0179) 1.053***	(0.0162)
rruption GUO							(0.188) 0.210^{***}	
OIIC							(0.0615)	
p GUO							-0.499	
tmentfreedom GUO							0.00354^{***} (0.000620)	
ve-to-total ne threshold '##P/T-Inc-thresh								$\begin{array}{c} 1.130^{***} \\ (0.0106) \\ -0.217^{***} \\ (0.0210) \end{array}$
rvations	470,420	449,755	156,283	359,243	359,243	359, 243	322,260	359,243
sted R-squared	0.001	0.377	0.768	0.769	0.769	0.725	0.770	0.783
FE		YES		YES	YES	\mathbf{YES}	YES	YES
at FE		YES		YES	YES	YES	YES	YES
алагу ғы idiary*Parent FE			YES	Y E.S	C I X	X E O	CH Y	К ПО
nt_Country*Year FE			YES					
Clust : 'CIT' stands for statu sed from the Heritage F iout any controls or othe	cered robust itory corpora oundation care or tweaks is u	standard errors ate income tax 1 alculation. 'sub used. In Regress	s on parent-year le rate, 'ln gdppc' foi o' stands for subsi sion (2), no firm fi	vel in parentheses if not t the log of GDP per capi diary and 'GUO' for glob xed effects are used. In S	marked otherwise *** (ta, 'ln unemployment oal ultimate owner coi specification (3), I use	k p<0.01, **], for the log o untry. In Spe d the most sti	p<0.05, * p<0.1 f the unemployment 1 cification (1), a simp rict fixed effects with	ate. 'Corruption' le OLS regression subsidiary-parent
parent country-year mx ncluded. In the last R¢ icients larger and still s	ted effects. I ∋gression (8) significant. ∤	In the following , the second 'p: All regressions a	; Regressions (4)-(assive-to-total inco are panel fixed effe	b), different clustering r ome' threshold is include ects estimations and win	ssults are shown. In S and interacted with sorized at the observa	ppecification (the <i>CFC</i> -va tion levels of	(7), additional parent ariable. The signs are 1% and 99% of finar	country controls as expected, the ncial profits. This
e shows that the baselir	ie results are	e robust to diffe	prent fixed effects,	clustering, controls and	the second important	CFC rule thr	reshold.	4

		Table 3.3	3: Robustnes	s of samp	le or variable	e selection	_			
Dependent Variable				L	og Financial Prof	fits				
	$\begin{array}{c} (1) \\ \text{no intermediate} \\ \& \text{ strict FE} \end{array}$	(2) control for intermediate	(3) at least 2 subs in 2 years	(4) Only OECD	(5) Exclude non threshold obs.	(6) Include U.S.	(7) EMPL Control	(8) TOAS Control	(9) OPPL Control	(10) tax diff Control
CFC ln Tangible Fixed Assets	-0.135** (0.0524) 0.112*** (0.00992)	-0.0610^{***} (0.0220) 0.122^{***} (0.00450)	-0.124^{***} (0.0280) 0.105 *** (0.00597)	-0.198^{***} (0.0151) 0.111^{***} (0.00393)	-0.159*** (0.0333) 0.112*** (0.00503)	-0.170^{***} (0.0133) 0.114^{***} (0.00338)	-0.167^{***} (0.0174)	-0.150^{***} (0.0143)	-0.227^{***} (0.0161)	-0.185^{***} (0.0151) 0.118^{***} (0.00362)
ln Employees							0.246^{***} (0.00807)			
ln Total Assets In Operational Profits								0.723^{***} (0.00674)	0.0911^{***}	
CFC RULE intermediate		0.0227							(0.00321)	
CIT intermediate		(0.0753) 1.136*** (0.187)								
tax difference		(101.0)								0.638***
CIT sub	1.335^{***}	0.221	0.873^{***}	0.756^{***}	1.234^{***}	0.00642	-0.485***	0.598***	0.0405	(0.133)
ln corruption sub	(0.394) $(0.323^{***}$	(0.190) 0.547*** (0.0590)	(0.240) 0.735^{***}	(0.1/2) 0.571*** (0.0505)	(0.224) 0.658^{***}	(161.0) 0.705***	(0.154) 0.866^{***}	(0.132) 0.626*** (0.0438)	(),11,0) (),799*** (),0505)	0.740^{***}
ln gdp sub	(0.110) 2.729*** (0.406)	1.460*** 1.460***	1.726*** 1.726***	(0.000) 3.037*** (0.031)	(0.0004) 3.513^{***}	(1.640 + 3.4) 1.640 + + + + + + + + + + + + + + + + + + +	0.510** 0.510**	(0.0420) 2.201*** (0.101)	(0.000) 2.262*** (0.008)	2.191*** (0.907)
In gdppc sub	-0.465	-0.103	-0.0913 -0.0913	-2.312^{***}	-2.502***	-0.699***	0.350	-1.934*** -1.934***	-1.321***	-1.348***
ln unemployment sub	(0.0409) -0.121*** (0.0409)	(0.200) -0.0863*** (0.0220)	(0.0280) -0.123*** (0.0280)	(0.220) 0.0511^{***} (0.0181)	(0.277) 0.108^{***} (0.0212)	(0.2400) 0.0114 (0.0158)	(0.123^{***}) (0.0191)	(0.192) -0.00352 (0.0155)	(0.2.0) 0.0680^{***} (0.0184)	(0.209) 0.0472^{***} (0.0168)
Observations Adjusted R-squared	60,847 0.772	248,596 0.753	156,297 0.769	301,613 0.772	197,798 0.780	$418,702 \\ 0.769$	289,989 0.773	430,686 0.787	289,318 0.773	359,243 0.769
Year & Parent & Subsidiary FE Subsidiary*Parent FE Parent-Country*Year FE	YES YES	YES	YES	YES	YES	YES	YES	YES	YES	YES
Note: 'CIT' stands for statutory c from the Heritage Foundation cald from reg.(3) in the above Table 3, but some controls for the intermec following Regressions (4)-(6), coun is included in the law (5), and incl difference between the two countri and 99% of Financial Profits. This	Clustered rol corporate income ta culation. 'sub' stata culation. 'sub' stata diate countries are try sample selectio fuding the U.S. in (es instead of the fo s table shows that	bust standard er ax rate, 'ln gdpp rds for subsidiar is time only dire included. In Sp n biases are excl 6). Different con reign subsidiary the baseline rest	rors on parent-ye cc' for the log of (y and 'intermedia ext cross-border o ecification (3) onl luded by observin, ntrol variables on country is used. ldts are robust to	ar level in pa GDP per capi ate' for an int wmership link ly parents wit g only OECD the subsidiar. All regression differently co	tentheses *** p< ta, 'hn unemploy rermediate owner ages without int ages without int h at least two su parents in (4), e: y level are used in s are panel fixed nducted variable	0.01, ** p<0 ment' for the country. In ermediate ow ubsidiaries for xcluding CFC n Specificatio effects estime and sample s	05, * p<0.1 log of the u. Specification ners are obs at least two at least two 3 subsidiary c ns (7)-(9). Ir thions and wi delection chec	aemployment (1), again t erved. In reg consecutive observations ' a the last Spe nsorized at t :ss:	rate. 'Corruphe same strict he same strict gression (2), tl years are observation where no low-i cefification (10) he observatior	tion' is used fixed effects ins is relaxed arved. In the ax threshold the tax rate i levels of 1%

In Table 3.3, I examine if the sample or the selected variables could bias my baseline results. Thereto, I use various different samples and control variables in the following robustness regressions. Again, all regressions show significant negative coefficients for the variable of interest, selection biases seem to not drive the results and the regressions confirm the effects of CFC rules on financial profits in low-tax subsidiaries. In Specification (1) of Table 3.3, I exclude all observations that have chain ownership structures with any intermediate companies and observe direct parents only, again with the most strict fixed effects to account for link-specific effects. In Regression (2), I include possible intermediate companies again, but this time control variables for a potential CFC rule and the statutory CIT rate in those countries are included additionally. One can see less observations and the effect of interest is smaller in size, but it is still negatively significant. In Specification (3), I test if another sample constraint could alter the outcome: in this regression, observations are included if the parent has at least two subsidiaries in two years. This should exclude purely (non-) shifting multinationals and provide a preferable comparison environment, but less observations. The results show quality-wise the same outcome, only with far less observations. In Regression (4), I restrict the sample to OECD countries only, excluding the other 25 countries and, again, the baseline results seem to be robust to that test.

As this study tries to compare as many CFC rules as possible from various countries, I included all of these countries so far, even though some of them do not use the low-tax threshold to determine whether subsidiaries are located in low-tax jurisdictions. Some others do, but not for all years. For all these country-years, a subsidiary counted as located in a low-tax country in all regressions so far, which would have driven my results, or rather the significance, to zero. In Regression (5), observations from country-years where only a low-tax threshold exists in the CFC rule as well as of parent countries without CFC rules are included to test for any result driving factors. The results stay the same and show the robustness of the baseline results. So far, in all regressions, the United States (U.S.) are excluded from the calculations, due to their "check the box" rules, which undermined their existing CFC rules.³⁶ In Regression (6), this exclusion is lifted and parents from the US as well as their subsidiaries are included, using the rules from the U.S. CFC law and the according thresholds. Despite, including a higher number of observations, the results stay robust.

Generally, it is not clear which control variable should be used to account for real foreign actions and business activity in the foreign subsidiary. To account and control for this issue, so far, tangible fixed assets are used. In Regressions (7) - (9), different variations for the firm-level control are utilized. All results still show negative significant influence of probably affected financial profits in low-tax subsidiaries, even though the number of observations and the size of the coefficient differs. In a further regression, I used log of sales instead, and the results are pretty similar to those of Regression (7).³⁷ These variables control in a different manner, and potentially even better, for the variability of financial reports, which are irrelevant for profit shifting. On the other hand, these variables are potentially more distorted due to transfer pricing. Interestingly, Regression (7) as well as the not presented one with log of sales instead of tangible fixed assets as control variables are the only regressions that show a clear negative significant sign for the foreign subsidiaries' CIT rate. This would be the expected sign in a normal FDI consideration, but not necessarily in the here observed profit shifting world, as discussed above. In Regression (10), the difference between the parent and the subsidiary CIT rate is used instead of the simple subsidiary CIT rate. The coefficient of interest is slightly larger and shows that with a higher tax rate differential, increasing financial profits accrue abroad, which is plausible and in line with the former mentioned studies about profit shifting.

One could have a further doubt in mind, namely that the results are driven by larger global tax system influences. As shown in Feld, Ruf, et al. (2016) or Clausing (2015), the general tax system probably not exclusively influences FDI, but also profit shifting activities. To show that my baseline results are not driven by these factors, I included a dummy variable

 $^{^{36}}$ For further details see Section 2.4.

³⁷ Due to the similarity, the results are not shown in the table but are available on request.

that is 1 if the country's tax system uses the exemption method (territorial tax system) and zero otherwise in Regression (1) in Table 3.4. One can see that the variable of interest is barely influenced by implementing this additional control. The control variable itself shows that subsidiaries from parents within territorial tax systems generally have less financial profits abroad, which is in line with the former mentioned literature.³⁸ Further important international tax laws concerning financial profits abroad are the already mentioned thincapitalization rules. In Regression (2), I checked for their influence by combining the used CFC low-tax dummy with these rules. The new dummy of interest is 1, if the subsidiary is located in a low-tax country according to the CFC laws of its parent *and* if thin-capitalization rules are implemented in the parent country. The results are still negatively significant and slightly larger in size, which shows that financial profits in foreign low-tax countries are even smaller if both anti-tax avoidance laws are implemented. Specifically, debt shifting from the parent towards the subsidiary is harder under these circumstances.

In Regression (3), the variable of interest from the baseline results is split up into three categories determined by the company size of the parents.³⁹ It seems the smaller the parent is the more influence does the CFC legislation have, or at least, that with a larger size of the parent company, the financial profits in foreign low-tax subsidiaries are less affected, even though the results are all negatively significant.⁴⁰ The following Regression (4) shows a comparable picture, but this time for the observed company size of the subsidiaries.⁴¹

³⁸ The reader should remember that in these regressions, parents from the U.S. are excluded. But, even if the parents and their subsidiaries are included in this calculation, the results stay the same; they are highly significant but the coefficient for the used non-double tax method is smaller in size. These results are not shown here but are available on request.

³⁹ For the determination of the three categories, I used the provided identifier in the ORBIS data. I excluded *smallfirms* as companies can end up in this category just by missing data at Bureau van Dijk. With subsidiaries in low-tax countries included are: 14,710 very large parents, 1,957 large parents and 1,871 medium-sized parents.

⁴⁰ Another possible reason for this finding could be the additionally important passive-to-total income threshold, which is examined in the next chapter. In short at this point: If a company is larger and has more real income abroad more profits can be shifted, as long as the passive-to-total income ratio does not exceed a certain threshold.

⁴¹ Here, the proportion of sizes is different: 4,998 very large subsidiaries, 16,291 large subsidiaries, and 28,756 medium-sized subsidiaries are located in low-tax countries and are presumably affected by CFC rules of their parents.

Again, we observe a larger influence on smaller subsidiaries. The reasons for this behavior are manifold and will not be discussed here as they are pure speculation on this level of observation.

As most CFC rules state that at least 50% of the foreign subsidiary must be under control of the parent to be affected by these laws, this threshold is set in all regressions so far. One could argue that even with 50% the possible profit shifting scope is limited. Therefore, in Regression (5), only subsidiaries that are owned with more than 99% are deemed to be possibly affected by profit shifting behaviour and the indicator variable turns 1 if this higher ownership threshold is reached. The results for this variable are still negatively significant but smaller in size, which probably shows that with less ownership shares, sufficient profit shifting decision making power is present already. The EEA exemption due to the Cadbury-Schweppes-Case, explained in Section 2.1 and with more results in Section 3.6.2.2, can annul the low-tax threshold legislation if the main part of the foreign income is from active nature. To demonstrate this implication, in Regression (6), the subsidiaries which are located in a low-tax country and which bunch at the given EEA threshold explained in Section 2.1 are added as an additional group of subsidiaries. Due to this exemption, I do not observe a negative influence of CFC rules on this very specific group of profit shifting vehicles but observe, as expected, higher financial profits abroad.

Dependent Variable			Log Fi	nancial Profits		
	(1) Double Tax Method	(2) Including Thin Cap	(3) Company Size Parent	(4) Company Size Subsidiary	(5) Only > 99 percent ownership CFC	(6) EEA Exemption
CFC	-0.164^{***} (0.0152)					-0.178^{***} (0.0150)
Territorial tax system at GUO $CFC \& TC$ Present	-0.133^{***} (0.0179)	-0.180***				
very large		(0.0150)	-0.106^{***}			
large			(0.0330) -0.156^{**} (0.0725)			
medium			-0.285^{***} (0.0827)			
very large sub				-0.150^{***} (0.0535)		
large sub				-0.171^{***} (0.0281)		
medium sub				-0.184^{***} (0.0168)		
$CFC_{alternative}$					-0.101^{***} (0.0211)	
EEA Buncher						0.438^{***} (0.148)
In Tangible Fixed Assets	$\begin{array}{c} 0.113^{***} \\ (0.00373) \\ 0.022 \end{array}$	$\begin{array}{c} 0.117^{***} \\ (0.00363) \\ 0.111 \end{array}$	0.112^{***} (0.00630)	0.118^{***} (0.00362)	(0.119^{***}) (0.00362)	0.118^{***} (0.00362)
ln corruption sub	0.232 (0.166) 0.658***	(0.111) (0.163) 0.604***	(0.252)	(0.108) (0.163) 0.702***	(0.101) (0.163) 0.631***	(0.102) (0.163) 0.700***
In edn sub	(0.0470) 2 172***	(0.094) (0.0460) 2.212***	(0.0802) 2 769***	(0.0460) 2 187***	(0.0452) 1 949***	(0.0459) 2 182***
ln gdppc sub	(0.213) -1 463***	(0.208) -1.346***	(0.362) -0.976***	(0.207)	(0.206) -1 156***	(0.207) -1 325***
In unemployment sub	$(0.213) \\ (0.0361^{**}) \\ (0.0173)$	$(0.210) \\ (0.0425^{**}) \\ (0.0168)$	$\begin{array}{c} (0.371) \\ -0.113^{***} \\ (0.0287) \end{array}$	$(0.209) \\ 0.0428^{**} \\ (0.0168)$	$(0.209) \\ 0.0355^{**} \\ (0.0168)$	$(0.209) \\ 0.0425^{**} \\ (0.0168)$
Observations Adjusted R-squared Year & Parent & Sub FE	337,790 0.771 YES	357,643 0.769 YES	130,661 0.760 YES	359,243 0.769 YES	359,243 0.769 YES	359,243 0.769 YES

Table 3.4: Robustness of further possible influences on CFC rules

Clustered robust standard errors on parent-year level in parentheses *** p<0.01, ** p<0.05, * p<0.1

Note: For the determination of the three categories 'very large', 'large' and 'medium' I used the provided identifier from the Orbis data. 'CIT' stands for statutory corporate income tax rate, 'ln gdppc' for the log of GDP per capita, 'ln unemployment' for the log of the unemployment rate. 'Corruption' is used from the Heritage Foundation calculation. 'sub' stands for subsidiary country. In Specification (1), an additional variable for tax system differentiation is added which is 1 if the parent country corporate tax system uses the exemption method and 0 otherwise. In Regression (2), the CFC variable is altered and is now only 1 if, additionally to the former restrictions, a thin capitalization rule is enacted in the parent country. In Specification (3), a distinction between the parent company sizes is taken and the former CFC variable is split up into three categories. In Regression (4), the same distinction is taken but this time for the subsidiary company size. In Specification (5), a different variable of interest is created by allowing only more than 99% of ownership instead of 50% to account for possible CFC rule effects as in the other regressions. Regression (6) shows that specific subsidiaries, which are located in low-tax countries at the EEA exemption threshold, have higher financial income due to another special EEA exemption explained in detail in Section 3.6.2.2. All regressions are panel fixed effects estimations and winsorized at the observation levels of 1% and 99% of financial profits. This table shows that also the inclusion of further international taxation issues has no generally changing influence to the baseline results, and that the coefficients are different in size for different company sizes.

In the last table of the robustness checks, Table 3.5, I checked for various alternative dependent variables to show that my results are not driven by the unusual variable choice. For readability reasons, I repeated the baseline result in Regression (1). In Specification (2), I used intangible fixed assets of foreign subsidiaries instead as dependent variable, as this measure is often used in other profit shifting literature. As written above, this variable is used to indicate shifting behavior as these assets often stand for patents, trademarks or other intellectual property which are comparably easy to shit. With a higher number of observations, one can see the same results, even with higher magnitude. If the subsidiary is located in a foreign low-tax country, as determined by the CFC laws of its parent, less intangible fixed assets are located in this subsidiary.⁴² To account for the fact that in all former financial profit regressions the logarithmic transformation is used and, therefore, only positive observations are included, I relaxed this often used restriction in Regression (3), and set the logarithm of financial profits over total profits as dependent variable, which includes some negative observations as well. The coefficient of the variable of interest is still negatively significant, even though smaller in size. One should keep in mind that fractions as dependent variables have their own flaws, e.g. presence of corner solutions (e.g., Loudermilk (2007)). To learn more about CFC rule influence, I included Specifications (4)-(6), consisting of far more observations due to richer variable data availability. Real activity abroad as proxied by *Sales* and *OperationalProfits* is higher in these low-tax subsidiaries, which could have many reasons. No specific assumption of the signs of these coefficients is expected. Regression (5) shows that less *FixedAssets* are used in these low-tax subsidiaries compared to others, which is in line with former results from Egger and Wamser (2015). As log of tangible assets still accounts for real FDI like machinery, the positive signs could indicate potential transfer mis-pricing in case of these specific low-tax subsidiaries, as men-

 $^{^{42}}$ In fact, nearly all former regressions run with intangible assets show a similar picture. They are not depicted here, but are available from the author upon request.

tioned above.⁴³ Additionally, this indicates that countries which actually aim to attract real investment from foreign countries could be better off if CIT rates were increased.

As it is not clear what happens with the less shifted profits, as shown in Regressions (4) and (5) in the baseline results, and where they end up, I included Regressions (7)-(9) to show some insights into these matters. Regressions (7) and (8) show that parents with subsidiaries located in low-tax countries pay comparably higher taxes in their home countries, which indicates that the higher taxation of CFC rules might work. The control variables used to cancel out other possible influences are showing the expected signs and are mostly significant, even with different standard error clustering.⁴⁴ Additionally, Specification (9) shows that subsidiaries located in foreign low-tax countries pay comparably less taxes in their host countries as well. This is plausible as we are observing subsidiaries in lower tax countries. These results suggest that CFC rules work as intended and that higher taxes are paid in the parent country.

 $^{^{43}}$ Interestingly, if the control variable of tangible fixed assets is removed, these signs turn negative and the coefficient of *FixedAssets* becomes larger as expected. This reinforces the hypothesis of a transfer pricing influence on active income or real FDI, which is not the focus of this study.

 $^{^{44}}$ The same fixed effects are used to account for the aforementioned possible influences in these regressions as well.

		Table 3.5	: Robustness of	depender	it variable	and further	insights		
Dependent variable	(1) Base result	(2) ln(Intangible Fixed Assets)	(3) ln(Financial Profits /Profits)	(4) $\ln(\mathrm{Sales})$	(5) ln(Fixed Assets)	(6) ln(Operational Profits)	(7) ln(taxes paid) Parent	(8) ln(taxes paid) cluster Parent	(9) ln(taxes paid) Subsidiary
<i>CFC</i> In Tangible Fixed Assets	-0.177^{***} (0.0150) 0.118^{***} (0.00362)	-0.266^{***} (0.00723) 0.223^{***} (0.00253)	-0.0473*** (0.0116) 0.0605*** (0.00273)	$\begin{array}{c} 0.00752^{**} \\ (0.00295) \\ 0.193^{***} \\ (0.00119) \end{array}$	-0.0462^{***} (0.00214) 0.711^{***} (0.00137)	$\begin{array}{c} 0.0220^{***} \\ (0.00477) \\ 0.173^{***} \\ (0.00144) \end{array}$	0.103*** (0.0204)	0.103** (0.0511)	-0.0694^{***} (0.00341)
ln Operational Profits							0.0397^{***}	0.0397^{***}	0.719^{***}
CIT sub	0.102	1.989 ***	0.618***	0.366^{***}	-0.0188	-0.493^{***}	0.207	0.207	2.404***
ln corruption sub	(0.103) 0.700^{***}	(0.0932) 0.291^{***}	(0.144) 0.806^{***}	(0.0421) 0.00286	(0.0271) 0.178^{***}	(0.0640) -0.197***	(0.188) 0.0464	(0.334) 0.0464	(0.0549) - 0.283^{***}
dus abg nl	(0.0459) 2.183^{***}	(0.0265) -3.265***	(0.0385) 0.691^{***}	(0.0105) - 0.501^{***}	$(0.00707) \\ 0.189^{***}$	(0.0158) - 0.0234	(0.0562) 0.893^{***}	(0.0907) 0.893^{**}	(0.0118) -0.319***
	(0.207)	(0.130)	(0.173)	(0.0431)	(0.0352)	(0.0746)	(0.256)	(0.396)	(0.0596)
In gappe sub	-1.320	(0.126)	(0.172)	(0.0446)	(0.0349)	(0.0722)	-1.029	(0.406)	(0.0546)
ln unemployment sub	0.0426^{**}	-0.504^{***}	0.166^{***}	-0.115***	0.0680^{***}	-0.192^{***}	-0.0385**	-0.0385	-0.127^{***}
	(0.0168)	(0.0102)	(0.0145)	(0.00341)	(0.00273)	(0.00613)	(0.0196)	$(0.0336)_{-0.267}$	(0.00517)
							(0.206)	(0.432)	
In corruption GUO							-0.734^{***}	-0.734^{***}	
ln gdp GUO							(0.0773) 1.429***	(0.158) 1.429^{**}	
•							(0.300)	(0.632)	
In gdppc GUO							1.225^{***} (0.356)	1.225^{*} (0.673)	
In unemployment GUO							-0.0684^{***} (0.0253)	-0.0684 (0.0483)	
Observations	359, 243	936,233	580,701	1,311,110	1,942,189	1,251,904	183,059	183,059	1,134,741
Adjusted R-squared Voor FF	0.769 VFS	0.801 VFS	0.501 VFS	0.903 VFS	0.959 VFS	0.805 VFS	0.893 VFS	0.893 VFS	0.897 VFS
Parent FE	YES	YES	YES	YES	YES	YES	YES	YES	YES
Subsidiary FE	YES	YES	YES	\mathbf{YES}	YES	\mathbf{YES}	YES	YES	YES
Note: 'CIT' stands for sta	Clustered r	obust standard e	rrors on parent-year lev cate 'ln ødnne' for the	rel in parentl log of GDP	neses if not ot ner canita 'l	her marked *** p	<0.01, ** p<0.05, for the low of the	, * p<0.1 unemployment ra	te 'Corruption' is
used from the Heritage For	indation cal	culation. 'sub' sta	uds for subsidiary and	'GUO' for a	global ultime	te owner country.	In specification ((1) the baseline reg	ression is repeated
for better comparison. In	the following	g Regressions (2)-	(6), different dependen	t variables i	nstead of fina	ncial profits are u	sed at the subsidi	lary level: in (2) ir	tangible assets, in

(3) financial profits over total profits to include possible negative observations, in Regressions (4)-(6) non-shifting parameters are used as dependent variable: in Specification (4) sales, in (5) fixed assets and in (6) operational profits. In the following Regressions (7)-(9), effects of CFC rules on tax payments are shown with simple robust standard errors. In Specifications (7) and (8), for taxes paid at the parent level with different clustering for robust and for robust clustering at the subsidiary country-year level. In Regression (9) taxes paid by the foreign subsidiary, are used as dependent variable instead. All regressions are panel fixed effects estimations and winsorized at the observation levels of 1% and 99% of Financial Profits. This table shows the robustness of the baseline results for different depended variables and shows further insights into CFC rule influences for other variables of interest.

3.6.2 How CFC rules are circumvented

Despite the shown impact of CFC rules, that these laws lead, generally, to less financial income in foreign low-tax subsidiaries abroad and, therefore, potentially to more CIT revenue globally, multinationals still try to minimize their tax payments and find ways to circumvent CFC rule taxation. A first possibility for multinationals to do so is shown in the previous section Regressions (4) and (5) of the baseline results in Table 3.1, and is achieved by shifting profits into subsidiaries which are located in countries with CIT rates above the minimum low-tax threshold but still below the home CIT rate of the CFC rule constrained parent. A second way, that is to move the headquarters into countries without CFC rules, is researched by Voget (2011) and has already been discussed above.

This subsection focuses on a third potential means to bypass CFC rule taxation: the usage of the given passive-to-total income threshold of many CFC rules. This corresponds to Option (1) in the conceptual framework Section 3.3. If the financial-to-active income ratio of a foreign subsidiary is below the given thresholds, the parent of that entity is usually not affected by CFC law taxation. As discussed above, Weichenrieder (1996) has shown theoretically that this threshold can lead to even more active foreign income to keep the ratio below the threshold. I provided empirical proof for these thoughts by showing that the deterrence effect seems to be larger, as can be seen in Regressions (4) and (6) with higher sales and operational profits in the last Table 3.5 of the previous section. Whether these profits are accruing from actual active activity or are rather re-framed profits is not detectable from the available data base. The result of Regression (5) in the same table is in favour of the latter one, as it shows lower fixed assets in these low-tax subsidiaries. The finding of deterrence for active investment in low-tax subsidiaries by CFC rules is in line with Egger and Wamser (2015), who find decreasing foreign real investments in the treatment case for German multinationals, i.e. if the subsidiary is below the low-tax threshold and affected by German CFC rules.

Altogether, the results in this subsection provide evidence that multinationals from around the globe use these given thresholds to shift their profits into low-tax countries.

3.6.2.1 Graphical evidence for the influence of the EEA exemption

As explained in detail in Section 2.2, in nearly all countries of the EEA which do have a CFC rule, financial profits from EEA country subsidiaries are exempt from CFC law taxation if the passive income is below 50% of the overall profits of that subsidiary. This results mainly due to an ECJ ruling (European Court of Justice (2006)) in late 2006. After that, many countries within, but also outside, the EU introduced such an exemption in their CFC laws.



Figure 3.2: Differences in financial income due to the EEA exemption

Notes: The figure shows financial profits of foreign subsidiaries held by owners located in the EEA affected by a CFC law amendment, which gave profit shifting advantages to specific EEA subsidiaries. The red line depicts the treatment group and consists of subsidiaries with a passive-to-total income ratio below 50% located in EEA countries. The blue line is the control group and consists of subsidiaries which are not advantaged by the CFC law amendment but their parents are located in the same countries as the treatment group.
In Figure 3.2, I provided graphical evidence with an event study type, where one can see the different accruing financial profits in foreign subsidiaries in relation to the EEA exemption introduction in the observed countries. The red line (treatment group) shows the increase of financial profits of EEA subsidiaries with a passive-to-total income ratio below 50%, after the home country of the owner introduced the mentioned EEA exemption. The blue line (control group) shows the decrease of financial profits in foreign subsidiaries, which are not affected by the EEA exemption but held by parents located in the same countries as the treatment group. Not every EEA government adjusted the CFC rule right after the ECJ case in the years 2006 or 2007 but, as one can see from the divergence of the two lines before the law amendment, that multinationals anticipated changes and started the re-shifting of profits earlier already. The increase of the red line from year -1 to 0 accounts for a plus of 35 percentage points in financial profits in these CFC law exempted subsidiaries. Otherwise, the decrease of the blue line from year -1 to 0 corresponds to 19% less financial profits in the subsidiaries of the control group. This picture gives rise to the idea that multinationals shifted their profits from low-tax countries around the world into EEA countries, which are exempt from CFC legislation as long as the financial profits are below the passive-to-total income threshold. The reader should keep in mind that this is not only true for the here examined European ultimate owners but also for subsidiaries in owner chains from other countries, which happen to be located in an EEA country with according CFC rules.

For German multinationals, the coherence of the EEA exemption and foreign passive assets as investment is already shown by Ruf and Weichenrieder (2013) using data about German multinationals. The OLS estimates for the treatment effect are slightly larger (with an increase up to 40% in passive assets), but in the range of my estimate for the increase in foreign financial profits within EEA subsidiaries. In the mentioned policy paper by Bräutigam, Spengel, and Streif (2017), the authors show with calculated effective CIT rates that the ECJ decision on tax neutrality issues, not only led to less strict CFC rules but in a second step also to the rise of IP box regimes within the EU.

3.6.2.2 Impact of the passive-to-total income threshold on financial income abroad

In this section, I observe the impact of another important threshold, which is, in different forms, often used in CFC rules around the world: the above explained passive-to-total income threshold. As one can easily see in Figure 3.3 this threshold seems to alter corporations' behavior significantly. Ruf and Weichenrieder (2013) show that the foreign low-tax intangible assets from German multinationals are negatively influenced by a similar effect. In this section, I provide evidence for the influence of these thresholds on direct foreign low-tax income for multinationals from various countries. Figure B.1 in Appendix B.4 shows a zoomed out angle with a broader range of income ratios to illustrate the broader picture. In Table B.5 in Appendix B.4 I attached a graph for every country that is used in this study. One can see clear bunching in the countries that use the passive-to-total income ratio of 50% to determine whether a foreign entity is a CFC.⁴⁵ In the second table of this Appendix, Table B.6 no bunching occurs in subsidiaries from parent countries without a CFC rule or comparable anti-tax avoidance scheme. These graphical findings show the expected behavior from the theoretical framework as well as the law specifications explained in Section 3.3; specifically the option choice (1) from the decision tree in Figure 3.1 and the following mathematically formulated thoughts.

Figure 3.3 shows the first result of the graphical bunching analysis, including the observed and counterfactual distributions as a red and nearly horizontal line. It shows the densities of subsidiaries with parents in countries with enacted CFC rules that do have such a passive-tototal income threshold at 50%. The solid line with dotted markers plots the observed number of companies in income bins of 0.1 percent income ratios. Each dot denotes the upper bound of a given bin and represents the number of companies in each bin. The red solid smooth line shows the counterfactual density based on fitting a seventh-order polynomial using company

⁴⁵ One can also see Hong Kong on that list, which does not have a CFC rule in its legislation but has implemented a comparable law that entails similar consequences for non-financial entities.



Figure 3.3: Bunching at passive income threshold of 50% ratio

Notes: The figure shows an aggregated picture of foreign subsidiaries with positive financial profits from parent home countries where a CFC rule is enacted. These subsidiaries bunch right before the 50% passive-to-total income ratio, which is often the threshold to count as a CFC and get taxed by the observed CFC laws. For visibility reasons, the depicted span reaches from a ratio of above 40% to below 60% of financial-to-total profits abroad before taxes. A zoomed out version with a larger range of income ratios is shown in Figure B.1 in Appendix B.4.

counts with taxable income ratios between 40% and 60%, except for firms in the excluded range close to the critical 50% ratio. Due to the fact that it is easier for companies with their book incomes to stay exactly at or right before the threshold than for individuals, the observable bunching is much clearer as it is in the known bunching literature about individuals. As described above, the potentially affected foreign income base depends on the law (if the full income or only the (incremental) passive income is affected), and no clear notch or kink dynamic is specifiable in this aggregated setting. For the same reasons, the potentially expected missing mass on the right hand side of the threshold is not discernible. The interesting fact here is not the calculable elasticity, as this would be very aggregated and not really usable in this case, as explained in Section 3.3. But rather the huge bunching in a multinational corporate world at a book ratio of profits, which is easily tweak-able by accountants. Further specific insights about this behavior are shown in this chapter. These results proof Conjecture 2 from the theoretical framework.

Companies which bunch at this specific threshold are interesting from the profit shifting perspective, as there is no further clear motivation other than profit shifting reasons to accumulate profits ending up right below the observed threshold. Not only small or super large multinationals are observed here, but firms of all sizes.⁴⁶ There are other ways for multinationals to shift their profits without getting taxed in the home country at the higher tax rates. The reader should keep in mind that the depicted bunchers are only a very specific fraction of multinationals, which try to shift their profits to get taxed at a lower rate by using this method of tax avoidance. Also, the bunchers are probably only the tip of the iceberg of multinationals that circumvent CFC rule taxation in low-tax countries by utilizing the passive-to-total income threshold. The following distinctions are provided for robustness evidence of the observed bunching causality.

Bunching differences over time

To illustrate the differences in behavior changes of multinationals, I also investigated the adjustments over time when CFC laws changed. As one can see in Figure 3.4, the CFC laws of Sweden, for example, changed for the multinationals from 2007 to 2008.

One can see that, after the introduction of the 50% ratio threshold for subsidiaries from Swedish headquartered companies, subsidiaries gathered right before this specific ratio of financial-to-total income abroad. Before the introduction no clear bunching picture is observable. Even though it is visible that less data points are available for the time before 2008, the data is sufficient to see the clear differences in frequencies. This indicates the usage of tax avoidance strategies, after they became possible, to stay right below or at the threshold to not get affected by the Swedish CFC rule.

⁴⁶ The interested reader finds some statistics in Table B.7 in Appendix B.4.



Figure 3.4: Bunching of Swedish subsidiaries over time

Notes: The figure shows bunching of foreign subsidiaries owned by Swedish parent companies before and after the passive-to-total income threshold introduction into Swedish CFC legislation. After implementing the threshold, the numbers of foreign subsidiaries bunch right before the 50% passive-to-total income ratio, for not getting affected by CFC taxation. For visibility reasons, the depicted span reaches from a ratio of above 40% to below 60% of financial-to-total profits before taxes abroad. The bin width of 0.45 percentage points is the smallest possible width for the few observations in the first time frame. For comparability reasons the bin width in the right hand graph is not much smaller, even though the bunching picture would be much more striking otherwise.

Bunching differences over locations - The EEA exemption

Not only the aforementioned low-tax threshold, which has been shown in Section 3.6.1, shows the influence of CFC laws on location decisions. The passive-to-total income threshold also affects the structure of subsidiaries, and more specifically, the asset and income structure within foreign subsidiaries. To show the differences in the behavior of multinationals concerning the income structure in foreign subsidiaries, I show the income ratio of subsidiaries located in the EEA area in contrast to the income ratio of subsidiaries in non-EEA countries in Figure 3.5. This can be understood as a further investigation of the already presented event study results in Figure 3.2 above. All subsidiaries are owned by parents in EEA countries with enacted CFC rules. As explained above, the big difference is that subsidiaries in EEA countries are allowed to have a ratio of up to 50% passive-to-total income, and only count as CFCs if they are above this threshold.



Figure 3.5: Bunching within EEA due to EEA exemption

Notes: The figure shows bunching of foreign subsidiaries from parent companies within the EEA. The left graph shows bunching of subsidiaries within the EEA and the right graph shows no bunching for subsidiaries outside the EEA. For the latter ones, no specific CFC rule threshold exists. For visibility reasons, the depicted span reaches from a ratio of above 40% to below 60% financial to total profits before taxes abroad.

In Figure 3.5 it is clearly visible that companies use these permitted tax avoidance opportunities, emerging due to the explained EEA exemption, to shift profits to their foreign subsidiaries located in the EEA area, which alters profit shifting and investment behaviour. This, again, indicates a clear usage of tax avoidance strategies to stay right below the threshold to get not effected by the CFC rules of the home countries.

(Non-) Bunching for Germany and the U.S. at save harbour thresholds

There are a few countries, in which the passive-to-total income threshold is not set at 50%, but at different percentage points. In this section, I show the specialties about the German and the U.S. income thresholds as examples. As one can see in Figure 3.6 German subsidiaries do not bunch at the actual German CFC passive-to-total income threshold of 10%, but at the special 50% threshold of the EEA exemption that accounts for German subsidiaries in the EEA as well. From an economics standpoint, it seems like German parents use European subsidiaries to shift their profits by staying right below or at the EEA exemption ratio. In the case of non-European subsidiaries, they would need to have at least 90% real income to shift profits which, apparently, is not as attractive as the workarounds given by the EEA exemption.



Notes: The figure shows no bunching of foreign subsidiaries from German parent companies at the actual passive-to-total income threshold of German CFC legislation, but bunching at the 50% EEA exemption threshold. For visibility reasons, the depicted spans reach from a ratio of above 0% to below 20% and above 35% to below 65% of financial-to-total profits abroad before taxes.

One can see a comparable picture for the case of subsidiaries from U.S. parents in Figure 3.7. Here, the original passive-to-total income threshold, or rather a save harbour threshold, is set at 5% of financial income abroad, meaning that if the subsidiary yields more than 5% financial income, the foreign income would be taxed according to the Subpart F Income rules (the CFC rules from the U.S.). As explained earlier, the U.S. CFC rules are rather easy to circumvent by the check-the-box-rules. Again, it is not really profitable for an U.S. company to set up a foreign low-tax subsidiary and shift only 5% of financial income. There are easier ways to shift more profits abroad for U.S parents.

From these two examples, one can see that the examined passive-to-total income threshold only alters multinationals' behavior if it is still profitable enough for these shifters. As seen in the theoretical framework and the decision tree in Figure 3.1 as well as in the former regressions in Section 3.6.1, if other options to circumvent possible CFC legislation are cheaper to achieve, companies would choose them to not get taxed by CFC laws.



Figure 3.7: Bunching of subsidiaries from U.S. parents

Notes: The figure shows no bunching of foreign subsidiaries from U.S. parent companies at the actual passiveto-total income threshold of U.S. CFC legislation of 5%. For visibility reasons, the depicted spans reach from a ratio of above 0% to below 15% of financial to total profits abroad before taxes.

3.6.2.3 Amount of shifted profits abroad

Using this clear bunching, it is possible to calculate the amount of profits, which are presumably shifted abroad from the high tax home countries, as there is no other reason to stay right below or at the threshold. Therefore, I use the (positive) reported financial profits of subsidiaries, which are located right below or at the passive-to-total income threshold of the parent CFC rules. These presumably shifted financial incomes og the bunching subsidiaries are simply summed up.⁴⁷ To not overstate these buncher profits, I calculate the average of the financial profits of all subsidiaries with passive-to-total income ratios spanning over the range from 40% to 60% of financial profits (excluding the 3% width bunching bin below the 50% threshold). These average financial profits are multiplied with the average number of subsidiaries in the buncher area, and then subtracted from the previously calculated shifted profits of bunchers. This deduction accounts for the visible counterfactual distribution, the red line in Figure 3.3, examined in this section. The three percent bin below the 50% threshold is used to account for shifting firms which are not located exactly at the limit. With

⁴⁷ Even if not all of these financial incomes are shifted abroad, there are likely more shifted profits below the threshold without bunching, which are not accounted for in this simple calculation. These would at least even out the possible overstatement of the summed bunching profits.

the used underlying data set and the time frame of this study, the numbers accumulate to around 11.67 billion USD of shifted profits out of CFC rule countries. This simple backof-the-envelope calculation is a very conservative and aggregated approximation of profits shifted into foreign low-tax subsidiaries to avoid taxation at higher tax rates at home, and to even circumvent CFC legislation at this specific threshold. Nevertheless, it is the first study to numerically approximate the profit shifting behavior of multinationals, which use thresholds of CFC rules to circumvent taxation by their parent home countries.

Further ways of circumventing CFC rules

Obviously, there are more ways to circumvent CFC legislation by utilizing or bypass other requirements of these laws. For example, one could think of specific ownership structures, where the true owner is hidden in chains that are untraceable, other legal tricks in framing the foreign entity's affiliation, or even specific deals with governments. These tax avoidance strategies are not detectable with the underlying data set, but could potentially be researched in further studies.

3.7 Conclusion

Profit shifting of multinationals into foreign low-tax countries and its interwovenness with international corporate taxation has never been more complex than it is today. Through my research, I have shown that the anti-tax avoidance legislation "CFC rule" is one avenue, which countries have to consider to solve these issues. I compare, for various countries, the decision influence of these laws on profit shifting behavior of multinationals abroad. Therefore, I examine the influence on foreign passive income, which arises very often due to the profit shifting behavior of multinationals, especially in larger and uncommon amounts. To accomplish this, I use a large firm-level data base and a self-conducted data set with specifics about CFC legislations over a span of 11 years for 61 countries worldwide. I find that CFC laws in general lead to around 19.1% less passive income in low-tax subsidiaries abroad. To identify these effects, specifications with varying time and cross-sectional scope are used in firm fixed effects regressions with various robustness checks. In an event study, it is shown that passive income abroad increases (decreases) significantly if the underlying CFC laws in the parent country are relaxed (strengthened). Additionally, I used graphical evidence to demonstrate that the changes in the behaviors of the multinationals are caused by CFC laws and showed clear bunching. I exhibited both from a time and a location perspective that this bunching behavior is due to the underlying CFC laws. Out of this bunching, I calculated the sum of 11.67 billion USD, which are shifted abroad to avoid taxation in high-tax home countries and to even circumvent CFC legislation by benefiting from the specified threshold. Even though these are more than 1 billion USD per year on average, one should keep in mind, as shown for example by Zucman (2015), that this figure is likely a lower bound. This is due to the fact that in the used Orbis firm-level data, probably only a few companies are used as shifting vehicles, and, generally, detailed data availability for these specific companies is scarce.

My findings not only prove that CFC legislations counteract profit shifting behavior of multinationals but also show that these laws can be circumvented. Companies are able to do so by using the different given thresholds in the laws to stay right above or below them. However, different CFC rules can influence investment behavior in distinct ways. It is easily imaginable that specific thresholds are stricter than others and that the CFC rule in total is thus more or less binding. This again could impact the investment behavior of multinationals and their investment location decisions, depending on their home country. An issue which this paper does not answer, is the question about the costs of shifting or changing the channels of passive income abroad. This study rather compares passive income in subsidiaries located in foreign countries and not directly changes in firm financing behavior in case the investment is already taken. To bring clarification to this issue, it would be useful to distinguish between changes in company behavior in case the initial investment is already taken and, for example, the subsidiary is already established. Unfortunately, this cannot be done with the given data. Another interesting query from the law perspective, that occurs out of the results of this study, is why this specific passive-to-total income threshold is mostly set at 50%. There is no objective reason other than the mentioned ECJ decision in the Cadbury-Schweppes case, and countries could effectively reduce profit shifting behavior by decreasing this threshold as companies would react to this.

This study demonstrates that CFC rules lead to less passive income in low-taxed subsidiaries abroad, i.e. decrease the profit shifting behavior of multinationals. Given this, the implementation and strengthening of CFC rules in further countries would be a legitimate option to counteract the profit shifting behavior of multinationals on a worldwide scale, which could be done unilaterally. To improve these efforts, governments need to work together to find a common ground on what is considered low-tax and where to set such thresholds. This is especially important as CFC rules are designed in such a way that the means by which the profits are shifted do not matter as much as the structure of the multinationals itself. For countries with territorial CIT systems, CFC rules provide a sufficient way to tackle BEPS, by deter profit shifting behaviour in the first place, but also by potentially tax foreign subsidiaries' profits, as long as these profits presumably slipped through home taxation. For low-tax countries, by contrast, it could be beneficial to increase their CIT rate to be slightly above the low-tax thresholds of CFC laws, which would, according to the results in this study, attract more subsidiaries as well as profits from multinationals and thus raise tax revenue. This mechanism is contrary to the common view of lowering tax rates to attract shifted income. To conclude, in its function as a vessel for collecting financial income in foreign subsidiaries, CFC rules are the slim mental bridge between two monumentally disparate tax systems that try to tax receding profits from companies originally acting in their countries.

Chapter 4

Multinational ownership structures and anti tax avoidance legislation

This study investigates if controlled foreign corporation (CFC) rules influence cross-border merger and acquisition (M&A) activity on a global scale. CFC rules are one main anti tax avoidance measure and potentially lead to immediate taxation of foreign subsidiaries' income at the parent level, without the necessity of repatriation. Analyzing a large corporate M&A data set using three econometric methods, we show how CFC rules distort global ownership patterns. First, we show that the probability of being the acquirer of a low-tax target decreases if CFC rules may be applicable to this target's income. Second, we show that CFC rules alter the acquirer's choice of targets' location. Third, we show that CFC rules negatively affect the probability of being the acquirer in cross-border M&A. Altogether, our study shows that for affected acquirer countries, CFC rules lead to less M&A activity in low-tax countries due to a reduced ability to shift income.

This chapter is partly online available as the paper: von Hagen and Prettl (2018), Multinational Ownership Structures and Anti Tax Avoidance Legislation, Working Paper (SSRN) and is based on joint work with Dominik von Hagen.

4.1 Introduction

Empirical literature provides extensive evidence of tax motivated income shifting strategies within multinational enterprises (MNEs).¹ Over the past years, tax policy makers have discussed and implemented several anti tax avoidance measures to fight against income shifting as the "Base Erosion and Profit Shifting" (BEPS) project of the Organisation for Economic Co-operation and Development (OECD) or the Anti Tax Avoidance Directive (ATAD) of the European Union (EU) (European Court of Justice (2006)) show. ATAD, for example, mandates all EU member states to implement certain anti tax avoidance measures by 2019. Lobby groups claim and countries fear a competitive disadvantage of these measures.² We investigate how forthcoming regulations like these impact ownership structures of multinational corporations on a global scale.

Anti tax avoidance measures have expanded rapidly in recent years. One current example is the US 'Public Law 115-97' or 'Tax Cuts and Jobs Act' of 2017 that strengthened US controlled foreign corporation (CFC) rules by introducing the additional Global Intangible Low-Taxed Income (GILTI) rule.³ As one of the most prominent anti tax avoidance measures, CFC rules trigger tax at an MNE's parent level and usually work as follows: If an MNE's foreign subsidiary fulfills certain requirements, at least part of its income is taxed by the MNE's parent country where CFC legislation is enacted, even if no repatriation takes place.⁴ Although tax revenue of the subsidiary's country is not directly affected by CFC rules, these laws make typical income shifting strategies into low-tax countries less

¹Income shifting is understood as reducing taxable income in high-tax countries by, for example, royalty or interest payments from high-tax to low-tax subsidiaries. See, e.g., Huizinga and Laeven (2008); Weichenrieder (2009); Grubert (2012); Dharmapala and Riedel (2013); Guenther, Matsunaga, and Williams (2017). No differentiation between "income" or "profit" shifting is necessary in this paper and the reader may see those terms as equivalent.

² e.g. OECD (2015b); OECD (2015a); Mazars (2015); PwC (2015); Picciotto (2017)

 $^{^{3}}$ The CFC rules in the US are more broadly known as "Subpart F income" rules and are not as strong as many others due to the 'check-the-box rules' as described in Section 4.2.

⁴Typically, three requirements are crucial for CFC rule application in the MNE's parent country: Low taxation of the foreign subsidiary, passive income of the subsidiary, and minimum ownership in the subsidiary. There is a high degree of variation in how CFC rules are specified, e.g., regarding what is considered low taxation or regarding a passive-to-active-income ratio that may trigger CFC rule application.

attractive for an MNE (e.g., Altshuler and Hubbard (2003); Ruf and Weichenrieder (2012)), because these strategies no longer reduce the MNE's global tax burden. As such, CFC legislation attempts to mitigate income shifting behavior of MNEs, while reducing potential competitive disadvantages of purely national companies, without income shifting opportunities. In our study, we investigate to what extent CFC rules affect one important form of foreign direct investment (FDI): cross-border mergers and acquisitions (M&A).

If a firm decides to engage in tax avoidance or to extend its existing tax avoidance strategies, it could establish a foreign subsidiary in a low-tax country as an income shifting vehicle, where profits are taxed at a low rate. There are two common ways to establish a foreign subsidiary: greenfield investment in a new firm or buying an existing firm. Our study focuses on the latter. Therefore, the existence and strength of CFC rules could impact cross-border M&A and, thereby, ownership structures of MNEs and their location decisions.

We investigate to what extent CFC rules influence ownership patterns on a global scale by analyzing the effect of CFC legislation on cross-border M&A. In our diverse econometric analyses, we investigate a large data set of worldwide M&A deals with more than 14,000 observations and a hand-collected detailed CFC rule data set of 29 countries, extended by countries that do not have CFC legislation, for the period 2002 to 2014. Besides graphical analyses, we apply multinomial and binomial choice models where we control for various firm- and country-specific variables to isolate the effect of CFC rules and their changes in our sample. As our identification strategy, we use differences in low tax rate thresholds of CFC rules and in statutory corporate income tax rates (STRs), which both vary over time and between countries. We find that CFC rules impact cross-border M&A activity in two ways.

First, we detect that CFC legislation alters the acquisition behavior of low-tax targets. In particular, we observe that the probability of acquiring a low-tax target is negatively influenced by potential CFC rule application on the low-tax target's income. Our explanation for this finding is that MNEs with parents in non-CFC rule countries (non-CFC rule MNEs) calculate higher reservation prices for low-tax targets than MNEs with parents in CFC rule countries (CFC rule MNEs), because these targets may be used as valuable income shifting vehicles within non-CFC rule MNEs. CFC rule MNEs, on the other side, have to account for the possible application of CFC legislation on low-tax targets' income, which decreases after-tax cash flows. Hence, they calculate lower reservation prices for cross-border M&A with targets located in low-tax countries than non-CFC rule MNEs. However, the economic magnitude of this effect is rather low: A ten percentage-point increase in additional CFC rule taxation leads to a 0.5% lower likelihood that an acquisition takes place.

Second, we detect that CFC legislation distorts the direction of cross-border M&A between firms. In particular, we observe that if a firm acquires another non-domestic firm, CFC legislation negatively affects the M&A direction, i.e., which firm becomes the acquirer and, thereby, the parent of the newly formed MNE. This finding is in line with previous research by Voget (2011), who detects that the presence of CFC rules increases the number of headquarter relocations. However, our approach differs from Voget (2011) in two ways: (1) we use a different identification strategy as we consider cross-border M&A in general and not the specific form of headquarters relocation, and (2), we analyze M&A observations from a different and larger database.

Our paper contributes to tax research and policy considerations in three ways. First, we contribute to empirical tax research on the effects of CFC rules on firm behavior, where little research has been undertaken so far (see Section 4.2). To our knowledge, two studies show the effect of CFC legislation from two countries on passive investment in foreign subsidiaries (Altshuler and Hubbard (2003); Ruf and Weichenrieder (2012)), and another two studies show the effect of CFC rules on headquarters relocation and real investment in foreign subsidiaries (Voget (2011); Egger and Wamser (2015)). Our study focuses on the effect of CFC legislation on firm ownership patterns, which has not been investigated. As Egger and Wamser (2015) point out, the reason why there are only a few empirical studies on CFC rules may be due to the difficulty of isolating the effect of anti tax avoidance measures on MNEs

that operate in multiple jurisdictions and avail complex group interrelations with respect to, for example, financing decisions. In addition, the effect of CFC legislation is difficult to identify as the applicability of CFC rules depends on the foreign subsidiary's characteristics as well as its host-country's characteristics. To overcome these identification difficulties, we investigate the effect of CFC legislation on the decision to integrate foreign firms into an MNE, which can be clearly identified via observed cross-border M&A. Moreover, we leverage the details of each country's CFC legislation by considering individual components of these laws such as tax rate thresholds and passive-to-active-income ratio thresholds.

Second, we contribute to empirical research in the field of M&A and their tax-related determinants. Indeed, there are many empirical studies on the effect of tax regulations on M&A from various perspectives, for example, repatriation taxes (Voget (2011); Hanlon, Lester, and Verdi (2015); Edwards, Kravet, and R. Wilson (2016); Feld, Ruf, et al. (2016)), international double taxation (Huizinga and Voget (2009); Huizinga, Voget, and Wagner (2012); Hagen and Pönnighaus (2017)) or capital gains taxes (B. C. Ayers, C. E. Lefanowicz, and J. R. Robinson (2003); B. Ayers, C. Lefanowicz, and J. Robinson (2007); Huizinga, Voget, and Wagner (2018); Todtenhaupt et al. (2020)). However, besides Voget (2011), our study is the first one that compares the effect of anti tax avoidance measures, and in particular the effects of the increasingly important CFC rules, on M&A activity over various countries. Because CFC legislation is present in 29 OECD, G20, and EU countries as shown in Figure 4.1, the strand of literature dealing with location choices of MNEs and their tax-related elements is highly important.

Third, understanding how CFC rules influence M&A activity on a global scale is also of economic interest, as cross-border M&A are an important form of FDI: In 2016, the value of cross-border M&A accounted globally for 869 billion USD, which slightly exceeded the value of announced greenfield projects (828 billion USD, UNCTAD (2017)). Hence, our analysis on distortionary tax effects on cross-border M&A, and thereby on international ownership structures, is also of interest from a global economic and not only from countries'

Figure 4.1: Presence of CFC rules over time for 49 countries (OECD, G20 and EU member countries).



Notes: This figure provides an overview on how many countries have implemented CFC rules. Source: Own data collection.

tax policy perspective. While we show that these distortionary effects from CFC rules are rather small so far, and, therefore, the claimed and feared competitive disadvantage due to CFC legislation for the M&A market is rather low, the distortions may increase due to ongoing tax policy changes globally.

The remainder of this paper proceeds as follows. Section 4.2 gives a brief review of empirical tax literature on CFC rules and on M&A activity. Section 4.3 provides our analysis of the effect of CFC rules on the acquisition of low-tax targets. Section 4.4 analyzes the effect of CFC rules on the direction of cross-border M&A. Finally, Section 4.5 sets forth our conclusions.

4.2 Literature

Despite the far-reaching consequences of CFC rules on MNEs' income shifting abilities and tax burdens, empirical studies on the effects of CFC rules on firm behavior are scarce.⁵ Altshuler and Hubbard (2003) find that tightening US CFC rules in 1986 has substantially reduced tax planning opportunities with financial services firms in low-tax countries. Three years later, Grubert and Altshuler (2006) show that the so-called check-the-box rule, which may allow for an escape from CFC rules for US MNEs, negated these effects. Voget (2011) detects in his study about international taxation and the relocation of headquarters that the presence of CFC rules increases the number of those legal structure movements. For a panel of German MNEs, Ruf and Weichenrieder (2012) detect that German CFC rules are effective in reducing passive investments in low-tax countries. These studies show that CFC rules reach the intended goal of reducing income shifting opportunities with low-tax subsidiaries. However, Egger and Wamser (2015) find that German MNEs, whose subsidiaries are subject to CFC rules, also show significantly lower fixed assets in these subsidiaries. They conclude that CFC rules lead to an increase in cost of capital if subsidiaries are subject to CFC rules. Hence, by influencing real activity abroad, the application of CFC rules potentially has non-intended "real" effects. These findings contradict the theoretical thoughts from Weichenrieder (1996) who shows that certain characteristics of CFC rules, such as an accepted passive-to active-income ratio, can lower the cost of capital in foreign subsidiaries under certain circumstances. A recent study from Haufler, Mardan, and Schindler (2018) shows from a theoretical perspective that CFC rules together with thin-capitalization rules should play a role in an optimal tax mix.

⁵A typical profit shifting strategy looks as follows: An MNE equips a subsidiary in a low-tax country with intellectual property (IP) and equity. This subsidiary then may license IP to the parent or subsidiaries in high-tax countries that pay transfer prices (royalties) in exchange for using IP. Further, the low-tax subsidiary may provide debt to the parent or subsidiaries in high-tax countries that pay interest in exchange for the internal loan. Taken together, the royalty and interest expenses reduce taxable income in high-tax countries and increase income in low-tax countries.

The effect of various taxes on M&A activity has been extensively addressed in empirical literature. Hanlon, Lester, and Verdi (2015) analyze the effect of locked-out cash of US MNEs on their cross-border M&A activity. This locked-out cash is not repatriated due to high repatriation tax costs to the US as the worldwide taxation system for dividends was operative in the US until 2017.⁶ The authors show that this locked-out cash is used in foreign M&A, which is considered less value-enhancing by the market. Similarly, Edwards, Kravet, and R. Wilson (2016) find that firms with high amounts of locked-out cash engage in less profitable M&A. Feld, Ruf, et al. (2016) show that acquirers from countries operating a territorial taxation system for foreign dividends have a competitive advantage on the crossborder M&A market to acquirers from countries operating a worldwide taxation system for foreign dividends. Huizinga and Voget (2009) find that the prospect of higher international double taxation of foreign dividends decreases the probability of attracting parent firms in a cross-border M&A. Finally, several studies investigate the effect of capital gains taxes on M&A activity. Such taxes could be seen as additional transaction costs, as the seller may be subject to capital gains taxation on selling the target. Several studies show that this so-called lock-in effect affects M&A activity (e.g., B. C. Ayers, C. E. Lefanowicz, and J. R. Robinson (2003); B. Ayers, C. Lefanowicz, and J. Robinson (2007); Todtenhaupt et al. (2020); Huizinga, Voget, and Wagner (2018)). Another working paper by Ohrn and Seegert (2019) points in the same direction on the national shareholder taxation level. About ownership structures in general, for example, Badertscher, Katz, and Sonja O. Rego (2013); McGuire, Wang, and R. J. Wilson (2014); Khan, Srinivasan, and Tan (2017); McClure et al. (2018) have shown that firm shareholding and decisions are influenced by tax avoidance possibilities or vice versa. Yet, an empirical study on the effect of anti tax avoidance measures and specifically CFC rules on M&A activity has not been undertaken to our knowledge.

We aim to contribute to the scarce literature on CFC rules by investigating to what extent CFC rules affect an important form of FDI—cross-border M&A activity—which accounted

⁶Which is confirmed by another, more recent study from Nessa (2017).

for almost 1 trillion USD of FDI in 2016 (UNCTAD (2017)). A comparable analysis has not yet been undertaken. In particular, in Section 4.3, we investigate how CFC rules influence the acquisition of low-tax targets that potentially fall under the scope of CFC rules. In Section 4.4, we investigate how CFC rules influence the direction of cross-border M&A between firms, i.e., which firm becomes the acquirer and, thereby, the parent of the newly formed MNE.

4.3 CFC rules and the acquisition of low-tax targets

4.3.1 Hypothesis development

Non-CFC rule MNEs face fewer constraints in implementing income shifting strategies within their group than CFC rule MNEs, because CFC rules aim at income shifted to low-tax subsidiaries within the MNE and, thereby, make typical income shifting strategies less attractive for an MNE.⁷ Generally CFC rules work as follows: The foreign entity has to be (1) controlled by the parent, (2) located in a low-tax country and often, (3) generate some passive income.⁸Following the argumentation and findings of Egger and Wamser (2015), CFC rules even increase the cost of capital of subsidiaries that fall under the scope of CFC rules. Consequently, it should be less attractive for a CFC rule MNE to acquire a low-tax target that may fall under the scope of CFC rules compared to a non-CFC rule MNE. Put differently, for a non-CFC rule MNE, a low-tax target could function—in addition to other synergies—as

⁷In our analysis on the effects of CFC rules on cross-border M&A activity, we consider CFC rules in the country of the MNE's parent to be relevant. The reason is straightforward: On the one side, a non-CFC rule MNE gets into a worse tax position if the acquisition is done via a CFC rule subsidiary; hence, the MNE would not acquire through this subsidiary. In support of this reasoning, Lewellen and L. Robinson (2013) find that the likelihood of choosing a subsidiary as a holding firm within an MNE is significantly lower if that subsidiary resides in a CFC rule country. On the other side, a CFC rule MNE does not get into a better tax position if the acquisition is done via a non-CFC rule subsidiary, because the parent's CFC rule would overall still be applicable in the MNE.

⁸ Sometimes more requirements have to be fulfilled by the parent or the subsidiary. With the underlying data these distinctions could not be shown and, therefore and for comprehensibility reasons, those will not be taken into account. Also, other mentioned studies about CFC rules observe only these three crucial factors, or less, and our results are significant at this level as well.

an income shifting vehicle within the MNE. This additional function could make a candidate target more valuable for this MNE compared to a CFC rule MNE without such income shifting opportunities. Due to this competitive advantage, non-CFC rule MNEs may calculate higher reservation prices for foreign low-tax targets compared to CFC rule MNEs. We, therefore, hypothesize the following, stated in alternative form:

Hypothesis 1a: In a set of candidate acquirers from various countries bidding for a given foreign low-tax target, the probability of being the actual acquirer is higher for a non-CFC rule MNE compared to an MNE that potentially has to apply CFC rules on this target's income.

Hypothesis 1a investigates the influence of CFC rules on the likelihood of acquiring a given target that acquirers from various countries bid for. We also take the "opposite" perspective that a given acquirer has the choice to buy a target out of a pool of targets from various countries. Based on the reasoning above – it is less attractive for a CFC rule MNE to acquire a low-tax target that may fall under the scope of CFC rules compared to a potentially higher taxed target that does not fall under the scope of CFC rules – we hypothesize the following, stated in alternative form:

Hypothesis 1b: In a set of candidate targets from various countries, the probability of being the actual target from a given acquirer is lower for targets that potentially fall under the scope of CFC rules of this acquirer compared to targets that do not fall under the scope of CFC rules of this acquirer.

4.3.2 Empirical approach

4.3.2.1 Acquirer perspective

Our empirical approach to analyze Hypothesis 1a, i.e., the probability of being the actual acquirer country among several candidate acquirer countries, follows the common assumption in M&A literature that M&A reflect synergies from combining two firms with all assets being priced at their fair value (e.g., Mitchell and Mulherin (1996); Becker and Fuest (2010); Feld, Ruf, et al. (2016)) where

$$V_{ijk} = \alpha CFC_{ijk} + \beta^T \boldsymbol{x}_{ijk} + \varepsilon_{ijk}$$

$$\tag{4.1}$$

is the value of target k in country j if it was owned by an acquirer from country i.⁹ The term CFC_{ij} reflects the higher burden of potential taxation of target income due to CFC rules in the acquirer country i if the target is located in country j. The variable vector \boldsymbol{x}_{ijk} and the residual ε_{ijk} contain various observable and unobservable variables to capture owner-country-specific synergies realized through a potential M&A. Coefficients α and β^T are the estimated parameters. This approach builds on the methodology used by Feld, Ruf, et al. (2016), where the target is the same for every concerned potential M&A; therefore, we automatically account for target firm, target country, and year fixed effects as they equally affect all candidate acquirers. Hence, these fixed effects do not need to be included. To control for acquirer country differences and specific effects, we include acquirer firm controls and account for country-pair specific effects.

We use the fact that a foreign firm from country i will acquire a target if the value for this target is higher than for any other candidate acquirer from country h, i.e.,

$$V_{ijk} \ge V_{hjk}, \forall h \in (1, \dots, I), \qquad (4.2)$$

where I indicates the number of candidate acquirer countries. We analyze the probability that a particular acquirer buys a target, depending on potential application of CFC rules in the country of that particular acquirer and given that we know that the transaction takes place, which is given by:

⁹ We suppress a time subscript t in the interest of readability of the model.

$$P(V_{ijk} > V_{hjk} | CFC_{1jk}, \boldsymbol{x}_{1jk}, \dots, CFC_{Ijk}, \boldsymbol{x}_{Ijk}) = \frac{exp(\alpha CFC_{ij} + \beta X_{ijk})}{\sum_{l=1}^{I} exp(\alpha CFC_{lj} + \beta X_{ljk})} \forall h, \quad (4.3)$$

where I stands for the candidate acquirer countries.¹⁰ Expression 4.3 considers a choice model assuming that M&A reflect synergies from combining two firms and that acquirers value the individual firms and the M&A correctly at their fair value. In particular, the dummy variable in this choice model takes the value of 1 if acquirer *i* chooses alternative *k* in country *j*. Using conditional logit and mixed logit regression models, we aim to calculate Expression 4.3.¹¹ Due to potential correlation between alternatives, the mixed logit approach with random drawing of observations allows us to model (i) random variations in the response probability to changes in variables, (ii) unrestricted substitution patterns, and (iii) correlated unobserved factors.

Identification

Our identification strategy is mainly based on acquirer CFC rules and target STRs. In particular, whether a certain target is potentially treated by CFC rules is due to substantial variation of CFC rules among candidate acquirer countries and, in addition, variation in STRs among target countries. For example, if a candidate acquirer country lowers the low tax rate threshold of its CFC rules, some targets in various countries that were previously captured by the CFC rule are now not affected anymore. Whereas, if a target country lowers its STR, targets located in this country may now be considered low-tax targets by some CFC rules. Figure C.1 in Appendix C.2 shows a stylized variation of the main identification, which

 $^{^{10}}$ As in Feld, Ruf, et al. (2016), at this point it is sufficient to analyze matching target firms with acquiring countries instead of the matching of target firms with particular acquiring firms, for which the construction of an appropriate choice set would be challenging and we do not have data in this regard. Instead, the accounted country-specific effects include variations in the number of candidate acquiring firms across countries.

¹¹ The presented multinomial choice model is based on Feld, Ruf, et al. (2016), p. 15, and can be understood as the polar case of a zero-sum world in which the gain of one acquirer is automatically the loss of all other acquirers.

captures variation observed in our data set, i.e., changes in CFC rules, their application, and STRs between and within countries over time.¹²

In our first approach, the difference between CFC rules among candidate acquirer countries is shown by a simple dummy variable. This dummy variable is coded one if a CFC rule is enacted in acquirer country i and is potentially applicable on target income, i.e., the STR in target country j is below the minimum low tax rate threshold of the CFC rule of the candidate acquirer country i. Almost all observed CFC rules have such a threshold to determine whether a foreign subsidiary's country is considered a low-tax country. Hence, the first variable of interest is constructed as

$$CFC^{dummy} = \begin{cases} 1, \text{ if } t_{i_{\text{threshold}}} > t_{j} \text{ or country } i \text{ applies CFC} \\ \text{rules without a tax rate threshold} \\ 0, \text{ otherwise,} \end{cases}$$
(4.4)

where $t_{i_{threshold}}$ is the tax rate threshold of the CFC rule of the candidate acquirer country i and t_j is the STR in the target country j.

In this first approach, the treatment effect is assumed to be homogeneous, as we are pooling treatments of different intensities. In a second approach, we consider heterogeneity by using the tax rate differential between the home and host countries as a finer metering of the treatment. In particular, we consider the additional taxes payable due to CFC rule application if the target is used as an income shifting vehicle¹³:

 $^{^{12}}$ The reasons for CFC rule changes are manifold. On occasion, CFC rules themselves get changed due to government policy reasons such as protecting their tax base. Usually, these CFC rule implementations and law changes result from self inflicted policy reasons, but sometimes they are stipulated by others (for example, by the mentioned intergovernmental BEPS and ATAD initiatives). The mentioned low-tax threshold of CFC rules is often bound to the country's STR in a way that this threshold is, for example, set at 60% of the acquirer country's own STR. If that STR changes, what happens quite frequently in the observed time span, the threshold changes as well, which affects then only some potential target countries and helps our identification.

 $^{^{13}\}tau_i - \tau_j$ (and not fully τ_i) are the additional taxes because the observed CFC rules grant a credit for the taxes paid by the foreign subsidiary in its host country.

$$CFC^{diff} = \begin{cases} \tau_{i} - \tau_{j}, \text{ if } t_{i_{\text{threshold}}} > t_{j} \text{ or country } i \text{ applies CFC} \\ \text{rules without a tax rate threshold} \\ 0, \text{ otherwise,} \end{cases}$$
(4.5)

where $\tau_i(\tau_j)$ is the statutory tax rate in the candidate acquirer (target) country. For both approaches, we expect a negative sign of the regression coefficients α according to Hypothesis 1a and 1b derived in Section 4.3.1.

In our robustness test, we check whether our results are robust to considering effective average tax rates (EATRs) as CFC rules usually take into account the effective tax burden of the foreign low-tax subsidiary.¹⁴ Because we do not observe the effective tax burden of the targets, we use country-level EATRs from the Oxford University Centre for Business Taxation to determine whether a target may fall under the scope of CFC rules:

$$CFC^{diffEATR} = \begin{cases} \tau_{i} - \tau_{j}, \text{ if } t_{i_{\text{threshold}}} > t_{j_{EATR}} \text{ or country } i \text{ applies CFC} \\ \text{rules without a tax rate threshold} \\ 0, \quad \text{otherwise.} \end{cases}$$
(4.6)

In a further robustness test, we consider the scope of income included by the CFC rule. Although some CFC rules only include passive income of the subsidiary, other CFC rules include passive and active income. Therefore, we let the treatment effect differ in this regard:

 $^{^{14}}$ For more about effective tax rates see, for example, Dyreng, Hanlon, et al. (2017).

$$CFC^{taxbase} = \begin{cases} \tau_{i}, \text{ if } t_{i_{thresh}} > t_{j} \text{ or country } i \text{ applies CFC} \\ \text{rules without a tax rate threshold} \\ \text{and has a full income tax base} \\ \frac{(\tau_{i} + \tau_{j})}{2}, \text{ if } t_{i_{thresh}} > t_{j} \text{ or country } i \text{ applies CFC} \\ \text{rules without a tax rate threshold} \\ \text{and has a passive income tax base} \\ \tau_{j}, \text{ otherwise.} \end{cases}$$
(4.7)

According to this differentiation, all targets are taxed at their STR. Further, this differentiation takes into account the additional CFC rule tax burden – assuming that active and passive income in the target are at the same height – in the following way: If CFC rules include the full target income once triggered, the total tax burden is set to the acquirer STR. If CFC rules include only target's passive income once triggered, the total tax burden is set to the average between target and acquirer STR.

Following Feld, Ruf, et al. (2016) we include several control variables in our regressions. We control for STR and economic indicators, such as GDP per capita, GDP growth, stock market capitalization per GDP, and credits granted to private sector per GDP in the country of the candidate acquirer. Further, we control for several distance variables, such as the distance between the acquirer and target country, whether the acquirer and target have a common language, whether the acquirer and target were ever in a colonial relation, and whether the legal systems of the acquirer and target country have common legal origins. The sources of the variables can be found in Table C.3 in Appendix C.2.

4.3.2.2 Target Perspective

The approach presented above takes an *acquirer perspective* by analyzing why a given target is bought by an acquirer from a specific country (Hypothesis 1a) for tax or specifically CFC rule reasons. In a second analysis, we follow the same logic but take a *target perspective* by analyzing the influence of CFC rules on the decision of why a given acquirer chooses to buy a target from a specific country (Hypothesis 1b).¹⁵

Building on Expression (1), we use the fact that a foreign firm will acquire a target in country j if the value for this target is higher than for any other candidate target from country g, i.e.,

$$V_{jik} \ge V_{gik}, \forall g \in (1, \dots, J), \qquad (4.4)$$

where J indicates the number of candidate target countries. We analyze the probability that an acquirer buys a particular target, depending on potential application of CFC rules in the country of the acquirer and given that we know that the transaction takes place, which is given by:

$$P\left(V_{jik} > V_{gik} | CFC_{1ik}, \boldsymbol{x}_{1ik}, \dots, CFC_{Jik}, \boldsymbol{x}_{Jik}\right) = \frac{exp(\alpha CFC_{ji} + \beta X_{jik})}{\sum_{l=1}^{J} exp(\alpha CFC_{li} + \beta X_{lik})} \forall g, \quad (4.5)$$

where J stands for the candidate target countries.¹⁶ Expression (9) considers again a choice model assuming that M&A reflect synergies from combining two firms and that acquirers value the individual firms and the M&A correctly at their fair value. In particular, the dummy variable in this choice model takes the value of 1 if acquirer *i* chooses target *k* in country *j*. Using conditional logit and mixed logit regression models, we aim to calculate

¹⁵Such a target perspective is also taken by Arulampalam, Devereux, and Liberini (2019)

 $^{^{16}}$ As in Feld, Ruf, et al. (2016) at this point it is sufficient to analyze the matching of target firms with acquiring countries instead of the matching of target firms with particular acquiring firms, for which the construction of an appropriate choice set would be challenging and we do not have data in this regard. Instead, the accounted country-specific effects include variations in the number of candidate target firms across countries.

Expression 4.5.¹⁷ Due to the correlation between alternatives, the mixed logit approach with random drawing of observations allows us to model (i) random variations in the response probability to changes in variables, (ii) unrestricted substitution patterns, and (iii) correlated unobserved factors. We use the same CFC variable differentiation method as in the acquirer perspective described above with the same identification strategy.

We include several control variables as well in this perspective, following Feld, Ruf, et al. (2016) and Arulampalam, Devereux, and Liberini (2019). We control for STR and economic indicators, such as GDP per capita, GDP growth, stock market capitalization per GDP, and credits granted to private sector per GDP in the country of the candidate target. Further, we control for several distance variables, such as the distance between the acquirer and target country, whether the acquirer and target have a common language, whether the acquirer and target were ever in a colonial relation, and whether the legal system of the acquirer and target country have common legal origins. Additionally, we include variables to control for the institutional framework of the candidate target country, such as corruption control, business start-up cost, unemployment rate and number of listed domestic firms, and the mentioned fixed effects. The sources of the variables can be found in Table C.6 in Appendix C.2.

4.3.3 Data

Data for the empirical analysis are taken from the Thomson Financial SDC database, which contains worldwide M&A transactions. We have selected all completed M&A for the period 2002 to 2014 through which majority control (>50%) of the targets has been attained.¹⁸ Further, for each M&A, country of the acquirer ultimate parent, direct acquirer, target ultimate parent and direct target must be given.¹⁹ In addition, we require that the acquirer

 $^{^{17}{\}rm The}$ presented multinomial choice model builds on Feld, Ruf, et al. (2016) and Arulampalam, Devereux, and Liberini (2019).

 $^{^{18}}$ All observed CFC rules have a participation threshold below or equal to 50% so that the majority control requirement of CFC rules is always fulfilled.

¹⁹Throughout our paper, we use the terms "ultimate parent" and "parent" synonymously.

ultimate parent and the target reside in different countries and that the acquirer ultimate parent and direct acquirer reside in the same country to reduce the possibility of a subsidiary in a third country involved in the M&A. To keep the mixed logit regressions computationally feasible, the set of considered candidate acquirer countries (Hypothesis 1a) or candidate target countries (Hypothesis 1b) is restricted to the 30 most frequent acquirer or target locations.²⁰ Furthermore, the additional observations per potentially included country are very low. These restrictions leave a sample of 14,421 cross-border M&A involving 55 countries to investigate Hypothesis 1b. Table C.2 and Table C.2 give an overview over the number of acquirer ultimate parents and targets in the respective cross-border M&A sample per country. In line with Di Giovanni (2005), we observe that countries with the largest financial markets have the most observations in both samples. Further, these tables provide information on whether CFC rules are implemented in those countries.

Data on CFC rules are based on IBFD (2002-2016), various corporate tax guides (Ernst & Young (2004-2016); Deloitte (2015); KPMG (2003-2018)), and the specific tax law of each country. We sampled various dimensions of CFC rules for the period 2002 to 2014, such as:

- tax rate threshold that triggers CFC rule,
- country lists that trigger (blacklists) or do not trigger (whitelists) CFC rule,
- threshold for passive-to-active-income ratio that triggers CFC rule,
- whether active or only passive income of CFCs is included at the parent level, or
- significant exemptions to CFC rule.

 $^{^{20}}$ To investigate Hypothesis 1a, important control variables are missing for Guernsey, Luxembourg and Taiwan so that we effectively consider 27 candidate acquirer countries. To investigate Hypothesis 1b, important control variables are missing for Indonesia and Sweden so that we effectively consider 28 candidate target countries.

4.3.4 Results

4.3.4.1 Graphical analysis

In this section, we graphically analyze whether acquisition behavior is affected by CFC rules with raw data. In particular, Figure 4.2 shows variation in acquisition behavior of acquirers from countries with and without CFC rules via density distributions. We observe that acquirers from CFC-rule countries tend to buy targets in high-tax countries, whereas acquirers from non-CFC rule countries tend to buy targets in low-tax countries.



Figure 4.2: Distribution of target country STR.

Notes: This figure shows the distribution of target country STR depending on whether the acquirer country applies CFC rules or not. It is clearly visible that acquirers from CFC rule countries acquire less low-tax targets than acquirers from non-CFC rule countries. *Source: M&A data set.*

Further, we analyze the acquisition behavior of acquirers from CFC rule countries regarding targets that have a lower STR than the acquirer. In particular, Figure 4.3 shows the distribution of targets depending on whether their STR is below or above the low tax rate threshold of the acquiring country's CFC rule. One can see that observed acquisitions increase significantly if the target is located in a country slightly above the low tax rate threshold. The summed up number of acquisitions included in the 5 percentage point range above the low tax rate threshold accounts for more than 40% of all observed acquisitions.



Figure 4.3: Distribution of cross-border M&A for acquirers from CFC rule countries.

Notes: This figure shows the distribution of acquired targets around the low tax rate threshold of CFC rules if target STR is lower than acquirer STR. It is clearly visible that acquirers from CFC rule countries acquire less targets if these targets have an STR below the low tax rate threshold. *Source: M&A data set.*

This indicates that acquirers from CFC rule countries choose targets with an STR slightly above the low tax rate threshold, most likely to facilitate tax savings via income shifting opportunities as there are no other obvious or known reasons for acting that clearly on a random threshold.²¹ In other words, these acquirers can shift income to lower taxed countries without potential CFC rule application; however, they are somewhat restricted in that behavior compared to acquirers from non-CFC rule countries. This may indicate that

²¹For anecdotic evidence, we asked accountants, participants at conferences and workshops, as well as colleagues, but no other reason than profit shifting behavior seems to be plausible to explain this behavior.

CFC rules could lead to overall higher global taxation of firm profits due to tighter income shifting possibilities.

Figure 4.4: Distribution of cross-border M&A for acquirers from CFC rule countries within the EAA before and after the Cadbury-Schweppes ruling in 2006.



Notes: This figure shows the distribution of acquired targets around the low tax rate threshold of CFC rules if target STR is lower than acquirer STR and acquirer and target reside within the EEA. It is clearly visible that acquisitions after the Cadbury-Schweppes ruling in 2006 increased in low-tax countries. *Source:* $M \mathscr{C}A$ *data set.*

Finally, we investigate the issue of potential non-application of CFC rules within the European Economic Area (EAA) due to the Cadbury-Schweppes ruling of the European Court of Justice in 2006. This ruling triggered a substantial mitigation of the application of CFC rules within the EEA. In simple words, the low tax rate threshold of CFC rules could be circumvented by a potential acquirer inside the EEA if the EEA target was still in compliance with another, less rigorous threshold about the passive-to-active-income ratio of that target. In line with this argumentation, Ruf and Weichenrieder (2013) find evidence for a relative increase in passive investments in low-tax EEA subsidiaries and a parallel decrease

in passive investments in non-EEA subsidiaries. Figure 4.4 shows that acquirers from the EAA acquired more low-tax EAA targets after 2006.

Taken together, the graphical analysis suggests that acquirers are influenced by CFC rules in their acquisition behavior.²² In the following, we investigate whether this graphical evidence is confirmed in a multivariate regression analysis.

4.3.4.2 Acquirer perspective

Results

Table 4.1 presents the baseline results of different multinomial choice models to test Hypothesis 1a on the influence of CFC rules on the likelihood of being the acquirer country of a given target (acquirer perspective). For each deal, the dependent variable equals one for the actual acquirer country of origin and zero for all other counterfactual acquirer countries. For definitions, data sources, and summary statistics of all variables see Table C.3 in Appendix C.2.

In the conditional logit regression (1), CFC^{dummy} from expression (4) is the variable of interest, which indicates potential taxation via CFC rules in the acquirer country via a dummy variable approach. We observe a negative coefficient, which suggests that potential taxation in the acquirer country due to CFC rule application has a negative influence on the probability of being the acquirer country for a given target. To be more specific, we consider CFC^{diff} from expression (5) in regression (2). CFC^{diff} is a continuous variable and takes values between 0 and 0.409; it measures the magnitude of a potential additional tax burden due to CFC rule application and the coefficient is significantly negative. The substantially lower *p*-value of CFC^{diff} (p<0.000%) compared to CFC^{dummy} (p=19.9%) is probably due to introducing heterogeneity to the treatment effect by considering the specific tax rate differential between the acquirer and target country in case CFC rules apply. The

 $^{^{22}}$ Due to data restrictions, we are not able to analyze if the acquired targets are actually used as profit shifting vehicles. However, there are no other obvious reasons—besides the presented ones—for the observed behavior in the shown graphs as CFC rules do not affect foreign subsidiaries if – put simply – no profit shifting takes place. For more details of how CFC rules work see Section 4.3

coefficient of -1.4569 implies as average partial effect that if the target is potentially treated by CFC rules and the difference between acquirer STR and target STR increases by 1%, the likelihood of acquiring this target decreases by 0.05%.

Taken together, we provide evidence that potential CFC rule application on a target's income reduces the probability of acquiring this target; this finding supports Hypothesis 1a. However, the calculated economic effect seems to be very low for small STR differences. Therefore, countries should not expect large negative effects of CFC rule implementation on their MNE's cross-border M&A activity. Besides the following robustness tests, further reassurance is served in the next Section 4.3.4.4.

As argued in Feld, Ruf, et al. (2016), a violation of the assumption of the independence of irrelevant alternatives (IIA) in the conditional logit model could be problematic because estimates may be biased. Further, and in our case potentially even more important, there may be unobserved heterogeneity in how CFC rules affect acquirers' target valuation. To account for such heterogeneity across firms in terms of M&A decisions and to address the IIA assumption, we randomize this heterogeneity and assume it to be normally distributed. Consequently, we randomize our variables of interest by using a mixed logit estimator. This randomization follows a normal distribution with mean g and covariance W; the parameters are estimated by simulated maximum likelihood with 50 Halton draws.²³ In our mixed logit regressions, we observe that the estimated standard deviations of the normal distribution are highly significant; therefore, we prefer this approach and apply mixed logit regressions in the remaining regressions.

In regression (3), we observe that applying the mixed logit model does not change the basic results as CFC^{diff} remains significantly negative at the 1% level and quantitatively stable. In regression (4), we cluster the standard errors at the target-country/year level and observe that CFC^{diff} is significant at the 5% level.

 $^{^{23}}$ In untabulated regression results, we find that using 100 Halton draws produces very similar results in both the acquirer and target perspective; these results are available upon request.

Most control variables are highly significant and show the expected signs. Regarding STR, we find a negative effect on the likelihood to be the successful bidder if the bidder is located in a high-tax country. This finding is in line with Becker and Riedel (2012), who find a negative effect of parent STR on investment in foreign subsidiaries. Helpman (2014) show that the productivity level of firms influences their investments abroad and firms with the highest productivity engage in FDI. Similar to other studies, we use GDP per capita and GDP growth as proxies for productivity levels in an acquirer country and find that GDP per capita has a significantly positive coefficient, while GDP growth is insignificant. Hence, a high level of GDP per capita has a positive impact on cross-border M&A activity. Stock market capitalization per GDP has the expected positive coefficient, which indicates that well-developed stock markets in the acquirer country offer good financing conditions to raise capital to fund cross-border M&A. The size of the private credit market has no significant effect. Cross-border M&A literature finds that lower bilateral transaction costs between the acquirer and target due to less cultural and geographic distance positively affect M&A activity (e.g., Di Giovanni (2005)). In line with these findings, we observe that the distance, a common language, past colonial relation, and a common legal system show the expected signs and are highly significant.

Robustness tests

Table 4.2 provides the results of our check on whether our baseline results are robust to specification variations. In regression (1), we include a dummy variable capturing the unilateral method (i.e., the credit or exemption method on foreign dividends) to avoid double taxation on foreign dividends which was research specifically in the work of Feld, Ruf, et al. (2016). A country's method to avoid double taxation could be potentially correlated with whether or not this country has CFC rules. The reason is as follows: If a country taxes foreign dividends under the credit method system, income shifted to tax havens will theoretically— be ultimately taxed on profit repatriation. However, the important difference between CFC rules and taxing foreign dividends taxation is the timing of taxation: While under CFC rules distinct foreign profits are immediately taxed at the parent level irrespective of dividend distribution, taxation under the pure credit method system can be deferred by the parent company until the actual dividend distribution taxation takes place. Under the exemption method system, profits shifted to tax havens are not taxed upon repatriation and the country may be more prone to introduce CFC rules. Indeed, under the Tax Cuts and Jobs Act of 2017, the US changed its international corporate tax system towards the exemption method system and strengthened its former rather weak CFC rules (as described above) by introducing the GILTI rule. To control for this potential interdependency, we include a variable for the method *ExemptionMethod*, which is one (zero) if the acquirer applies the territorial/exemption (worldwide/credit) tax system, to avoid double taxation, and the coefficient of CFC^{diff} remains significantly negative; however, the coefficient decreases by around half. The significantly positive coefficient of *ExemptionMethod* indicates that the likelihood of being the acquirer increases if the acquirer resides in a country that exempts foreign dividends of the target from taxation, which is in line with the result of Feld, Ruf, et al. (2016).
Explanatory variables	(1)	(2)	(3)	(4)
	Conditional logit	Conditional logit	Mixed logit	Mixed logit
CFC^{dummy}	-0.0523^{a}			
	(0.0407)			
CFC^{diff}		-1.4569^{***}	-1.2387^{***}	-1.2387^{**}
		(0.3277)	(0.3482)	(0.5606)
STR	-2.0538^{***}	-1.7568^{***}	-2.0903***	-2.0903**
	(0.6319)	(0.633)	(0.6442)	(0.8423)
GDP per capita	1.0541^{***}	1.0452^{***}	1.1104^{***}	1.1104^{***}
	(0.1619)	(0.1625)	(0.1652)	(0.2118)
GDP growth	-0.0034	-0.0032	-0.0041	-0.0041
	(0.0076)	(0.0075)	(0.0076)	(0.0099)
$Stock\ market\ capitalization$	0.0005^{***}	0.0005^{***}	0.0005^{***}	0.0005
per GDP	(0.0002)	(0.0002)	(0.0002)	(0.0003)
Size of private credit market	0.0007	0.0006	0.0007	0.0007
	(0.0006)	(0.0006)	(0.0006)	(0.0011)
Distance	-0.5852***	-0.5789^{***}	-0.5906^{***}	-0.5906***
	(0.0114)	(0.0115)	(0.0119)	(0.0217)
Common language	1.8148^{***}	1.8112^{***}	1.8494^{***}	1.8494^{***}
	(0.062)	(0.062)	(0.0629)	(0.1289)
Past colonial relationships	0.3020^{***}	0.2868^{***}	0.2994^{***}	0.2994^{***}
	(0.036)	(0.0359)	(0.0364)	(0.0569)
Common legal system	0.1029^{***}	0.1145^{***}	0.1117^{***}	0.1117^{**}
	(0.0251)	(0.0252)	(0.0254)	(0.047)
Acquirer country FE &				
target country FE &	YES	YES	YES	YES
target firm FE & year FE				
Observations	$317,\!835$	$317,\!835$	$317,\!835$	317,835
Log-likelihood	-32,188	-32,178	-32,165	-32,165

Table 4.1: Effect of acquirer CFC rules on probability of being the acquirer country

Note: The table shows regressions of probability of being the acquirer country on (potential) CFC rule application; see Expression 4.3. For each deal, the dependent variable equals one if country *i* is the actual acquirers country of origin, and zero if country *h* is a counterfactual acquirer country. For variable definitions and data sources, see Table C.3. Only cross-border M&A where the direct acquirer country is equal to the acquirer ultimate parent country are considered. All regressions control for acquirer country fixed effects, which are available upon request. The variables of interest follow a random distribution in the mixed logit regressions. Regressions (1) and (2) are estimated by a conditional logit model and regressions (3) and (4) are estimated by a mixed logit model. Regression (4) is identical to regression (3) except for standard errors, which are robust to clustering on the target-country-year level. *, **, and *** denote statistical significance at 10%, 5%, and 1% levels, respectively. Robust standard errors are provided in parentheses. *a* The level of statistical significance is 19.9%.

			hrongentite of	name and man	u) Ammin l	cobustness cn	lecks)	
ory variables	(1) Controlling for exemption or credit method	(2) Using target effective average tax rate	(3) Considering EAA exemption (post 2006)	(4) Considering included income of CFC rule	(5) Randomizing STR	(6) Excl. acquirers from AU&CA&NZ	(7) Excl. acquirers from CA&UK&US	(8) OLS regression
4	-0.6035*				-1.2130*** (0.9507)	-1.6977*** (0.3500)	-1.0453*	
f EATR	(2150.0)	-1.2961*** (0.2169)			(TURG-U)	(oore n)	(0700·0)	
f E E A		(7015.0)	-1.5406^{***}					
sbase			(0.0491)	-1.7810^{***}				
mmy				(0.0990)				-0.0108^{***}
	-2.3967*** (0.6431)	-1.9075*** (0.6969)	-1.9575***	-2.0217*** (0.6499)	-2.1346^{***}	-1.6298^{**}	-1.9436^{***}	(0.0020) -0.0093 (0.0107)
ionMethod	(0.0440^{***})	(e0e0.0)	(0.0440)	(0.040.0)	(0.0412)	(0.0114)	(0.077.0)	(VETO'O)
sr capita	(0.0859) 1.2497*** (0.1661)	1.0501^{***}	1.1225*** 10.1225	1.1152^{***}	1.0906^{***}	1.1571^{***}	1.0672*** /0.1005)	0.0116^{***}
owth	(1001.0)	(0.1021) -0.0034	(0:010) -0.0040	-0.0044	-0.0046	0.0051	(0.1509) -0.0106	(0.001)
arket capitalization per GDP	(0.000)	(0.0076) $(0.0005^{***}$	(0.0076) $(0.0005^{***}$	(0.0076) 0.0005^{***}	(0.0077) 0.0005^{***}	(0.0085) 0.0004^{***}	(0.0086) 0.0003**	(0.0003) 0.0000***
private credit market	(0.0002) $0.0012*$	(0.0002) 0.0006	(0.0002)	(0.0002)	(0.0002) 0.0007	(0.0002) 0.0006	(0.0002) 0.0010	(0.000)
2	(0.0007) -0.5657***	(0.0006) -0.5890***	(0.0006) - 0.5884^{***}	(0.0006) - 0.5948^{***}	(0.0007) - 0.5919^{***}	(0.0008) -0.5696***	(0.0010) -0.6515***	(0.0000) - 0.0351^{***}
	(0.0121)	(0.0115)	(0.0119)	(0.0119)	(0.0119) 1 0500***	(0.0143)	(0.0175)	(0.007)
11 initianate	(0.0641)	(0.0625)	(0.0630)	(0.0631)	(0.0627)	(0.0676)	(0.0770)	(0.0041)
onial relationships	0.2454^{***}	0.3005^{***}	0.2971^{***}	0.3004*** (0.032E)	0.2937^{***}	0.2334^{***}	0.4303^{***}	0.0132^{***}
$n \ legal \ system$	(0.0946^{***})	(0.030^{***})	(0.0304) 0.1139^{***}	(0.1136^{***})	(0.03(1)) 0.1122^{***}	(0.1244^{***})	$(0.1925^{***}$	(0.0046^{***})
5	(0.0258)	(0.0251)	(0.0254)	(0.0254)	(0.0256)	(0.0258)	(0.0277)	(0.0010)
r country FE & target country urget firm FE & year FE	YES	YES	YES	YES	YES	YES	YES	YES
tions	294,697 30 036	317,835 32,175	317,835 32.164	317,835	317,835	243,136 25 045	151,651	317, 835
ad adiusted	000,00-	- 12,20-	-07 [,] 70-	-07, TUT	-02,104	07-07-	-10,400	0.136

Table C.3. Only cross-border M&A where the direct acquirer country is equal to the acquirer ultimate parent country are considered. All regressions control for acquirer country fixed effects, which are available upon request, and are estimated by a mixed logit model. The variables of interest follow a random distribution. Regression (1) additionally controls for double taxation avoidance method, regression (2), (3) and (4) check whether our variables of interest is robust to using effective average tax rates, considering potential non-application of CFC rules within the EEA and considering the included income by CFC rules as shown in Section 4.3.2.1. In regression (5), also *STR* follows a random distribution. Regressions (6) and (7) exclude certain countries. Regression (8) is estimated by an OLS regression; the constant is not reported but available upon request. *, **, and *** denote statistical significance at 10%, 5%, and 1% levels, respectively. Robust standard errors are provided in parentheses. equals one if country i is the actual acquirer's country of origin, and zero if country h is a counterfactual acquirer country. For variable definitions and data sources, see

In regressions (2), (3) and (4), we vary the calculation of our variable of interest by considering target effective average tax rates $(CFC^{diffEATR})$, potential non-application of CFC rules within the EEA ($CFC^{diffEEA}$), and the included income by CFC rules ($CFC^{taxbase}$). In regression (5), we additionally randomize STR and in regression (6), we exclude acquirers from Australia, Canada, and New Zealand because their CFC rules do not explicitly mention a tax rate threshold that potentially changes country-pairwise over time, from which our main identification is coming from. Regression (7) excludes the largest acquirer countries (Canada, United Kingdom, and United States), which account for around half of our observations. The exclusion of the US further checks for a potential bias due to the so-called check-the-box rule, which was introduced in the US in 1997 and may allow for an escape from CFC rules for US MNEs under specific circumstances by using hybrid entities (e.g., Sonja Olhoft Rego (2003); Grubert and Altshuler (2006); Mutti and Grubert (2009)). Finally, in regression (8), we run an ordinary least squares (OLS) regression with acquirer country, target country, and year fixed effects, i.e., assuming that the probability is a linear function of the explanatory variables. The coefficient of CFC^{dummy} is significantly negative at the 1% level. However, given that the range of probabilities of the logistic regression is from 0.01 to 0.823, assuming a linear function is not appropriate and linear probability regressions lead to biased estimates. Therefore, we do not use OLS regression in our baseline results. We observe that all robustness tests validate our baseline results, both quantitatively and qualitatively.

Explanatory variables	(1)	(2)	(3)	(4)	(5)
	Excl.	Profitable vs.	Incl. target	Incl. target	Incl. target
	control	non-profitable	assets & target	sales	EBITDA
	variables	targets	return on assets		
CFC^{diff}	-4.1258***		-3.1934***	-2.8136***	-2.1391*
	(0.3294)		(1.1995)	(0.7548)	(1.2086)
$CFC^{profitable}$. ,	-1.9250**		, ,	
		(0.9653)			
$CFC^{non_profitable}$		-5.5943***			
		(1.7488)			
STR		0.8489	0.4872	-0.6872	-0.5640
		(1.5131)	(1.5582)	(1.2818)	(1.8920)
GDP per capita		1.6639***	1.8388***	1.2574***	1.1308**
		(0.3762)	(0.3851)	(0.3246)	(0.5062)
GDP growth		0.0383**	0.0455**	0.0166	0.0272
		(0.0195)	(0.0202)	(0.0176)	(0.0258)
Stock market capitalization		0.0003	-0.0002	-0.0000	-0.0007
per GDP		(0.0004)	(0.0005)	(0.0004)	(0.0006)
Size of private credit market		0.0001	0.0003	-0.0010	-0.0008
		(0.0017)	(0.0018)	(0.0014)	(0.0023)
Distance		-0.5018^{***}	-0.4904***	-0.4932***	-0.5148***
		(0.0313)	(0.0338)	(0.0266)	(0.0422)
$Common\ language$		1.7924^{***}	1.6550^{***}	1.5999^{***}	1.4257***
		(0.1765)	(0.1951)	(0.1562)	(0.2360)
$Past\ colonial\ relationships$		0.2783^{***}	0.2070^{**}	0.1570^{**}	0.1919^{*}
		(0.0862)	(0.0921)	(0.0731)	(0.1080)
Common legal system		0.2239^{***}	0.3270^{***}	0.3013^{***}	0.3555^{***}
		(0.0654)	(0.0713)	(0.0560)	(0.0860)
Acquirer country FE &					
target country FE &	YES	YES	YES	YES	YES
target firm FE & year FE	1 20	1.20	1 20	1 20	
	015 005		F 0.000		24.405
Observations	317,835	55,715	52,809	78,495	34,405
Log-likelihood	-35,450	-5,495	-5,157	-7,715	-3,287

Table 4.3: Further robustness tests of effect of acquirer CFC rules on probability of being the acquirer country

Note: Regressions of probability of being the acquirer country on (potential) CFC rule application; see Expression 4.3. For each deal, the dependent variable equals one if country i is the actual acquirer's country of origin, and zero if country h is a counterfactual acquirer country. For variable definitions and data sources, see Table C.3. Only cross-border M&A where the direct acquirer country is equal to the acquirer ultimate parent country are considered. All regressions control for acquirer country fixed effects, which are available upon request, and are estimated by a mixed logit model. The variables of interest follow a random distribution. Regression (1) drops all control variables and regression (2) distinguishes between profitable and non-profitable targets. Regression (3) includes the interaction between acquirer country fixed effects and *TargetAssets* and the interaction between acquirer country fixed effects and *TargetEOA*. Regression (4) includes the interaction between acquirer country fixed effects and *TargetEBITDA*. The coefficients and standard errors of these interactions are shown in Table C.4 in the appendix. *, **, and *** denote statistical significance at 10%, 5%, and 1% levels, respectively. Robust standard errors are provided in parentheses.

Table 4.3 provides further robustness tests. In regression (1), we exclude all control variables except for the acquirer country fixed effects to check if there is a bias due to correlation between CFC^{diff} and the control variables. We find that CFC^{diff} decreases substantially and remains significant. Further, we check whether our results are robust to differentiating between profitable and loss-making targets in regression (2). Due to missing firm level variables, the sample decreases substantially. We find that the coefficients of $CFC^{profitable}$ and $CFC^{non-profitable}$ remain significantly negative. Interestingly, the effect is

more pronounced for loss-making targets; the difference between the coefficients is significant at a *p*-value of 1.9% (two-sided). One possible reason could be that non-CFC rule acquirers are more interested in acquiring low-tax loss-making targets than CFC rule acquirers, because non-CFC rule acquirers may shift income to the loss-making targets and, thereby, net out the losses—or even use existing loss carryforwards if possible—of these targets; CFC rule acquirer on the other hand are restricted in their shifting possibilities as, for example, due to these rules often not too much financial income is allowed to show up at the target. This explanation would be in line with e.g., Langenmayr and Lester (2018); Simone, Klassen, and Seidman (2017); Maydew (1997). Finally, regressions (3), (4) and (5) control for targetspecific financial data (total assets, return on assets, sales and earnings before interest, taxes, depreciation and amortization) by interacting these consolidated profit and loss statement and balance sheet items with each candidate acquirer country. While again the sample size decreases substantially, we observe that CFC^{diff} remains significantly negative.

4.3.4.3 Target perspective

Results

As described in Section 4.3.2.2, we analyze for each given acquirer the origin of the eventual target country among a choice set of various target countries. Table 4.4 presents the baseline results of different multinomial choice models to test Hypothesis 1b on the influence of CFC rules on the likelihood of being chosen as the target country of a given acquirer. For each deal, the dependent variable equals one for the actual target country of origin and zero for all other counterfactual target countries. For definitions, data sources and summary statistics of all variables see Table C.6.²⁴

In the conditional logit regression (1), the dummy variable CFC^{dummy} has a significantly negative coefficient, which indicates that potential CFC rule application on a candidate target's income has a negative effect on actually choosing the target country as a location.

²⁴The underlying base data in both perspectives are the same. The actual observations in both perspectives differ slightly, which is due to availability restrictions of different necessary control variables.

 CFC^{diff} is a continuous variable and takes values between 0 and 0.284; it measures in more detail the magnitude of a potential additional tax burden due to CFC rule application and – similar to the result in Section 4.3.4.2 – the significance level increases compared to the mere dummy variable approach (CFC^{dummy}). In line with Hypothesis 1b, this finding indicates that potential CFC rule application on target's income negatively influences the target location choice of a given acquirer, as intended by these deterrence laws. Therefore, from a global perspective and with an increasing number of countries introducing or strengthening CFC rules, this finding may further indicate higher overall tax revenue due to potentially less income shifting from firms in high-tax countries. The coefficient of -1.7115 is slightly larger than the coefficient under the acquirer perspective and may indicate that CFC rules have a somewhat stronger effect on target location choice than on who becomes the acquirer. The coefficient implies as average partial effect that if the target is potentially treated by CFC rules and the difference between acquirer STR and target STR increases by 1%, the likelihood of acquiring this targets decreases by 0.06%.

To cope with a possible violation of the IIA and the potentially unobserved heterogeneity in how CFC rules affect acquirers' target valuation decisions (see Section 4.3.4.2), we use again a mixed logit estimator and randomize our variables of interest in the remaining regressions. Again, we observe that the estimated standard deviations of the normal distribution are highly significant; therefore, we prefer this approach and apply mixed logit regressions in the remaining regressions. We observe a further decrease of CFC^{diff} and the significance level remains stable in regression (3) and regression (4), where we cluster the standard errors at the acquirer-country/year level. In regression (5), we again run an OLS regression and observe a significantly negative coefficient of CFC^{diff} .

Regarding significant control variables, we observe that STR has a positive effect on target location choice, which is an unexpected result as FDI literature generally suggests a negative effect of host country STR on host country investment (e.g., Feld and J. Heckemeyer (2011)). An explanation for this result could be that cross-border M&A are less sensitive to host

Explanatory variables	(1)	(2)	(3)	(4)
- •	Conditional logit	Conditional logit	Mixed logit	Mixed logit
CFC^{dummy}	-0.1078**	~ ~	~	
	(0.0450)			
CFC^{diff}		-1.7115***	-2.8880***	-2.8880***
		(0.3921)	(0.5306)	(0.8075)
STR	2.6019^{***}	2.4139***	2.0753***	2.0753**
	(0.6293)	(0.6309)	(0.6398)	(0.8535)
GDP per capita	-0.0639	-0.0388	-0.0848	-0.0848
	(0.1740)	(0.1739)	(0.1744)	(0.3059)
GDP growth	0.0142^{*}	0.0143^{*}	0.0134^{*}	0.0134
-	(0.0081)	(0.0081)	(0.0081)	(0.0112)
Stock market capitalization per GDP	-0.0003	-0.0003	-0.0003	-0.0003
	(0.0002)	(0.0002)	(0.0002)	(0.0003)
Size of private credit market	-0.0019**	-0.0019**	-0.0021***	-0.0021**
	(0.0008)	(0.0008)	(0.0008)	(0.0011)
Distance	-0.5799***	-0.5740***	-0.5736***	-0.5736***
	(0.0112)	(0.0114)	(0.0114)	(0.0188)
Common language	1.9043^{***}	1.9006^{***}	1.9162^{***}	1.9162^{***}
	(0.0639)	(0.0638)	(0.0641)	(0.1225)
Past colonial relationships	0.2992^{***}	0.2777^{***}	0.2712^{***}	0.2712^{***}
	(0.0375)	(0.0377)	(0.0378)	(0.0489)
Common legal system	0.0172	0.0311	0.0345	0.0345
	(0.0269)	(0.0271)	(0.0272)	(0.0483)
Corruption control	0.1651^{*}	0.1644^{*}	0.1600^{*}	0.1600
	(0.0859)	(0.0860)	(0.0863)	(0.1337)
Business start-up cost	-0.0073**	-0.0072**	-0.0075**	-0.0075^{*}
	(0.0033)	(0.0033)	(0.0033)	(0.0044)
Unemployment rate	-0.0004	0.0001	0.0004	0.0004
	(0.0064)	(0.0064)	(0.0064)	(0.0085)
Domestic firms	0.1775^{**}	0.1651^{*}	0.1834^{**}	0.1834
	(0.0848)	(0.0846)	(0.0848)	(0.1338)
Acquirer country FE &				
target country FE &	YES	YES	YES	YES
target firm FE & year FE	_~			
Observations	317,444	317.444	317.444	317.444
Log-likelihood	-31,158	-31,151	-31,144	-31,144

Table 4.4: Effect of acquirer CFC rules on	probability	of being	target	country
--	-------------	----------	--------	---------

Note: Regressions of probability of being the target country on (potential) CFC rule application in acquirer country; see Expression 4.5. For each deal, the dependent variable equals one if country j is the actual target's country of origin, and zero if country g is a counterfactual target country. For variable definitions and data sources, see Table C.6. Only cross-border M&A where the direct acquirer country is equal to the acquirer ultimate parent country are considered. All regressions control for target country fixed effects, which are available upon request. The variables of interest follow a random distribution in the mixed logit regressions. Regressions (1) and (2) are estimated by a conditional logit model and regressions (3) and (4) are estimated by a mixed logit model. Regression (4) is identical to regression (3) except for standard errors, which are robust to clustering on the acquirer-country-year level. *, **, and *** denote statistical significance at 10%, 5%, and 1% levels, respectively. Robust standard errors are provided in parentheses.

country STRs (e.g., Hebous, Ruf, and Weichenrieder (2011); Herger, Kotsogiannis, and McCorriston (2016) or that income shifting structures within the acquiring MNE mitigate this effect (e.g., Arulampalam, Devereux, and Liberini (2019)). Additionally, variation of STR is also used to compose our variable of interest, which may lead to interdependencies. Finally, the significantly positive effect of STR does not prove to be robust.

Regarding control variables, GDP per capita and stock market capitalization per GDP have insignificant coefficients, whereas GDP growth has a significantly positive effect in some regressions, i.e., targets located in growing economies are more likely to be acquired. Further, the control variable for the size of the private credit market has a significantly negative effect on target location choice. The explanation for this finding may be the following: If a target is located in a country with a low ratio of private credits granted to the private sector, the supply of credits may be limited. Consequently, credit supply for internal expansion is limited, which makes targets in these countries more likely to be acquired (Arulampalam, Devereux, and Liberini (2019)). Similar to the findings in Section 4.3.4.2, we observe that lower bilateral transaction costs between the acquirer and target positively affect target location choice: the distance, a common language and past colonial relationships have the expected significant coefficient; the variable controlling for a common legal system has an expected positive though insignificant estimate. Finally, the control variables for the institutional framework in the candidate target country have significant explanatory power. A high degree of corruption control, a large number of listed firms, and low business startup cost increase the chances to be chosen as a target location; unemployment rate has an insignificant effect.

Robustness tests

In Table 4.5, we provide similar robustness tests as in Table 4.2 and yield similar results. Regressions (1), (2), and (3) take into account target effective average tax rates $(CFC^{diffEATR})$, potential non-application of CFC rules within the EEA $(CFC^{diffEEA})$ and the included income by CFC rules $(CFC^{taxbase})$. In regression (4), we additionally randomize STR and in regression (5), we exclude acquirers from Australia, Canada, and New Zealand because their CFC rules do not explicitly mention a tax rate threshold. Regression (6) excludes the largest target countries (Germany, United Kingdom, and United States), which account for almost half of our observations. In regression (7), we include a variable controlling for the extent of business disclosure as a further variable for the institutional framework in the candidate target country. This variable is not included in our baseline results because its inclusion significantly drops the observation number. Finally, in regression (8), we run an ordinary least squares (OLS) regression. The coefficient of CFC^{dummy} is significantly negative at the 1% level; however, given that the range of probabilities is from a 0.01 to 0.779, linear probability regressions lead to biased estimates. Therefore, we again do not use OLS regression in our baseline results. We observe that all robustness tests resemble our baseline results, both quantitatively and qualitatively.

Table 4.6 provides further robustness tests yielding similar results as presented in Table 4.3. In regression (1), we exclude all control variables except for the target country fixed effects to check if there is a bias due to correlation between CFC^{diff} and the control variables. Again, we find that CFC^{diff} decreases substantially and remains significant. Further, we check whether our results are robust to differentiating between profitable and loss-making targets in regression (2). We find that the coefficients of $CFC^{profitable}$ and $CFC^{non_profitable}$ remain significantly negative; however, in this robustness test, there is no significant difference between the coefficients of $CFC^{profitable}$ and $CFC^{non_profitable}$. Finally, in regressions (3), (4) and (5), we include acquirer-specific financial data (total assets, return on assets, sales and earnings before interest, taxes, depreciation and amortization) by interacting these consolidated profit and loss statement and balance sheet items with each candidate target country. We again observe a substantial sample decrease due missing firm level variables, but the results prove to be robust.

Table 4.5: Effect	of acquirer C	FC rules on pr	obability of be	eing a <i>target</i>	country (Rc	bustness che	$\operatorname{cks})$	
Explanatory variables	(1) Using target effective average	(2) Considering EAA exemption	(3) Considering included income	(4) Randomizing STR	(5) Excl. acquirers from	(6) Excl. acquirers from	(7) Incl. business disclosure	(8) OLS regression
CFC ^{di} ff	tax rate	(post 2006)	of CFU rule	-2.9635***	AU&CA&NZ -3.0176***		ndex -2.1462***	
				(0.5612)	(0.5315)	(0.6091)	(0.5646)	
$CFC^{diffEATR}$	-1.6836^{***}							
$CFC^{diffEEA}$		-3.2489^{***} (0.5360)						
CFC tax base			-1.3819^{a} (0 9350)					
CFC^{dummy}								-0.0163^{***}
STR	2.3923^{***}	1.9682^{***}	3.8860^{***}	1.8021^{***}	2.2549^{***}	-0.7337	1.8860^{**}	(0.0429*
GDP per capita	(0.6354) -0.0710	(0.6407) -0.0803	(1.0668) -0.1884	(0.6577) - 0.3431^{*}	(0.6744) 0.0169	(0.9266)-0.5203***	(0.7650) 0.3354	(0.0251) 0.0117^{*}
GDP $growth$	$(0.1744) \\ 0.0139^{*}$	$(0.1749) \\ 0.0137^{*}$	$(0.1798) \\ 0.0140^{*}$	(0.1848) 0.0119	(0.1825) 0.0109	(0.1978) 0.0186^{**}	$(0.2291) \\ 0.0204^{**}$	(0.0063) 0.0004
Stock market capitalization per GDP	(0.0081) -0.0003	(0.0081) -0.0003	(0.0082) -0.0002	(0.0083) -0.0001	(0.0087) -0.0003	(0.0093) -0.0002	(0.0087) -0.0002	(0.0002) - 0.0000^{**}
Size of printe credit market	(0.0002)	(0.0002)	(0.0002)	(0.002)	(0.0002) -0.0029***	(0.0002) -0.0034***	(0.0003)-0.0025***	(0.000)
Distances	(0.008) 0.5824***	(0.0008) 0.5715***	(0.0008) 0.5010***	(0.008) 0 5025***	(0.000) 0 5563***	(0.0011)	(0.0009)	(0.000)
Destance	(0.0113)	(0.0115)	(0.0122)	(0.0125)	(0.0145)	(0.0166)	(0.0123)	(0.007)
Common language	1.9332^{***} (0.0639)	1.9217^{***} (0.0640)	1.9710^{***} (0.0670)	2.0260^{***} (0.0684)	1.9892^{***} (0.0685)	2.0413^{**} (0.0805)	1.9405^{***} (0.0687)	0.1293^{***} (0.0049)
$Past\ colonial\ relationships$	0.2986***	0.2636***	0.2760***	0.2637***	0.2214***	0.3984***	0.2497^{***}	0.0051***
Common legal system	(0.0377) 0.0139	(0.0378) 0.0364	$(0.0384) \\ 0.0282$	(0.0387) 0.0315	(0.0403) 0.0482^{*}	(0.0485) 0.0919^{***}	$(0.0413) \\ 0.0162$	(0.0018) 0.0020^{*}
Communition control	(0.0269) 0.178 4*	(0.0272) 0.1504 $*$	(0.0278) 0.1641 *	(0.0280) 0.1525*	(0.0278) 0.1277	(0.0341)	(0.0291) 0.3170***	(0.0012)
	(0.0860)	(0.0865)	(0.0875)	(0.0889)	(0.0922)	(0.1113)	(0.1135)	(0.0027)
$Business\ start-up\ cost$	-0.0074** (0.0033)	-0.0074** (0.0033)	-0.0076** (0.0033)	-0.0081** (0.0034)	-0.0052 (0.0035)	-0.0071* (0.0037)	-0.0064* (0.0036)	0.0002**
$Unemployment\ rate$	0.0003	0.0002	-0.0026	-0.0032	0.0033	-0.0134^{*}	0.0055	0.0006***
Domestic firms	(0.0064) 0.1715**	(0.0064) 0.1794^{**}	(0.0065) 0.2252^{***}	(0.0065) 0.2844^{***}	(0.0067) 0.2078^{**}	(0.0081) 0.2547^{***}	(0.0071) 0.0623	(0.0002) 0.0030^{*}
	(0.0848)	(0.0849)	(0.0861)	(0.0876)	(0.0883)	(0.0907)	(0.1015)	(0.0018)
Business disclosure							0.0820 (0.0686)	
Acquirer country FE & target country FE & target firm FE & year FE	YES	YES	YES	YES	YES	YES	YES	YES
Observations	317,444	317,444	317,444	317,444	255,172	161,910	264,159	317,444
Log-likelihood R squared adjusted	-31,155	-31,140	-31,136	-31,119	-26,594	-19,327	-26, 172	0.133
Note: Regressions of probability of being the t one if country j is the actual target's country	arget country on (p y of origin, and zer	otential) CFC rule a o if country g is a	application in acqui counterfactual targ	rer country; see et country. For	Expression 4.5. For variable definition	or each deal, the d ns and data sourc	ependent variable es, see Table C.6	equals . Only
cross-border M&A where the direct acquirer of available upon request, and are estimated by :	ountry is equal to tl a mixed logit mode	ie acquirer ultimate l. The variables of i	parent country are interest follow a ran	considered. All idom distribution	regressions contro a in the mixed log	d tor target countr git regressions. Re	y fixed effects, wr gression (1), (2)	and (3)
check whether our variable of interest is robus income by CFC rules. In regression (4), also	t to using effective STR follows a ram	average tax rates, co dom distribution. F	onsidering potential Regressions (5) and	non-application (6) exclude cert	of CFC rules wit ain countries and	hin the EEA and regression (7) co	considering the in nsiders a further	ncluded control
variable (Business disclosure). Regression (8)	is estimated by an (OLS regression; the e	constant is not repo	rted but availabl	e upon request. *	;, **, and *** denc	ote statistical sign	ificance
at 10%, 5%, and 1% levels, respectively. Kobu ^a The level of statistical significance is 13.9%.	st standard errors a	re provided in parei	ntneses.					

Explanatory variables	(1)	(2)	(3)	(4)	(5)
	Excl.	Profitable vs.	Incl. acquirer	Incl.	Incl. acquirer
	control	non-profitable	assets & acquirer	acquirer	EBITDA
	variables	targets	return on assets	sales	
CFC^{diff}	-6.4155***		-3.5409***	-3.4268***	-3.2957***
	(0.4292)		(0.6830)	(0.6655)	(0.7050)
$CFC^{profitable}$		-6.4673***		. ,	. ,
		(1.6700)			
$CFC^{non_profitable}$		-7.2323***			
		(1.9287)			
STR		-1.8795	2.4216^{***}	2.7097^{***}	2.7031^{***}
		(1.7514)	(0.7889)	(0.7979)	(0.8450)
GDP per capita		0.2851	-0.1952	-0.0804	-0.3150
		(0.5944)	(0.2289)	(0.2319)	(0.2494)
$GDP \ growth$		-0.0329	0.0119	0.0107	0.0096
		(0.0227)	(0.0101)	(0.0104)	(0.0111)
Stock market capitalization per GDP		-0.0003	0.0000	-0.0002	-0.0000
		(0.0007)	(0.0003)	(0.0003)	(0.0003)
Size of private credit market		-0.0050***	-0.0027***	-0.0029***	-0.0026**
		(0.0018)	(0.0010)	(0.0010)	(0.0011)
Distance		-0.4524^{***}	-0.5450***	-0.5504^{***}	-0.5388^{***}
		(0.0303)	(0.0145)	(0.0152)	(0.0162)
$Common\ language$		2.0888^{***}	1.6471^{***}	1.5955^{***}	1.4247^{***}
		(0.1776)	(0.0896)	(0.0895)	(0.1006)
Past colonial relationships		0.2331^{***}	0.2761^{***}	0.2821^{***}	0.2991^{***}
		(0.0901)	(0.0462)	(0.0468)	(0.0489)
Common legal system		0.1076	0.1376^{***}	0.1668^{***}	0.2000^{***}
		(0.0681)	(0.0363)	(0.0363)	(0.0388)
$Corruption \ control$		0.0070	0.1240	0.0248	0.0192
		(0.2145)	(0.1076)	(0.1088)	(0.1168)
$Business \ start-up \ cost$		-0.0087	-0.0122***	-0.0110***	-0.0091**
		(0.0089)	(0.0041)	(0.0042)	(0.0046)
$Unemployment\ rate$		-0.0252	-0.0091	-0.0124	-0.0128
		(0.0160)	(0.0081)	(0.0082)	(0.0087)
Domestic firms		0.4353^{*}	0.1074	0.0945	0.1462
		(0.2224)	(0.1060)	(0.1069)	(0.1119)
Acquirer country FE &					
target country FE &	YES	YES	YES	YES	YES
target firm FE & year FE			-		
Ob a server t i server	917 444	F2 070	015 107	200 170	100 000
Upservations	317,444	53,270	210,197	200,170	180,202
LOG-IIKEIIIIOOU	-34,219	-0,020	-20,017	-19,010	-11,400

Table 4.6: Further robustness tests of effect of acquirer CFC rules on probability of being the acquirer country

Note: This table shows regressions of probability of being the target country on (potential) CFC rule application in acquirer country; see Expression 4.5. For each deal, the dependent variable equals one if country j is the actual target's country of origin, and zero if country g is a counterfactual target country. For variable definitions and data sources, see Table C.6. Only cross-border M&A where the direct acquirer country is equal to the acquirer ultimate parent country are considered. All regressions control for target country fixed effects, which are available upon request, and are estimated by a mixed logit model. The variables of interest follow a random distribution in the mixed logit regressions. Regression (1) drops all control variables and regression (2) distinguishes between profitable and non-profitable targets. Regression (3) includes the interaction between target country fixed effects and AcquirerAoA. Regression (4) includes the interaction between target focus the interaction between target country fixed effects and AcquirerSales. Regression (5) includes the interaction between target country fixed effects and standard errors of these interactions are shown in Table C.7. *, **, and *** denote statistical significance at 10%, 5%, and 1% levels, respectively. Robust standard errors are provided in parentheses.

4.3.4.4 Comparison and further robustness of both perspectives

So far, we are not able to control for country-pair specific fixed effects. The number of required dummy variables appears to be too large for the logistic regressions as the maximum likelihood estimation did not achieve convergence. But, as presented above, the OLS regressions in Table 4.2 and Table 4.5 show similar results as their logistic counterparts. Therefore, we test with further OLS regressions for various further endogeneity questions that may arise.²⁵

In Table 4.7, we include four more regressions from the acquirer and target perspective taken in our analysis above. Regressions (1) and (2) show that even with applied acquirer country-year and target country-year fixed effects, which control for changes within a country over time (e.g., the introduction or change of other anti tax avoidance rules), our results are robust and statistically significant. To interpret these robustness test results better, we used the CFC^{dummy} variable again. The interpretation of regression (1) yields that if a target is located in a low-tax country in terms of a CFC rule definition, the probability that this target is bought by an acquirer from that CFC rule country is 1.2% lower than from a non-CFC rule country. Regression (2) shows for the target perspective that a target in a specified low-tax country is chosen by an acquirer from a CFC rule country with a 1.8% lower probability.²⁶

In the following regressions we apply the most strict fixed effects that we could control for: acquirer country-year, target country-year, and country-pair fixed effects. In regressions (3) and (4), we observe non statistical significant results, which are quite smaller in size and even change signs with these strong fixed effect controls. In regressions (5) and (6), the CFC^{diff} variable is used instead as this specification includes more of the underlying heterogeneity of the observations. In this case, the coefficient for the acquirer perspective stays insignificant, but the coefficient for the target perspective shows statistical significance.

 $^{^{25}}$ Thereby, these regressions do not control for target firm specific effects anymore but different stronger fixed effect controls can by applied.

²⁶The reader should keep in mind that the preferred regression methods for binary variables are logistic methods and, therefore, the numbers presented in this robustness section should be interpreted with caution.

In Regression (1) to (6), we keep the sample size at the same level as in the regressions in the previous sub-sections. In Regressions (7) and (8), we drop that restriction and see similar results for the full sample size that is used in these calculations where no observations are dropped due to none missing control variables.²⁷

Comparing the results from Section 4.3.4.2, Section 4.3.4.3, and Section 4.3.4.4 reveals that the coefficients are always larger in the target perspective version. Additionally, one can observe in the last four regressions in this subsection that the target perspective results prove to be more robust. These findings suggest that CFC rule influence on M&A decisions is more important on the choice of targets in low-tax countries than on the question of who becomes the new parent of a new target in a low-tax country, which accounts for the two perspectives taken above. This would be in line with the intended deterring effect of CFC rules and is speaking against some competitive disadvantages arguments of lobbyists, especially as the size of the effect in the M&A market seems to be rather low as shown in this section.

²⁷Due to missing control variables in previous sub-section regressions, some observations had to be dropped.

Table 4.7: Further robustn	less tests of a	acquirer CF(C rules in th	le form of O	LS regressio	ns for acqui	rer and target	country probabilities
Explanatory variables	(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)
	Acquirer	Target	Acquirer	Target	Acquirer	Target	Acquirer	Target
	Perspective	Perspective	Perspective	Perspective	Perspective	Perspective	Perspective	Perspective
CFC^{dummy}	-0.0118^{***}	-0.0182^{***}	0.0021	0.0001				
	(0.0021)	(0.0012)	(0.0044)	(0.0023)				
CFC^{diff}					0.0008	-0.0458^{*}	0.0063	-0.0323^{*}
					(0.0374)	(0.0262)	(0.0265)	(0.0193)
Distance	-0.0352^{***}	-0.0328^{***}						
	(0.0007)	(0.0007)						
$Common\ language$	0.0963^{***}	0.1303^{***}						
	(0.0041)	(0.0049)						
Past colonial relationships	0.0132^{***}	0.0053^{***}						
	(0.0019)	(0.0018)						
Common legal system	0.0045^{***}	0.0023^{**}						
	(0.0010)	(0.0012)						
Target Country-Year FE	YES	YES	YES	YES	\mathbf{YES}	YES	\mathbf{YES}	\mathbf{YES}
Acquirer Country-Year FE	\mathbf{YES}	\mathbf{YES}	\mathbf{YES}	\mathbf{YES}	\mathbf{YES}	YES	\mathbf{YES}	\mathbf{YES}
Country-Pair FE			\mathbf{YES}	\mathbf{YES}	\mathbf{YES}	\mathbf{YES}	\mathbf{YES}	YES
Observations	317, 835	318,666	317,826	318,545	317, 826	318,545	471, 474	497, 147
R squared adjusted	0.1374	0.1321	0.1805	0.1800	0.1805	0.1801	0.1617	0.1575
Note: This table shows ordin	nary least squ	ares (OLS) reg	ressions with	various fixed e	effects as robu	stness checks.	For each deal, t	he dependent variable
equals one if country $i(j)$ is	s the actual a	cquirer's (targe	et's) country c	of origin, and	zero if country	y h (g) is a cc	unterfactual acq	luirer (target) country
in the relevant perspective s	shown alternat	ely. For varial	ole definitions	and data sour	ces, see Table	C.3 and Tabl	e C.6. Only cro	ss-border M&A where
the direct acquirer country is	s equal to the	acquirer ultim	ate parent cou	intry are consi	idered. All reg	ressions contr	ol for target coun	ntry-year and acquirer
country-year fixed effects.	In regressions	(1) and (2) , s	ome country-	pair specific c	ontrol variabl	es are include	d. In the follow	ring regressions (3) to
(8), country-pair fixed effect	ts are included	l to account fc	r these effects	s even stricter	. In regression	(5) to (8) , i	another variable	of interest with more
heterogeneity is used. The s.	ample base of	regressions (7)	and (8) is set	to its initial	size as in thes	e fixed effects	regressions no ol	bservations have to be
deleted due to potential mis	sing control ve	ariables as in t	he former logi	stic regression	s above. *, **	, and *** den	ote statistical sig	sniftcance at $10%$, $5%$,
and 1% levels, respectively.	Robust standa	urd errors are p	provided in pa	rentheses.				

4.4 CFC rules and the direction of cross-border M&A

4.4.1 Hypothesis development

In this section, we consider the direction of cross-border M&A. In particular, we investigate whether CFC rules affect the decision as to which firm becomes the parent firm of a newly created MNE through a cross-border M&A. Following the finding of Voget (2011) that CFC rules trigger the relocation of headquarters, we argue that CFC rules negatively influence the direction of a cross-border M&A between two firms from different countries, i.e., we expect that it is more probable that the non-CFC rule firm acquires the CFC rule firm. The reasoning is as follows: If the non-CFC rule firm becomes the new MNE's parent, potential (new) income shifting strategies may arise by setting up or using an already existing tax haven subsidiary within the MNE, which potentially decreases the overall tax burden. These (new) income shifting strategies would not exist if the CFC rule firm became the acquirer due to potential CFC rule application on low-tax subsidiaries' income. We, therefore, hypothesize the following, stated in alternative form:

Hypothesis 2: The probability of being the acquiring firm in cross-border M&A is higher for firms in non-CFC rule countries compared to firms in CFC rule countries.

This analysis is different to the analysis presented in Section 4.3, where we investigate whether CFC rules affect the decision to acquire a target if CFC rules are potentially applied to this target's income. By analyzing the effect of CFC rules on the direction of cross-border M&A, we consider whether CFC rules negatively affect the choice of who becomes the parent of the newly created MNE.

4.4.2 Empirical approach

To analyze the direction of observed cross-border M&A, we assume that firm a acquires firm b and that a and b do not reside in the same country. Under the assumption that M&A reflect synergies from combining these two firms and that investors value the individual firms

and the M&A correctly, it follows that the value when a acquires $b(V_{ab})$ is higher than the value when b acquires $a(V_{ba})$, i.e., $V_{ab} - V_{ba} > 0$. Based on Hypothesis 2 derived under Section 4.4.1, we argue that CFC rules have an impact on this valuation. In particular, CFC rules lead to a competitive disadvantage for parent firms as those firms have less income shifting opportunities within their group and have to fear potential CFC rule application on low-tax subsidiaries' income, at which these laws are aiming. We consider the following expression to analyze the direction in cross-border M&A, depending on the CFC rules of the two involved firms and given that we know that the transaction takes place:

$$P(V_{ab} > V_{ba}|X) = E\left(Y|\Delta CFC + \Delta X\right) = \frac{exp(\beta(\Delta CFC + \Delta X))}{1 + exp\left(\beta\left(\Delta CFC + \Delta X\right)\right)}$$
(4.6)
he dependent variable $Y = \begin{cases} 1 \text{ if } V_{ab} - V_{ba} > 0\\ 0 \text{ if } V_{ab} - V_{ba} \le 0 \end{cases}$.

Using logit regression models, we aim to calculate $P(V_{ab} > V_{ba}|X)$, i.e., we always consider the setting that *a* acquires *b* ($V_{ab} - V_{ba} > 0$ in Equation 4.6). This consideration implies that y, our dependent variable, always takes the value 1.²⁸ The variable of interest is ΔCFC , which measures the difference in CFC rules between *a* and *b*. We consider two approaches in calculating ΔCFC .

with t

First, we construct a CFC dummy variable (ΔCFC_dummy) that measures whether CFC rules are present in the residence countries of a and b. If, for example, the country of a does not apply CFC rules (0) and the country of b applies CFC rules (1) in the M&A year, ΔCFC_dummy takes the value 0-1 = -1.

Second, we consider individual characteristics of CFC rules to allow for more heterogeneity among CFC rules. We construct a CFC variable (ΔCFC_value), which is zero for non-CFC rule countries and one for CFC rule countries. In addition to that, we consider the CFC rule countries in more detail and group them regarding their CFC rule harshness

²⁸The presented binary choice model is based on the methodology used by Huizinga and Voget (2009), pp. 1229ff.

among the two main CFC rule features, which can be derived from all observed CFC rules: The lowest possible tax haven STR and the passive-to-active-income ratio accepted by CFC rules. This approach can increase ΔCFC_value up to the value 3. Among the CFC rule countries, the lowest possible tax haven STR is set to the tax rate threshold of the CFC rule.²⁹ For CFC rule countries with a tax haven STR equal or above its median value of 15%, we add 1 to ΔCFC_value . Similarly, we consider the passive-to-active-income ratio, which determines the amount of passive income that is allowed so that CFC rules are not triggered. The median value of the passive-to-active-income ratio is 10%; for CFC rule countries with a passive-to-active-income ratio below 10%, we add 1 to ΔCFC_value . If, for example, a firm residing in the Netherlands acquires a firm residing in the Republic of Korea, ΔCFC_value takes the value 0-2 = -2.

We expect a negative coefficient for both $\Delta CFC_{-}dummy$ and $\Delta CFC_{-}value$, indicating that it is more likely that the firm without CFC rules or with less harsh CFC rules becomes the acquiring firm. We are aware of the fact that these $CFC_{-}value$ variables have some subjectivity built in, but they account more precisely for the individual CFC rule considerations and variations between the countries and over time. Therefore, $CFC_{-}value$ extends our study in this last approach in a meaningful way.

Following Huizinga and Voget (2009), we control for firm characteristics and macroeconomic conditions in the two countries captured by ΔX . On the firm level, we include the firms' consolidated financial data. We control for relative size of the two firms ($\Delta Size$) and expect a positive coefficient, as larger firms are considered more likely to acquire smaller firms. $\Delta Leverage$ considers the difference in leverage ratio between the two firms. Following

²⁹For EEA member states in the years after the decision of the European Court of Justice in the case "Cadbury-Schweppes" (C-194/04) in 2006, we set the tax haven tax rate equal to the lowest STR within the EU, because since this decision, CFC rules are de facto not applicable within the EU. In support of this reasoning, Ruf and Weichenrieder (2013) and Section 3.6.2 provide evidence for an increase of profit shifting within the EEA after this decision (see also Section 4.3.4.1).

³⁰These thresholds are subjective; however, they split the CFC rule countries into two equal halves and allow a grouping of the CFC rule countries according to their relative CFC rule harshness.

Desai and Hines (2002), we argue that firms with higher leverage have lower borrowing costs. Thus, these firms have higher borrowing capacity, which makes them more likely to be the acquirer. ΔPTI measures the relative difference between pre-tax income of the two firms. Similar to our expectation of $\Delta Size$, we expect that firms with higher profits are more likely to acquire firms with lower profits.

On the country-level, we control for the difference in STRs (ΔSTR). We have no expectation on its coefficient as high-tax countries may have a better investment environment whereas low-tax countries may attract firms due to tax savings. Based on the finding of Huizinga and Voget (2009) that taxation of dividend repatriation affects M&A direction, we include the difference in both countries' double taxation avoidance method on foreign dividends (ΔDTM) , where 0 (1) stands for the credit (exemption) method. We expect a positive coefficient for this variable. We also include the two countries' relative stock market size $(\Delta StockMrk)$, which proxies for the relative ease to raise capital at stock markets and we expect a positive coefficient. In addition, we include the two countries' relative difference between domestic credits granted to the private sector ($\Delta CreditMrk$). Similar to the argumentation in Section 4.3.4.3, we argue that if a firm is located in a country with a low ratio of credits granted to the private market, the supply of credit may be limited and, hence, the possibility to finance an acquisition via credit is limited. Thus, we expect a positive coefficient. Finally, to control for the price level in an economy, we include the difference in the inflation rate (Δ Inflation) between both countries. We have a negative expectation on its coefficient.

Further, we include country fixed effects that reflect whether the country is the acquirer or the target country: For each M&A, the acquirer country gets the value of 1 and the target country gets the value of -1; all other countries get the value of 0 for the respective M&A.

Following Huizinga and Voget (2009), our logit regression is estimated using maximum likelihood estimation without a constant. The reason is straightforward: Since we always consider the setting that firm a acquires firm $b (V_{ab} - V_{ba} > 0$ in equation 4.6), the dependent variable is always one and, consequently, there is no variation in the dependent variable and the constant would be a perfect fit.

4.4.3 Data

The M&A data analyzed in this section are the same as described in Section 4.3.3 with two exceptions. First, we relax the restriction to the 30 most frequent acquirer or target locations. Second, we require that the direct acquirer and the direct target reside in the same country as their respective ultimate parent to reduce the possibility of a subsidiary in a third country being involved in the M&A. In addition, as outlined above, we need consolidated financial data of both firms as control variables, which reduces our sample to 1,199 cross-border M&A involving 30 countries.³¹ Table C.9 in the appendix gives an overview over the number of acquirer ultimate parents and target ultimate parents in this cross-border M&A sample per country. Further, this table provides information on whether CFC rules are implemented in those countries.

4.4.4 Results

Table 4.8 shows the results of the binary choice model to test Hypothesis 2 on the influence of CFC rules on the direction of cross-border M&A between two firms, i.e., which firm becomes the acquirer. For definitions, data sources and summary statistics of all variables see Table C.10 in Appendix C.3.

In regressions (1) and (2), we find that CFC rules negatively affect the probability of which firm becomes the acquirer. In particular, we find a significant coefficient at the 5% level for

³¹We experience this sharp decrease in cross-border M&A observation due to the lack of important financial control variables. However, this decrease is not due to specific countries or a specific financial control variable. Hence, we assume that the smaller sub-sample is a representative subset of the larger one and that focusing on this subset does not bias our subsequent empirical work. This argumentation follows Huizinga and Voget (2009), p. 1228, who face the same problem using firm level data in an SDC data set and who observe a similar decrease in sample size. To expand our sub-sample, we follow Huizinga and Voget (2009) and use Compustat North America and Compustat Global databases that are together global in coverage to fill-up firm level control variables. We use CUSIP and SEDOL firm identification codes to link the Compustat databases with the SDC database.

Explanatory variables	Leve	el of	Level of	f acquirer ult	. par. & targ	get ult. par.
	direct a	acquirer				
	& direc	t target				
	(1)	(2)	(3)	(4)	(5)	(6)
ΔCFC_value	-1.127**		-1.438**		-2.025^{a}	
	(0.530)		(0.701)		(1.558)	
ΔCFC_dummy		-2.027^{*}		-3.543**		-10.944***
		(1.132)		(1.754)		(2.620)
ΔSTR	0.168^{*}	0.096^{**}	0.278^{***}	0.062	0.693^{***}	0.079
	(0.086)	(0.038)	(0.105)	(0.043)	(0.254)	(0.058)
ΔDTM	-0.242	0.201	-0.910	-0.399	-1.833**	-0.881
	(0.652)	(0.671)	(0.853)	(0.879)	(0.927)	(1.040)
$\Delta Size$	5.101^{***}	5.509^{***}	5.480^{***}	5.698^{***}	7.523***	6.037^{***}
	(0.398)	(0.409)	(0.501)	(0.477)	(1.403)	(0.886)
ΔPTI	1.177^{***}	1.128^{***}	1.399^{***}	1.307^{***}	1.571	0.906
	(0.407)	(0.375)	(0.466)	(0.366)	(1.040)	(0.844)
$\Delta Leverage$	0.158^{**}	0.216^{**}	0.123^{*}	0.206^{**}	-0.098	-0.372
	(0.068)	(0.086)	(0.068)	(0.083)	(0.983)	(0.638)
$\Delta StockMrk$	4.914***	2.802^{**}	6.446^{***}	3.004^{**}	9.175^{***}	2.896
	(1.615)	(1.292)	(2.278)	(1.459)	(3.105)	(2.410)
$\Delta CreditMrk$	-6.363***	-2.533*	-8.826***	-3.069	-9.829*	0.013
	(1.848)	(1.403)	(2.851)	(1.884)	(5.900)	(4.130)
Δ Inflation	0.193	0.083	0.321	0.132	0.245	0.002
	(0.205)	(0.171)	(0.245)	(0.210)	(0.534)	(0.427)
Country fixed effects	YES	YES	YES	YES	YES	YES
Observations	$1,\!199$	1,580	989	1,305	418	492
Number of countries	30	31	30	30	29	29
Log-likelihood	-99.2	-133.6	-70.2	-100.7	-24.8	-38.1
Time period	2002 - 2014	1995 - 2014	2002 - 2014	1995 - 2014	2002 - 2014	1995-2014

Table 4.8: Effect of CFC rules on direction of cross-border M&A

Note: Logit regressions of probability of being the acquirer country on (potential) CFC rules in a crossborder M&A; see Equation 4.6. For variable definitions and data sources, see Table C.10. All regressions control for country fixed effects, which are available upon request. Regressions (1) and (2) consider M&A where the direct acquirer and direct target reside in the same country as their respective ultimate parents. Regressions (3) and (4) are the same as (1) and (2), but require that the direct acquirer and the direct target are the respective groups' ultimate parents. Regressions (5) and (6) are the same as (3) and (4), but exclude M&A involving the United States. Regressions (2), (4) and (6) consider in addition years 1995-2001; due to a lack of more detailed historic CFC rule data ΔCFC -value cannot be constructed for the time period 1995-2001. *, **, and *** denote statistical significance at 10%, 5%, and 1% levels, respectively. Robust standard errors are provided in parentheses. ^a The level of statistical significance is 19.4%.

 ΔCFC_value . This finding suggests that when two firms perform a cross-border M&A, it is less likely that the firm with the harsher CFC rule becomes the acquiring firm. For the dummy variable approach (ΔCFC_dummy), we observe a significantly negative coefficient at the 10% level. Hence, the mere presence of CFC rules seems to affect cross-border M&A direction. These results prove to be robust in regressions (3) and (4), where we analyze a slightly smaller sample by considering only cross-border M&A directly between the ultimate parents, i.e., the acquirer is the acquirer ultimate parent and the target is the target ultimate parent. In regressions (5) and (6), we consider the same setting as in regressions (3) and (4), but exclude M&A that involve the United States. We do this to check that the results are not biased by potential check-the-box rule application in the US, which may allow for an escape from CFC rules for US MNEs under specific circumstances by using hybrid entities (e.g., Grubert and Altshuler (2006); Mutti and Grubert (2009)). Although this exclusion decreases the sample by more than half, we still observe a significantly negative estimate for ΔCFC_dummy . The coefficient of ΔCFC_value remains also negative; however, its *p*-value drops to 19.4%.

Taken together, we provide evidence for Hypothesis 2 that the direction of cross-border M&A between firms is negatively affected by the presence and harshness of CFC rules. This finding contributes to previous research documenting that headquarters relocation is influenced by CFC rules (Voget (2011)). Our interpretation of this finding is that if the non-CFC rule firm acquirers the CFC rule firm, new income shifting opportunities may potentially come up within the newly formed MNE, which may decrease the tax burden in the future. If the CFC rule firm acquires the non-CFC rule firm, these income shifting opportunities are rather unattractive due to CFC rules in the new parent country. In addition, the CFC rule firm has to fear potential CFC rule application on low-tax subsidiaries' income if such subsidiaries are already present in the acquired firm. The firms involved in the M&A are quite large with an average value of total assets of the acquirers (targets) of 38.3 (2.4) bio. USD. Hence, it is reasonable to assume that at least some of the involved firms are already MNEs with implemented income shifting strategies within their group if no CFC rules are present in the ultimate parent country.

Regarding control variables, we find, as expected, that firm size has a significantly positive impact on the likelihood of being the acquiring firm and, in most regressions, firm profitability, firm leverage, STR and stock market size have a significantly positive effect on M&A direction. Credit market size has an unexpected negative effect in most regressions. We observe non-significant estimates for inflation rate and the method to avoid double taxation.

4.5 Conclusion

In this study, we investigate the impact of an increasingly important anti tax avoidance measure on cross-border M&A activity of corporations on a global scale. In particular, we consider important characteristics of CFC legislation from a variety of countries and apply different logit regression models on a large worldwide cross-border M&A data set. Considering individual M&A, we find that the probability of being the acquirer of a lowtax target decreases if CFC rules may be applicable on this target's income. This finding implies that acquirers from non-CFC rule countries have a competitive advantage in bidding for targets in low-tax countries. This is explained by a higher reservation price of these non-CFC rules acquirers due to potential firm value increasing income shifting opportunities after the M&A. Further, we show that the acquirer's location choice of a target is negatively affected if the target may fall under the scope of CFC legislation of an acquirer. The reasoning behind this result is the same as before but the underlying perspective is different. Thereby, we find evidence that CFC rules affect M&A activity on the bidding side, i.e., non-CFC rule acquirers have competitive advantages in bidding for a given target, and on the target side, i.e., low-tax targets are rather acquired by non-CFC rule acquirers. These two findings provide robust evidence that CFC legislation distorts ownership of low-tax targets although the economic magnitude of the effects is rather small. Finally, we show that CFC rules negatively affect the direction of cross-border M&A, i.e., countries with CFC legislation are less likely to attract parent firms in a newly created MNE after M&A.

However, our results should not be interpreted as suggesting that countries should get rid of CFC rules if undesired tax distortions of M&A, which can lead to ownership inefficiencies, shall be mitigated. Moreover, our findings suggest that CFC legislation seems to reach the intended goal of reducing income shifting opportunities with low-tax subsidiaries in our cross border M&A context. In other words, our results suggest that the specific way of investing in foreign low-tax countries to shift income afterwards is limited by existing CFC rules in the acquirer country. Therefore, CFC legislation can be used by countries to counteract tax avoidance behavior of their MNEs, which could result in an increase in tax revenue on an overall global scale. However, the shown effects are of small economic magnitude, which indicates that CFC rules do not distort M&A to a high degree. Therefore, national tax policy makers do not have to fear a large negative impact of CFC legislation on their MNEs' cross-border M&A activity.

Nevertheless, the parallel presence and non-presence of CFC rules across countries is problematic to a certain degree due to competitive disadvantages on the cross-border M&A market and potentially tax-biased ownership structures on a global scale. Thereby, we contribute to a strand of literature where little research has been undertaken so far. Further, our findings are particularly interesting in light of current tax policy developments. Although the BEPS project of the OECD recommends an implementation of effective CFC rules in the OECD and G20 countries (OECD/G20 (2015)), the European Council even issued a legally binding directive requiring EU member states to implement CFC legislation by 2019 (European Council (2016)). In other words, at the latest from 2019 onwards, firms residing in the EU may face competitive disadvantages in M&A activities due to tax legislation, compared to firms residing in OECD and G20 member states, which do not follow the BEPS project's suggestion to implement effective CFC rules and lower their MNEs' tax avoidance opportunities. Although our finding's magnitude based on historic data is rather small in size, it indicates that more coordination regarding countries' international tax law seems to be necessary for tax induced distortion not to be increased due to upcoming tax rule changes. This is of particular relevance if tax avoidance behavior of MNEs is considered unfavorable on a global scale and intended measures to prevent this behavior are supposed to be fruitful.

Appendix A

CFC rule characteristics and changes per country

1	
0	
\sim	
0	
÷	
4	
Q	
8	
64	
ц	
5	
-Ĕ	
_	
Ie	
Ц	
g	
Ð	
e	
В	
-8	
-	
e	
÷	
<u>ب</u>	
é	
б	
Ĩ	
ŝ	
e.	
Ξ	
nt	
n	
ō	
C	
×.	
(F)	
F	
æ	
\sim	
펑	
\mathcal{Q}	
프	
0	
Ŀ.	
ncl.	
(incl.	
s (incl.	
es (incl.	
ries (incl.	
ntries (incl.	
untries (incl.	
ountries (incl.	
countries (incl.	
7 countries (incl.	
27 countries (incl.	
or 27 countries (incl.	
for 27 countries (incl.	
e for 27 countries (incl.	
de for 27 countries (incl.	
vide for 27 countries (incl.	
lwide for 27 countries (incl.	
ldwide for 27 countries (incl.	
orldwide for 27 countries (incl.	
vorldwide for 27 countries (incl.	
worldwide for 27 countries (incl.	
ss worldwide for 27 countries (incl.	
iles worldwide for 27 countries (incl.	
rules worldwide for 27 countries (incl.	
7 rules worldwide for 27 countries (incl.	
'C rules worldwide for 27 countries (incl.	
FC rules worldwide for 27 countries (incl.	
CFC rules worldwide for 27 countries (incl.	
t CFC rules worldwide for 27 countries (incl.	
ut CFC rules worldwide for 27 countries (incl.	
oout CFC rules worldwide for 27 countries (incl.	
about CFC rules worldwide for 27 countries (incl.	
ι about CFC rules worldwide for 27 countries (incl.	
ta about CFC rules worldwide for 27 countries (incl.	
ata about CFC rules worldwide for 27 countries (incl.	
Data about CFC rules worldwide for 27 countries (incl.	
: Data about CFC rules worldwide for 27 countries (incl.	
1: Data about CFC rules worldwide for 27 countries (incl.	
A.1: Data about CFC rules worldwide for 27 countries (incl.	
A.1: Data about CFC rules worldwide for 27 countries (incl.	
le A.1: Data about CFC rules worldwide for 27 countries (incl.	
ble A.1: Data about CFC rules worldwide for 27 countries (incl.	
Table A.1: Data about CFC rules worldwide for 27 countries (incl.	

white list/ black list	wl	Ι		wl^1	wl wl	wl wl	wl	Ι		bl	Ы		4	1 1	I	bl	Ι		I	Ι	I	I	I	Ы Ы	10	I	I	on next page
passive income save harbour	5%**	I		50%	33 <i>%</i> 33 <i>%</i>	50%	50%	50% **	50%**	50% **	$50\%^{**}$	$50\%^{**}$	10%	10%	0/01	30%	10% **	10% ** 50% **	50%**	I	I	50%	50%	* * 20 * * 20 10	02/DC	1 20	0.70	Continued
EEA exemption	1	I		I	(N	I	I	$_{\rm Yes}$	Yes	Yes	$_{\rm Yes}^{\rm Yes}$	I	Yes Vec	TCS	Yes	I :	Yes Voc	Yes	Yes	Yes	I	I	- 7	Ies	I	1	
ax threshold relative % of own CIT rate	I	I	1 1	50%	80% 75%	50% 33%	33%	50% 50%	30% 50%	50%	50%	66% 50%	65%	65% 85%	07.00	50%	67%	67% 67%	53%	66.67%	66.67%	56% - 80%	57%	 200	0%.DC	61%	4970 - 3070	
low-t absolute CIT rate	0	0	0 0	12.5%	$\frac{22.5\%}{21\%}$	12.5% 7%	2%	17.4% 15.60	15.6%	14.7%	12~%	$\begin{array}{c} 21\% \\ 16.67\% \end{array}$	25%	25% 95%	0/07	13%	10.67%	10.67%	10%	12%	13.33%	20%	15%	107 - 1 - 1	10.1%	25%	20.70	
min. control	$50\%^{*}$	20%*	$\frac{10\%}{10\%}$	50%*	50%* 50%*	50% * 50% *	$50\%^{*}$	50%* 50%*	50%*	50%*	50%*	50%* 50%*	50%	50% 50%	0/00	50%	1%	1% סג% *	$25\%^{*}$	50%	50%	50%	50%	50% 50%	%ne	50%	%ne	
tax base	passive income	full income	passive income quasi passive income	full income	passive income passive income	full income full income	full income	full income	full income	full income	full income	full income full income	passive income	passive income	panteent anteend	full income	Dividends	dividends	dividends and more	full income	full income	passive income	passive income	full income	run income	full income	run income	
Years	2004 - 2014	2004 - 2014	2004 - 2008 2009 - 2014	2008 - 2014	2004 - 2005 2006	2007 - 2014 2004 - 2010	2011 - 2014	2004 2005 2008	2009 - 2011	2012 - 2013	2014	2004 2005 - 2014	2004 - 2006	2007 2008 - 2017	5TO7 - 0007	2014	2004 - 2007	2008 - 2009 ⁵ 2010	2011 - 2014	2010	2011 - 2014	2004 - 2013	2014	2004 - 2009	7010 - 2017	2004 - 2009	7010 - 2017	
Country	Australia	Brazil	Canada	China	Denmark	$Estonia^3$		Finland				France	Germany	5		Greece	Hungary			Iceland		Israel		Italy		Japan		

¹ Effective since 2009.

² No clear exemption but law adjustments for national positions.
³ The Estonian CFC rules are only applicable for Estonian individuals, not for companies.
⁴ No legally binding black or white list exists in Germany, but an unofficial list as guidance for all years.
⁵ Until the end of 2009 CFC rules affected only Hungarian individuals and no corporations.

Country	Years	tax base	TableA.1 min. control	 – continued from pre low- absolute CIT rate 	wious page tax threshold relative % of own CIT rate	EEA	passive income save harbour	white list/ black list
Lithuania	2004 - 2014	passive income	50%	11.25% - 15%	75%	-		bl & wl
Mexico	2004 2005 - 2014	quasi passive income quasi passive income	50%	$^-$ 22.50% - 21%	_ 75%		20% 20%	bl bl
New Zealand	2004 - 2009 2010 - 2014	quasi passive income quasi passive income	50%	1 1	1 1	1 1	- 5%**	wl _
Norway	2004 - 2006 2007 - 2013 2014	full income full income full income	50% 50% 50%	$\frac{18.67\%}{18.67\%}$	%29.99 86.67% 86.67%	${ m Yes}$	9	1 1 1
Portugal	2004 - 2006 2007 - 2011 2012 - 2013 2014	after tax income after tax income after tax income after tax income	50% 50% 50%	16.5% 15% 13.8%	60% 60% 60%	Yes	25%7 25% 25%	bl bl bl
South Africa	2004 - 2007 2008 - 2012 2013 - 2014	after tax income after tax income after tax income	50% 50% 50%	$^{-}_{25.91\%}$	75% 75%		5% ** 5% ** 5% **	1 1
South Korea	$\begin{array}{c} 2004 - 2005\\ 2005 - 2008\\ 2009\\ 2010 - 2011\\ 2011 - 2014\\ 2012 - 2014 \end{array}$	full income full income full income full income full income	50% 20% 20%	15% 15% 15%	51% - 55% 55% 62% 62%	1 1 1 1 1	50%** 50%** 50%** 50%**	bl bl
Spain	2004 - 2006 2007 2008 - 2014	passive income passive income passive income	50% 50% 50%	26.25% 24.38% 22.5%	75% 75%	Yes Yes Yes	15% 15% 15% 115%	1 1 1
Sweden	2004 - 2007 2008 2009 - 2012 2013 - 2014	full income full income full income full income	25% 25% 25%	$15.4\% \\ 15.4\% \\ 14.47\% \\ 12.1\%$	55% 55% 55% 85% 85%	_ Yes Yes	1 1 1 1	wl wl wl wl
Turkey	2007 - 2014	full income	50%	10%	50%	I	25%	I
United Kingdom	2004 - 2006 2007 2008 - 2010 2011 2013 2013 2013 2013	full income full income full income full income full income passive income passive income	50% 50% 50% 50% 50%	22.5% 22.5% 19.5% 18% 17.25%	75 75 75 75 75 75 75 8 8 8 8 8 8 8 8 8 8	Yes Yes Yes Yes Yes	50% 50% 50% 50% 50%	
USA M_{a4a} * a_{a4a} f_{a4a}	2004 - 2014	quasi passive income	50%	36%	%06	I	5%8	I
Note: [*] or de fact Source: Informati	o control. ** A: on is gathered	n active business test is ei from the various specific	nacted. laws, the Eur	opean Tax Handbool	ss, which are published every	year by the	International Bure	au for Fiscal

Documentation (IBFD) and various other sources for cross-checks.

 $^{^{6}}$ For the full time span until 2014 Norway had an CFC exemption if the subsidiary is in a tax treaty country and has mainly active income. ⁷ Portugal exempts foreign income – for the full time frame until 2014 – from CFC legislation if these profits are from local activities and mainly not

financial.

⁸ Note that the USA had the 'check the box' rule enacted which allowed easy circumvention of their CFC rules (see e.g., Grubert and Altshuler (2006).

Appendix B

CFC rules and **Profit** Shifting

B.1 Mathematical Proofs

Proof of Equation 10:

Including expressions 3.4, 3.5 and the taxes for both, parent and subsidiary, into the overall profits results in:

$$\pi = [f_{k_A} - c_{A(k_A)}] (1 - \Phi) + [f_{k_B} - c_{B(k_B)}] + \Phi (f_{k_A} - c_{A(k_A)}) - k_A (1 - \Phi) t_A - [k_B t_B + \Phi k_A t_B] - C_{\Phi k_A}.$$
(B.1)

Using $y_x = f_{k_x} - c_x$ for $x \in [A, B]$ and deriving with respect to k_A gives:

$$\frac{\delta\pi}{\delta k_A} = y_A' + y_B' - t_A + \Phi t_A - \Phi t_B - C'_{\Phi k_A} \stackrel{!}{=} 0.$$
 (B.2)

Including the marginal tax rate τ from expression 3.9 yields:

$$y_A' + y_B' = \tau + C_{\Phi} ,$$
 (B.3)

which at the end results in equation 3.10.

B.2 Further bunching methodology elaboration

As shown in the above mentioned bunching literature, the elasticity of taxable corporate income could be derived by

$$\epsilon = \frac{\Delta \pi^* / \pi^*}{\Delta \tau / (1 - \tau)} , \qquad (B.4)$$

or, due to a rather large notch,¹ as a parametric formula

$$\epsilon = -\frac{\log(1 + \Delta \pi^* / \pi^*)}{\log(1 - \Delta \tau / (1 - \tau))}, \qquad (B.5)$$

which is a generalization of Equation B.4 and both are local elasticities at the threshold level π^* only. As explained, this study does not calculate elasticities of taxable income but rather uses the observable bunching, and later the counterfactual distribution, to show and quantify the behavioral responses of multinational's profit shifting due to CFC legislation.

As we can abstract from income effects, due to the above mentioned facts, and because we have no tax rate changes on inframarginal income below the threshold, the fraction of firms who choose to locate right before the threshold can be denoted as $B(t_1, t_2) = \int_K^{K+\Delta x} g(x) dx$, where g(x) is the density distribution of taxable income when there is a constant marginal tax rate over the whole distribution. Under the assumption that g(x) is uniform around the threshold, the elasticity of corporate taxable income at the threshold is

$$e \cong \frac{B(t_1, t_2)/g(K)}{Kln(\frac{1-t_1}{1-t_2})} = \frac{b(t_1, t_2)}{Kln(\frac{1-t_1}{1-t_2})}$$
(B.6)

where $b(t_1, t_2)$ expresses the fraction of firms who bunch at the passive-to-total income threshold relative to the counterfactual density. The tax rates t_1 and t_2 and the threshold point are given policy parameters, $b(t_1, t_2)$ needs to be estimated empirically instead, so that e can be identified.

Therefore, I group companies in small passive-to-total-income ratio bins indexed by j and estimate the counterfactual distribution using the following regression:

$$c_{j} = \sum_{i=0}^{q} \beta_{i} \cdot (z_{j})^{i} + \sum_{i=z_{-}}^{z_{+}} \gamma_{i} \cdot \mathbf{1}[z_{j} = i] + \epsilon_{j} , \qquad (B.7)$$

where c_j is the number of companies in bin j, z_j is the level of shifted profit in bin j, $[z_-, z_+]$ is the excluded range around the threshold, q is the order of the flexible polynomial and γ is a bin fixed effect for each bin in the excluded range. Omitting the contribution of the dummies in the excluded range, I get the initial estimate of the counterfactual distribution (i.e. $\hat{c}_j = \sum_{i=0}^{q} \hat{\beta}_i \cdot (z_j)^i$). Excess bunching is then given by the difference between the counterfactual and observed bin counts within the excluded range (i.e. $\hat{B}^0 = \sum_{i=2}^{z_+} (c_j - \hat{c}_j^0)$).

¹ Income effects of tax changes on profits π are ruled out due to the fact that firms would choose a totally different tax avoidance strategy if this threshold opportunity is not profitable any more. This behavior would not be observed in the present setting. Possible optimization frictions are discussed below.

This oversimplified calculation could overestimate \hat{B} and, therefore, further refinements should be implemented if tax elasticity are calculated. Namely, one could add an additional set of round number dummies to control for bunching at other round income ratios $(\sum_{r \in R_k} \rho_{r_k} \cdot \mathbf{1}[z_j = i])$. Even though there are no obvious reasons why a company should aim for a specific round number *ratio* it is known from other literature about individuals that round numbers seem to attract optimization attention. Additionally, following Chetty et al. (2011), one could shift the counterfactual distribution to the right of the threshold until it satisfies the constraint that the number of companies in the observed distribution is equal to the one of the counterfactual distribution. To analyze and show the specific firm behavior I observe passive income subsidiary effects of time and destination differences.

B.3 Descriptives and more robustness checks

Country	Total	(ln) Fixed	(ln) Financial Profits	Number of Parents	Number of Subsidiaries	Avg. CIT
		12221	1 101105	1 4101105		
Australia	15892	13621	4710	1415	2931	30.0
Austria	69425	64910	11722	4757	9571	25.3
Belarus	3019	2072	380	502	539	22.2
Belgium	47971	44348	12392	2517	6416	34.0
Bermuda	16565	12876	5012	305	2254	0.0
Brazil	2094	1744	417	113	357	34.0
British Virgin Isl.	33276	30177	7476	2631	5442	0.0
Bulgaria	30509	24631	2784	3606	3871	10.8
Canada	18123	15094	4785	771	2606	29.9
Cayman Islands	13043	11344	3533	582	2192	0.0
Chile	908	809	160	67	139	18.1
China	14491	12891	3195	1187	2459	27.0
Croatia	5333	5054	1115	487	682	20.0
Curacao	11851	10671	1894	601	1507	32.8
Cyprus	72477	67019	14894	6822	11511	10.6
Czech Republic	24845	22373	3685	3848	4220	20.6
Denmark	45080	42067	13193	2880	6760	25.5
Estonia	3536	3047	633	396	502	21.6
Finland	19451	18369	6179	747	2564	25.4
France	102377	91522	32800	3473	13774	33.3
Germany	386581	364254	61412	34022	52229	32.0
Greece	15242	14245	2039	1596	1960	25.3
Hong Kong	13160	12000	2746	1465	2370	16.7
Hungary	8574	7051	1699	1276	1539	17.8
Iceland	3205	2826	752	217	456	18.2
India	12622	11285	3246	482	1830	33.9
Ireland	29556	25349	7452	1263	3757	12.5
Israel	9239	8459	1986	576	1432	27.0
Italy	349350	336898	49417	36540	44610	33.1
Japan	59996	55805	22322	1340	8439	39.8
Korea, Republic	5607	5145	1491	405	888	25.0
Continued on next pag						n next page

Table B.1: Subsidiary-year observations of countries in data set - by parent countries

Country	Total	(\ln) Fixed	(ln) Financial Profits	Number of Parents	Number of Subsidiaries	Avg. CIT rate in %
.		7155005				1400 111 70
Latvia	7725	5914	705	879	947	15.0
Liechtenstein	9094	8348	1697	584	1235	12.5
Lithuania	6666	5906	872	907	1032	15.5
Luxembourg	91221	84675	21773	4294	13009	29.1
Malaysia	2961	2354	564	167	488	25.9
Malta	4682	4257	951	359	731	35.0
Mexico	2442	2054	698	64	317	29.5
Netherlands	107158	98437	27498	5231	14495	26.5
New Zealand	2754	2524	747	183	466	30.0
Norway	67596	63554	23112	5997	8865	27.9
Panama	6222	5289	1178	544	925	27.8
Poland	28029	26366	7019	4031	4525	19.0
Portugal	23208	21884	3244	2381	3117	25.5
Romania	33917	32398	7482	4281	4362	16.1
Russia	232330	168768	15934	39970	44507	21.2
Serbia	10528	10138	2388	1200	1254	10.8
Seychelles	4083	3522	892	500	712	36.2
Singapore	7452	6597	1771	502	1250	18.0
Slovakia	11510	10578	1324	1574	1707	19.8
Slovenia	8036	7539	1518	684	1024	20.7
South Africa	5690	4537	1572	136	852	33.9
Spain	158162	152278	32405	15933	19700	31.7
Sweden	49359	45681	15526	1765	6677	26.2
Switzerland	87311	78795	20086	4347	12094	19.2
Taiwan, Republic	6882	6552	2159	488	1131	20.7
Turkey	7727	7025	1356	965	1169	21.0
Ukraine	4539	3805	568	566	716	23.7
Unit. Arab. Emi.	5712	5024	1573	311	866	0.0
United Kingdom	124239	109591	34302	6098	17575	27.1
USA	273178	240715	95386	7023	37896	39.2
Total	2833811	2563061	611821	228853	403451	2833811

Table B.2: Subsidiary-year observations of countries in data set - by parent countries

Note: The observations numbers of columns four to six are for the (ln) financial profit observations, as they are used mostly and, therefore, give a better impression.

	. '	lable B.3: I	Jescriptive 2	tatistics		
	Obs	Mean	Std. Dev.	Min	Max	Data source
Financial Profits	2,298,714	1487.695	141,539.3	-4.28e+07	7.31e+07	ORBIS
Tangible Fixed Assets	2,676,250	11067.23	352,577.7	-180,941.7	4.53e+08	ORBIS
Intangible Fixed Assets	2,655,578	3358.205	1,662,297	-1,490,282	2.70e+09	ORBIS
Fixed Assets	2,780,597	56858.57	1958454	-793,784.6	2.70e+09	ORBIS
Total Assets	2,807,825	186,250.1	7,870,930	-634,888	$3.20e{+}09$	ORBIS
$\operatorname{Employees}$	1,752,595	146.4313	1,369.771	0	478,980	ORBIS
Operational Profits	2,282,224	-122.7246	3,505,755	-5.29e+09	4.02e + 07	ORBIS
CIT rate sub country	2,830,149	.2692973	.0669161	0	.55	KPMG & Tax Guides
Corruption Index	2,828,578	20.94826	26.01556	1.3	92	Transparency International
GDP sub Country	2,831,831	1.93e+12	1.78e + 12	4.77e+08	1.72e+13	World Bank
GDP per capita sub Country	2,831,831	33,706.48	11,971.33	520.9566	136, 135.5	World Bank
Unemployment rate sub Country	2,832,390	8.325034	4.136631	1.	37.6	World Bank
CIT rate parent country	2,833,811	.2782615	.086963	0	.4069	KPMG & Tax Guides
Note: 'sub' stands for subsidiary.	'CIT' stand	s for statute	ory corporat	e income tax	. All compa	ny specific financial variables

••••• đ • . Ĺ C ٢ T, L, L are measured in thousands of USD.

Dependent Variable	Log Financial Profits abroad			
	(1)	(2)	(3)	(4)
	treated	effective CIT	dif. tax base	split of non treated
Cfa mila	-0.164***	-0.181***		
sub in low tax country (CFC)	(0.0162)	(0.0154)		
CEC + All income has			-0.152^{***}	
CFC + All income base			(0.0173)	
CEC + Potycon all			-0.174**	
and Passive income base			(0.0769)	
CFC + Passive income base			-0.219***	
			(0.0390)	
CFC (Below threshold)				-0.203***
				(0.0234)
Above threshold				0.0890***
				(0.0260)
Higher than own CIT				0.0282
				(0.0328)
In Tangible Fixed Assets	0.121^{***}	0.121^{***}	0.121^{***}	0.120***
	(0.00378)	(0.00378)	(0.00378)	(0.00402)
CIT sub	0.346**	0.319^{*}	0.304^{*}	0.186
	(0.171)	(0.171)	(0.173)	(0.187)
In corruption sub	0.556***	0.568***	0.556***	0.675***
	(0.0524)	(0.0524)	(0.0524)	(0.0550)
ln gdp sub	2.246***	2.175^{***}	2.245^{***}	2.258***
	(0.219)	(0.218)	(0.219)	(0.228)
ln gdppc sub	-1.393***	-1.342***	-1.392***	-1.286***
	(0.224)	(0.224)	(0.224)	(0.233)
ln unempl sub	-0.0548***	-0.0485**	-0.0542***	-0.0509**
	(0.0189)	(0.0189)	(0.0189)	(0.0198)
CIT GUO	0.281	0.292	0.275	1.265***
	(0.186)	(0.186)	(0.187)	(0.216)
In corruption GUO	0.327***	0.330***	0.314***	0.290***
	(0.0618)	(0.0614)	(0.0624)	(0.0646)
ln gdp GUO	0.303*	0.330**	0.280*	0.421**
	(0.163)	(0.163)	(0.163)	(0.170)
ln gdppc GUO	-0.554***	-0.571***	-0.541***	-0.705***
	(0.175)	(0.175)	(0.175)	(0.181)
ln unempl GUO	0.221***	0.217***	0.217***	0.192***
	(0.0195)	(0.0195)	(0.0195)	(0.0206)
Observations	350,144	350,144	350,144	320,159
Adjusted R-squared	0.783	0.783	0.783	0.785
# of CFC sub	$50,\!525$	$50,\!525$	$50,\!525$	$23,\!587$
Year FE & Parent FE & Subsidiary FE	YES	YES	YES	YES
~ Substatuty I L				

Table B.4: Influence of enacted CFC rules and their characteristics on profit shifting Dependent Variable

Clustered standard errors on parent firm level in parentheses *** p<0.01, ** p<0.05, * p<0.1

Note: 'CIT' stands for statutory corporate income tax rate, 'ln gdppc' for the log of GDP per capita, 'ln unempl' for the log of the unemployment rate. 'ln corruption' is used from the Heritage Foundation calculation. 'sub' stands for subsidiary and 'GUO' for global ultimate parent. In specification (2) the effective average CIT's are used to determine whether a subsidiary is located in a foreign low-tax country (by the CFC law of the parent country) or not. In specification (3) the probably treated subsidiaries are differentiated between the concerned tax base of their parents CFC rules. In the last regression (4) only CFC rules with thresholds are observed and the non treated subsidiaries are split into two groups, higher or lower than the CIT rate of the parent. All regressions are panel fixed effects estimations and winsorized at the observation levels of 1% and 99% of Financial Profits.

B.4 Further bunching insights



Figure B.1: Bunching at passive income threshold of 50% ratio

Notes: The figure shows an aggregated picture of foreign subsidiaries with financial profits above 20,000 USD from parent home countries where a CFC rule is enacted. These subsidiaries bunch right before the 50% passive-to-total income ratio, which is often the threshold to get affected as a CFC and get taxed by the observed CFC laws. The observed span reaches from a ratio of 10% to 90% financial-to-total profits before taxes.



Table B.5: Countries with passive to total income threshold at 50% in CFC rule or a comparable law.

Note: The table shows foreign subsidiaries from various parent countries that have a passive-to-total income threshold at 50% or a comparable legislation included in their CFC laws. HongKong does not have a CFC rule but a comparable law to classify entities as non-financial entities. Estonian CFC rules affect only Estonian individuals. Only subsidiaries with positive financial income are used to generate the graphs.



Table B.6: Countries *without* a passive to total income threshold at 50% in CFC rule or a comparable law.

Continued on next page
Continuation of Table B.6



Continued on next page

Continuation of Table B.6



* These countries do have a CFC rule in the observed time frame.

Note: The table shows foreign subsidiaries from various parent countries that do not have a passive-to-total income threshold at 50% or a comparable legislation included in their CFC laws or no CFC regulation at all. Only subsidiaries with positive financial income are used to generate the graphs.

Variable	Obs	Mean	Min	Max
Fixed Assets	2'193	161'068.7	0	42'100'000
Intangible Fixed Assets	2'100	5'244.59	-8'670.99	1'926'008
Employees	1'611	281.60	0	20'479
Sales	1'662	146'775.50	0	19'100'000
Financial Profits	2'202	9'878.13	20.36	3'432'162
Profits before Taxes	2'202	2'0471.31	41.06	7'026'615
Tax payed	2'210	3'521.67	-182'225	1'681'732
R & D expenses	106	23'127.61	0	2'106'390

Table B.7: Descriptive statistics about bunching subsidiaries abroad from Figure 3.3.

Note: This table provides some statistics about the subsidiaries which bunch at the 50% income ratio threshold in figure 3.3. Financial numbers are in thousands.

Appendix C CFC rules and M&A

C.1 Stylized identification variable example

Base case in $t=0$		1	
STR Parent	Parent Country 1 30%	Parent Country 2 $26,25\%$	Parent Country 3 20%
CFC rule with min. tax threshold at	$<\!25~\%$	<90% of own STR (i.e., 23.63%)	none
Subsidiary Country A (STR = $20,5\%$)	Х	X	
Subsidiary Country B (STR = 19%)	Х	Х	
Subsidiary Country C (STR = 12%)	Х	Х	
Change in CFC law threshold in	t=1		
	Parent Country 1	Parent Country 2	Parent Country 3
STR Parent	30%	26,25%	20%
CFC rule with min. tax threshold at	$<\!20~\%$	<80% of own STR (i.e., 21%)	none
Subsidiary Country A (STR = 20.5%)		(1.0., 2 170) X	
Subsidiary Country B (STR = 19%)	Х	Х	
Subsidiary Country C (STR = 12%)	Х	Х	
Change in parent country STR i	n t $=2$		
	Parent Country 1	Parent Country 2	Parent Country 3
STR Parent	$\mathbf{25\%}$	21,25%	16%
CFC rule with min. tax threshold at	$<\!20~\%$	$<\!\!80\% ext{ of own STR} $ (i.e., 17%)	none
Subsidiary Country A (STR = $20,5\%$)			
Subsidiary Country B (STR = 19%)	Х		
Subsidiary Country C (STR = 12%)	Х	Х	
Change in subsidiary country ST	TR in $t=3$		
	Parent Country 1	Parent Country 2	Parent Country 3
STR Parent	25%	$21,\!25\%$	16%
CFC rule with min. tax threshold at	$<\!\!20~\%$	$<\!80\% \text{ of own STR} $ (i.e., 17%)	none
Subsidiary Country A (STR = 18%)	Х		
Subsidiary Country B ($\mathbf{STR} = \mathbf{16\%}$)	Х	Х	
Subsidiary Country C ($\mathbf{STR} = \mathbf{10\%}$)	Х	Х	

Table C.1: Identification Variable Example.

Note: An "X" indicates that this subsidiary country is potentially affected by CFC rules. These depicted changes in the different triggering law settings occur over time in various countries so that various subsidiaries are potentially affected by CFC legislation and others are not. In our regressions, we use various fixed effects and other control variables to account for other potentially influencing effects.

C.2 Descriptives and robustness check tables on lowtax target acquisition

Country	CFC	Number of	Number of	Country	CFC	Number of	Number of
Country	rule	acquirers	targets		rule	acquirers	targets
Australia	1	923	663	Japan	1	529	166
Austria	0	125	73	Korea, Rep.	1	187	147
Belarus	n/a	none	6	Latvia	n/a	none	2
Belgium	0	154	186	Lithuania	n/a	none	14
Bermuda	n/a	none	29	Malaysia	0	212	157
Brazil	n/a	none	251	Malta	n/a	none	4
British Virgin Islands	n/a	none	70	Mexico	n/a	none	197
Bulgaria	n/a	none	30	Netherlands	0	421	355
Canada	1	1,124	1,074	New Zealand	1	68	196
Cayman Islands	n/a	none	17	Norway	1	296	144
Chile	n/a	none	95	Panama	n/a	none	10
China	1	338	846	Poland	n/a	none	140
Croatia	n/a	none	20	Portugal	n/a	none	69
Cyprus	n/a	none	16	Russian Federation	0	39	112
Czech Republic	n/a	none	81	Seychelles	n/a	none	2
Denmark	1	42	158	Singapore	0	490	271
Estonia	n/a	none	12	Slovak Republic	n/a	none	16
Finland	1	62	142	Slovenia	n/a	none	15
France	1	644	667	South Africa	n/a	none	119
Germany	1	622	842	Spain	1	324	360
Greece	n/a	none	25	Sweden	1	71	369
Hong Kong SAR, China	0	560	343	Switzerland	0	344	209
Hungary	n/a	none	45	Taiwan, China	n/a	none	105
Iceland	n/a	none	3	Turkey	n/a	none	79
India	0	337	214	Ukraine	n/a	none	31
Ireland	0	342	152	United Kingdom	1	$1,\!670$	1,772
Israel	1	206	129	United States	1	4,020	2,857
Italy	1	271	314	Total		$14,\!421$	$14,\!421$

Table C.2: Cross-border M&A sample (2002-2014) for analyzing the effect of acquirer CFC rules on the probability of being the **acquirer country** (Section 4.3.4.2).

Note: This table shows the number of acquirer ultimate parents and targets per country in our cross-border M&A sample to investigate Hypothesis 1a. In this context, cross-border M&As are defined as acquirer ultimate parent and target residing in different countries; the direct acquirer and acquirer ultimate parent reside in the same country. To keep the mixed logit regressions computationally feasible, the set of considered candidate acquirer countries is restricted (see section 4.3.3). In this table CFC rule takes the value one, if the acquirer country has implemented CFC rules in 2014 and zero otherwise.

Table C.3: Definition	, data sources and summary statistics of variables for analyzing effect of acq	uirer CFC rules on prob	ability of b	seing acqu	irer country	(section 4.	3.4.2).
Variable	Definition	Data source	Obs.	Mean	Std. Dev.	Min	Max
CFC^{dummy}	Binary dummy variable coded one if target country STR is smaller than acquirer country's tax rate threshold of CFC rule or acquirer country applies CFC rules without a tax rate threshold, and 0 otherwise	TG	317,835	0.111	0.315	0	1
$CFC^{di}ff$	Difference between acquirer country STR and target country STR if target country STR is smaller than acquirer country's tax rate threshold of CFC rule or acquirer country applies CFC rules without a tax rate threshold, and 0 otherwise	TG	317,835	0.012	0.043	0	0.409
$CFC^{di}ffEATR$	Difference between acquirer country STR and target country STR if target country EATR is smaller than acquirer country's tax rate threshold of the CFC rule or acquirer country applies CFC rules without a tax rate threshold, and 0 otherwise	TG; CBT	317,835	0.011	0.039	-0.011	0.409
$CFC^{diffEEA}$	Same as CFC^{diff} ; however, set to zero if acquirer and target country are both EEA member states and M&A year is after 2006	TG	317,835	0.012	0.042	0	0.409
CFCtaxbase CFCprofitable CFCnon-profitable STR	See expression 4.7 Same as CFC^{diff} ; however, for non-profitable targets set to zero Same as CFC^{diff} ; however, for profitable targets set to zero STR in candidate acquirer country, including typical local taxes	TG TG; SDC; CNA; CG TG; SDC; CNA; CG TG	317,835 55,715 55,715 317,835	$\begin{array}{c} 0.318 \\ 0.007 \\ 0.003 \\ 0.291 \end{array}$	$\begin{array}{c} 0.066\\ 0.034\\ 0.021\\ 0.071 \end{array}$	$\begin{array}{c} 0 \\ 0 \\ 0 \\ 0.125 \end{array}$	$\begin{array}{c} 0.409\\ 0.395\\ 0.409\\ 0.409\end{array}$
ExemptionMethod	Binary dummy variable coded one if candidate acquirer country unilaterally applies the exemption method to avoid double taxation of foreign dividends, and 0 if it unilaterally applies the credit method	TG	294,697	0.606	0.489	0	-
GDP per capita GDP growth	GDP per capita in candidate acquirer country (natural logarithm) Growth of GDP in candidate acquirer country (in $\%)$	World Bank World Bank	317,835 317,835	10.416 3.095	0.62 3.168	7.942 -7.821	11.284 15.240
Stock market capital- ization per GDP	Stock market capitalization of listed domestic firms in candidate acquirer country (in $\%$ of GDP)	World Bank	317,835	121.5	175.6	15.767	1,254.5
Size of private credit market	Domestic credit to private sector in candidate acquirer country (in $\%$ of GDP)	World Bank	317,835	115.3	39.525	31.081	233.4
Distance	Simple distance (in km) between most populated cities of candidate acquirer and target country (natural logarithm)	Mayer and Zignago (2011)	317,835	8.498	1.100	4.088	9.883
Common language	Common language index between candidate acquirer and target country (0 (low) to 1 (high similarity))	Melitz and Toubal (2014)	317,835	0.242	0.217	0	0.983
Past colonial relationships	Binary dummy variable coded one if candidate acquirer and target country were ever in a colonial relationship, and 0 otherwise	Mayer and Zignago (2011)	317,835	0.095	0.294	0	1
Common legal system	Binary dummy variable coded one if legal system of candidate acquirer and target country have common legal origins, and 0 otherwise	Head et al. (2010)	317,835	0.319	0.466	0	1
TargetAssets	Pre-deal consolidated target total assets in the last year before the effective $M\&A$ date (natural logarithm)	SDC; CNA; CG	52,809	18.118	2.297	11.513	28.060
TargetROA	Pre-deal consolidated target pre-tax income in the last year before the effective $M\&A$ date divided by pre-deal consolidated target total assets in the last year before the effective $M\&A$ date	SDC; CNA; CG	52,809	-0.036	0.844	-11.800	18.000
TargetSales	Pre-deal consolidated target net sales in the last year before the effective $M\&A$ date (natural logarithm)	SDC; CNA; CG	78,495	17.667	2.320	6.908	26.216
					Cont	inued on n	ext page

	Max	24.300	siness
	Min	7.601	e for Bu
	Std. Dev.	2.093	iversity Centre
	Mean	16.369	xford Uni
	Obs.	34,405	stands for Oy oal,
previous page	Data source	SDC; CNA; CG	nds for Tax Guides, 'CBT' stands for Compustat Glob
TableC.3 - continued from	ariable Definition	Pre-deal consolidated target EBITDA (earnings before interest, argetEBITDA taxes, depreciation and amortization) in the last year before the effective M&A date (natural logarithm)	lote: Data on country fixed effects are not reported but are available upon request. 'TG' stan 'axation, 'SDC' stands for SDC Platinum, 'CNA' stands for Compustat North America, 'CG'

Table C.4: Supplemental regression results for candidate acquirer country fixed effects interacted with target-specific financial data.

Regression (3) of table	4.3	Regression (4) of table	4.3	Regression (5) of table 4	.3
Australia * TargetAssets	-0.1275**	Australia*TargetSales	-0.1167***	Australia*TargetEBITDA	-0.1229*
	(0.0526)		(0.0417)		(0.0696)
$Austria^{*}TargetAssets$	0.0927	Austria*TargetSales	0.0242	Austria*TargetEBITDA	0.2592^{**}
	(0.0960)		(0.0851)		(0.1150)
Belgium*TargetAssets	0.0394	Belgium*TargetSales	-0.0256	Belgium*TargetEBITDA	0.0561
	(0.0890)		(0.0693)		(0.1021)
Canada * TargetAssets	-0.1606***	Canada*TargetSales	-0.1735***	Canada*TargetEBITDA	-0.1486**
	(0.0541)		(0.0380)		(0.0643)
China ⁺ 1 argetAssets	0.0502	China ⁺ LargetSales	-0.0781	China* TargetEBITDA	-0.0301
Donmanh*Tanact A costs	(0.0579)	Donmank*TanactSaloo	(0.0307)	Dommark*TargetEPITDA	(0.1096)
Denmark TurgetAssets	(0.1467)	Denmark TargetSules	(0.1215)	Denmark TargetEBITDA	(0.1813)
Finland*TargetAssets	0.0130	Finland*TaractSales	0.1080***	Finland * Target FRITD 4	0.0561
Finiana TargetAssets	(0.1863)	1 mana Targeisales	(0.0728)	Finiana TargetEBTTEA	(0.0490)
France * Target Assets	0 1841***	France*TargetSales	0.1561***	France*TaraetEBITDA	0 1999***
Transo Targoniosolo	(0.0477)	Transes Targeboates	(0.0420)	Transs TargetEBTTETT	(0.0603)
$Germanu^*TargetAssets$	0.1779***	Germanu*TargetSales	0.1239***	$Germany^*TargetEBITDA$	0.2245***
	(0.0482)		(0.0479)		(0.0636)
$HongKongSARChina^*$	-0.0375	HongKongSARChina*	-0.0809*	HongKongSARChina*	-0.0597
TargetAssets	(0.0544)	TargetSales	(0.0477)	TargetEBITDA	(0.0725)
India * TargetAssets	-0.1437**	India * TargetSales	-0.0593	India * Target EBIT DA	-0.3182***
	(0.0591)		(0.0369)		(0.0755)
Ireland * TargetAssets	-0.1022**	Ireland * TargetSales	-0.0565	Ireland*TargetEBITDA	-0.1737**
	(0.0504)		(0.0410)		(0.0714)
Israel*TargetAssets	-0.0013	$Israel^{*}TargetSales$	-0.0859	$Israel^*TargetEBITDA$	0.0781
	(0.0810)		(0.0572)		(0.1288)
Italy * TargetAssets	0.0162	Italy * TargetSales	0.0067	Italy*TargetEBITDA	0.0309
	(0.0585)		(0.0457)		(0.0794)
Japan*TargetAssets	0.1112^{**}	Japan*TargetSales	0.1007^{**}	Japan*TargetEBITDA	0.0818
	(0.0461)		(0.0404)		(0.0696)
KoreaRep*TargetAssets	0.0875	KoreaRep*TargetSales	-0.0338	KoreaRep*TargetEBITDA	0.2206
	(0.1026)		(0.0893)		(0.2751)
Malaysia*TargetAssets	-0.1075	Malaysia*TargetSales	-0.1171*	Malaysia*TargetEBITDA	-0.2086
	(0.1090)		(0.0707)		(0.1310)
Netherlands*TargetAssets	0.1765***	Netherlands*TargetSales	0.0893*	Netherlands*TargetEBITDA	0.1696**
	(0.0504)		(0.0458)		(0.0699)
New Zeal and * Target Assets	-0.0111	New Zeal and * Target Sales	0.2038**	New Zeal and * Target EBIT DA	-0.1343
N. YT IA I	(0.1395)		(0.0951)		(0.1243)
Norway [*] TargetAssets	-0.2134***	Norway [*] TargetSales	-0.1773***	Norway [*] TargetEBITDA	-0.2307**
	(0.0732)		(0.0423)		(0.1167)
RussianFederation " 1 argetAssets	0.0481	RussianFederation * 1 argetSales	-0.1325	RussianFederation * TargetEBITDA	0.2715
Singanone*Tanact Acasta	(0.2429)	Sin agnone * Tanget Sales	(0.1597)	Sin agnono * Tanget FPITD A	(0.1787)
Singupore TurgetAssets	-0.0009	Singupore TurgerSures	-0.0877	Singapore TargetEDITDA	-0.0784
Spain * Target Assets	0.2229***	Spain * Taraet Sales	0.1261**	Spain * Target EBITDA	0.1338
Spann Targen13503	(0.0759)	Spann TargetSates	(0.0589)	Spann TangetEBTTER	(0.0972)
Sweden * Taraet Assets	0.3177***	Sweden *TaraetSales	-0.0665	Sweden*TargetEBITDA	0.0543
Dweach Turgen13503	(0.1215)	Sweach TargetSuies	(0.0901)	bucach TargetEBTTEN	(0.1561)
Switzerland * TargetAssets	0.1798***	Switzerland * TaraetSales	0.0347	Switzerland*TargetEBITDA	0.1748**
	(0.0563)		(0.0557)		(0.0872)
UnitedKingdom*TargetAssets	-0.0638	UnitedKingdom*TargetSales	-0.1709***	$UnitedKingdom^*TargetEBITDA$	-0.0150
0 0	(0.0475)	0 0	(0.0314)	5 5	(0.0577)
Australia*TargetROA	0.0451		· · · ·		()
-	(0.1562)				
Austria*TargetROA	-0.3821**				
	(0.1873)				
Belgium*TargetROA	0.0782				
	(0.3381)				
$Canada^{*}TargetROA$	0.0885				
	(0.2366)				
China*TargetROA	-0.3323**				
	(0.1653)				
Denmark*TargetROA	0.3034				
	(0.2514)				
Finland*TargetROA	0.4007^{**}				
	(0.1818)				
France*TargetROA	0.1699				
	(0.1596)				
$Germany^*TargetROA$	-0.3493**				

Continued on next page

(0.1597)

		TableC.4 – continued from previous page	
Regression (3) of table	4.3	Regression (4) of table 4.3	Regression (5) of table 4.3
HongKongSARChina*TargetROA	0.0771		
	(0.1329)		
India * Target ROA	0.0564		
	(0.1776)		
Ireland*TargetROA	0.2417^{*}		
	(0.1374)		
Israel*TargetROA	-0.3429**		
	(0.1377)		
Italy * Target ROA	-0.1279		
	(0.1952)		
Japan*TargetROA	0.4780^{***}		
	(0.1482)		
KoreaRep*TargetROA	-0.3778**		
	(0.1693)		
Malaysia*TargetROA	0.1243		
	(0.1701)		
Netherlands*TargetROA	0.3409		
	(0.2256)		
NewZealand*TargetROA	0.3107^{**}		
	(0.1298)		
$Norway^*TargetROA$	-0.0062		
	(0.1873)		
Russian Federation * Target ROA	0.1880		
	(0.3663)		
Singapore*TargetROA	-0.2435*		
	(0.1407)		
Spain*TargetROA	0.1719		
	(0.2793)		
Sweden*TargetROA	7.1903**		
	(3.2794)		
Switzerland * TargetROA	-0.2943*		
	(0.1715)		
$United Kingdom^*Target ROA$	0.2905^{**}		
	(0.1420)		

Note: Table reports supplemental results of regressions (3), (4) and (5) of table 4.3. In particular, the coefficient of the interaction between candidate acquirer country fixed effects with target-specific consolidated financial data (target total assets, target return on assets, target net sales and target earnings before interest, taxes, depreciation and amortization) are shown. In all regressions, the US represent the base category *, **, and *** denote statistical significance at 10%, 5%, and 1% levels, respectively. Robust standard errors are provided in parentheses.

Table C.5: Cross-border M&A sample (2002-2014) for analyzing effect of acquirer CFC rules on probability of being *target* country (Section 4.3.4.3).

Country	CFC	Number of	Number of	Country	CFC	Number of	Number of
Country	rule	acquirers	targets		rule	acquirers	targets
Australia	1	712	801	Japan	1	431	170
Austria	0	77	none	Korea, Rep.	1	162	153
Belarus	0	1	none	Lithuania	1	5	none
Belgium	0	123	197	Malaysia	0	178	174
Bermuda	0	56	none	Malta	0	5	none
Brazil	1	40	320	Mexico	1	54	270
British Virgin Islands	0	28	none	Netherlands	0	296	404
Bulgaria	0	1	none	New Zealand	1	92	141
Canada	1	1,824	594	Norway	1	130	260
Cayman Islands	0	17	none	Panama	0	5	none
Chile	0	19	none	Poland	0	25	170
China	1	271	897	Portugal	1	35	none
Croatia	0	1	none	Russian Federation	0	51	82
Cyprus	0	35	none	Seychelles	0	7	none
Czech Republic	0	7	none	Singapore	0	416	290
Denmark	1	118	35	Slovak Republic	0	2	none
Estonia	0	1	none	Slovenia	0	5	none
Finland	1	112	44	South Africa	1	58	156
France	1	490	708	Spain	1	239	369
Germany	1	433	951	Sweden	1	365	none
Greece	1	17	none	Switzerland	0	268	240
Hong Kong SAR, China	0	487	377	Taiwan, China	0	90	none
Hungary	1	7	none	Turkey	1	17	none
Iceland	1	38	none	Ukraine	0	8	none
India	0	295	227	United Kingdom	1	2,023	1,084
Ireland	0	253	181	United States	1	$2,\!647$	3,818
Israel	1	172	none				
Italy	1	198	334	Total		$13,\!447$	13,447

Note: This table shows the number of acquirer ultimate parents and targets per country in our cross-border M&A sample to investigate Hypothesis 1b. In this context, cross-border M&As are defined as acquirer ultimate parent and target residing in different countries; the direct acquirer and acquirer ultimate parent reside in the same country. To keep the mixed logit regressions computationally feasible, the set of considered candidate target countries is restricted (see section 4.3.3). CFC rule takes the value one, if the acquirer country has implemented CFC rules in 2014.

Variable	Definition	Data source	Obs.	Mean	Std. Dev.	Min	Max
CFC^{dummy}	Binary dummy variable coded one if target country STR is smaller than acquirer country's tax rate threshold of CFC rule or acquirer country applies CFC rules without a tax rate threshold, and 0 otherwise	Tax guides	317,444	0.345	0.475	0	1
CFC^{diff}	Difference between acquirer country STR and target country STR if target country STR is smaller than acquirer country's tax rate threshold of CFC rule or acquirer country applies CFC rules without a tax rate threshold, and 0 otherwise	Tax guides	317,444	0.037	0.063	0.000	0.284
$CFC^{dif}fEATR$	Difference between acquirer country STR and target country STR if target country EATR is smaller than acquirer country's tax rate threshold of the CFC rule or acquirer country applies CFC rules without a tax rate threshold, and 0 otherwise	Tax guides; Oxford University CBT	317,444	0.031	0.057	-0.033	0.284
$CFC^{diffEEA}$	Same as CFC^{diff} ; however, set to zero if acquirer and target country are both EEA member states and M&A year is after 2006	Tax guides	317,444	0.035	0.062	0.000	0.284
$CFC^{taxbase}$	See expression 4.7.	Tax guides	317,444	0.305	0.058	0.125	0.409
CFC profitable	Same as CFC^{diff} ; however, for non-profitable targets set to zero	Tax guides; SDC Platinum; Compustat North America; Compustat Global	53,270	0.026	0.057	0.000	0.284
CFCnon_profitable	Same as CFC^{diff} ; however, for profitable targets set to zero	Tax guides; SDC Platinum; Compustat North America; Compustat Global	53,270	0.013	0.042	0.000	0.277
STR GDP per capita GDP growth	STR in candidate target country, including typical local taxes GDP per capita in candidate target country (natural logarithm) Growth of GDP in candidate target country (in $\%$)	Tax guides World Bank World Bank	317,444 317,444 317,444 317,444	$\begin{array}{c} 0.287 \\ 10.267 \\ 3.221 \end{array}$	$\begin{array}{c} 0.071 \\ 0.687 \\ 3.206 \end{array}$	0.125 7.942 -7.821	$\begin{array}{c} 0.409 \\ 11.284 \\ 15.240 \end{array}$
Stock market capitalization per GDP	Stock market capitalization of listed domestic firms in candidate target country (in $\%$ of GDP)	World Bank	317,444	124.1	178.4	17.020	1,254.5
Size of private credit market	Domestic credit to private sector in candidate target country (in $\%$ of GDP)	World Bank	317,444	109.5	47.091	13.353	233.4
Distance	Simple distance (in km) between most populated cities of acquirer and candidate target country (natural logarithm)	Mayer and Zignago (2011)	317,444	8.609	1.046	5.153	9.883
Common language	Common language index between acquirer and candidate target country (0 (low similarity) to 1 (high similarity))	Melitz and Toubal (2014)	317,444	0.235	0.212	0.000	0.991
Past colonial relationships	Binary dummy variable coded one if acquirer and candidate target country were ever in a colonial relationship, and 0 otherwise	Mayer and Zignago (2011)	317,444	0.103	0.304	0	1
Common legal system	Binary dummy variable coded one if legal system of acquirer and candidate target country have common legal origins, and 0 otherwise	Head et al. (2010)	317,444	0.329	0.470	0	1
Corruption control	Corruption control index of candidate target country (-3 (low control) to 3 (high control))	World Bank	317,444	1.072	0.976	-1.088	2.527
					Conti	nued on n	ext page

Variable	TableC.6 - continued from previo	us page Data source	Obs.	Mean	Std. Dev.	Min	Max
Business start-up cost	Cost of business start-up procedures in candidate target country (in $\%$ of GNI per capita)	World Bank	317,444	9.601	12.746	0.000	78.400
Unemployment rate	Unemployment rate in candidate target country (in $\%$ of total labor force)	World Bank	317,444	7.031	5.050	2.493	27.140
Domestic firms	Number of listed domestic firms in candidate target country (natural logarithm)	World Bank	317,444	6.426	1.232	3.714	8.638
$Business \ disclosure$	Business extent of disclosure index of in candidate target country (0 (less disclosure) to 10 (more disclosure))	World Bank	264, 159	7.188	2.344	0	10
AcquirerAssets	Pre-deal consolidated acquirer total assets in the last year before the effective $M\&A$ date (natural logarithm)	SDC Platinum; Compustat North America; Compustat Global	215,197	20.280	2.808	11.513	28.710
AcquirerROA	Pre-deal consolidated acquirer pre-tax income in the last year before the effective $M\&A$ date divided by pre-deal consolidated acquirer total assets in the last year before the effective $M\&A$ date	SDC Platinum; Compustat North America; Compustat Global	215,197	0.035	5.999	-191.9	360.5
A cquirer Sales	Pre-deal consolidated acquirer net sales in the last year before the effective $M\&A$ date (natural logarithm)	SDC Platinum; Compustat North America; Compustat Global	206,176	19.979	2.732	8.219	26.834
AcquirerEBITDA	Pre-deal consolidated acquirer EBITDA (earnings before interest, taxes, depreciation and amortization) in the last year before the effective M&A date (natural logarithm)	SDC Platinum; Compustat North America; Compustat Global	180, 202	18.594	2.365	9.210	24.723
Note: Data on country f	ixed effects are not reported but are available on request.						

Table C.7: Supplemental regression results for candidate target country fixed effects interacted with acquirer-specific financial data.

Regression (3) of table 4.	6	Regression (4) of table 4.	.6	Regression (5) of table 4.0	6
$Australia^*AcquirerAssets$	-0.0867***	Australia*AcquirerSales	-0.0542***	$Australia^*AcquirerEBITDA$	-0.0819***
	(0.0194)		(0.0199)		(0.0230)
$Belgium^*AcquirerAssets$	-0.0737**	Belgium*AcquirerSales	-0.0633**	$Belgium^*AcquirerEBITDA$	-0.1133^{***}
	(0.0302)		(0.0301)		(0.0389)
$Brazil^*AcquirerAssets$	0.0321	$Brazil^*AcquirerSales$	0.1174^{***}	$Brazil^*AcquirerEBITDA$	0.1288^{***}
	(0.0301)		(0.0373)		(0.0361)
$Canada^*\!AcquirerAssets$	-0.1900***	Canada * AcquirerSales	-0.1707^{***}	$Canada^*A cquirer EBITDA$	-0.1391***
	(0.0244)		(0.0245)		(0.0298)
China*AcquirerAssets	-0.1894***	China*AcquirerSales	-0.1697^{***}	China*AcquirerEBITDA	-0.1784^{***}
	(0.0212)		(0.0201)		(0.0279)
Denmark*AcquirerAssets	-0.0393	Denmark*AcquirerSales	-0.0148	Denmark*AcquirerEBITDA	-0.0599
	(0.0754)		(0.0855)		(0.0968)
Finland * Acquirer Assets	-0.0406	Finland * AcquirerSales	0.0486	Finland * A cquirer EBITDA	-0.1441
	(0.0725)		(0.0660)		(0.1023)
France*AcquirerAssets	-0.0699***	France*AcquirerSales	-0.0638***	France*AcquirerEBITDA	-0.1531^{***}
	(0.0168)		(0.0174)		(0.0216)
$Germany^*A cquirer Assets$	-0.0929***	$Germany^*AcquirerSales$	-0.0944***	Germany*AcquirerEBITDA	-0.1481***
	(0.0156)		(0.0160)		(0.0196)
HongKongSARChina*AcquirerAssets	-0.2496***	HongKongSARChina*AcquirerSales	-0.2166***	HongKongSARChina*AcquirerEBITDA	-0.2576***
	(0.0345)		(0.0277)		(0.0399)
India*AcquirerAssets	0.0178	$India^*A cquirerSales$	0.0684*	India*AcquirerEBITDA	0.0444
	(0.0334)		(0.0371)		(0.0420)
Ireland * Acquirer Assets	-0.0215	Ireland * AcquirerSales	-0.0067	Ireland * Acquirer EBITDA	-0.0489
	(0.0349)		(0.0344)		(0.0421)
Italy * Acquirer Assets	0.0233	Italy * AcquirerSales	0.0241	Italy*AcquirerEBITDA	-0.0159
	(0.0291)		(0.0300)		(0.0359)
Japan*AcquirerAssets	0.0125	Japan*AcquirerSales	-0.0390	Japan*AcquirerEBITDA	-0.0462
	(0.0403)		(0.0456)		(0.0554)
KoreaRep * AcquirerAssets	0.0294	KoreaRep * AcquirerSales	0.0095	KoreaRep * AcquirerEBITDA	0.0552
	(0.0504)		(0.0494)		(0.0542)
Malaysia*AcquirerAssets	-0.2115^{***}	Malaysia*AcquirerSales	-0.1429^{***}	Malaysia*AcquirerEBITDA	-0.2109***
	(0.0426)		(0.0421)		(0.0558)
Mexico * Acquirer Assets	-0.3658***	Mexico * AcquirerSales	-0.1508***	Mexico * Acquirer EBITDA	-0.0526
	(0.0316)		(0.0437)		(0.0550)
Netherlands*AcquirerAssets	-0.0799***	Netherlands*AcquirerSales	-0.0567^{***}	Netherlands*AcquirerEBITDA	-0.1291^{***}
	(0.0209)		(0.0215)		(0.0259)
New Zeal and * Acquirer Assets	-0.1727***	New Zeal and * Acquirer Sales	-0.1197***	New Zeal and * Acquirer EBITDA	-0.3288***
	(0.0307)		(0.0266)		(0.0381)
$Norway^*AcquirerAssets$	-0.1155***	Norway*AcquirerSales	-0.0915***	$Norway^*A cquirer EBITDA$	-0.1021***
	(0.0273)		(0.0262)		(0.0340)
$Poland^*AcquirerAssets$	-0.0356	Poland * AcquirerSales	-0.0602	Poland * Acquirer EBITDA	-0.0928*
	(0.0452)		(0.0441)		(0.0500)
RussianFederation*AcquirerAssets	-0.0841	RussianFederation * AcquirerSales	-0.1421**	RussianFederation*AcquirerEBITDA	-0.0242
	(0.0558)		(0.0607)		(0.0884)
$Singapore^*AcquirerAssets$	-0.1589***	Singapore * AcquirerSales	-0.1096***	$Singapore^*AcquirerEBITDA$	-0.1836***
	(0.0315)		(0.0268)		(0.0356)
SouthAfrica*AcquirerAssets	-0.1952^{***}	SouthAfrica*AcquirerSales	-0.1524^{***}	SouthAfrica*AcquirerEBITDA	-0.1421***
	(0.0376)		(0.0371)		(0.0543)
Spain*AcquirerAssets	-0.0371	Spain*AcquirerSales	-0.0328	Spain*AcquirerEBITDA	-0.0454
	(0.0317)		(0.0312)		(0.0358)
Switzerland * Acquirer Assets	-0.0841***	Switzerland * AcquirerSales	-0.0741***	Switzerland * Acquirer EBITDA	-0.0619*
	(0.0264)		(0.0285)		(0.0350)
$UnitedKingdom^*AcquirerAssets$	-0.0884***	$UnitedKingdom^*AcquirerSales$	-0.0762***	$UnitedKingdom^*AcquirerEBITDA$	-0.1113***
	(0.0174)		(0.0179)		(0.0214)
Australia * Acquirer ROA	-0.0370				. ,
-	(0.0295)				
$Belgium^*AcquirerROA$	0.0158***				
<i>.</i>	(0.0057)				
$Brazil^*AcquirerROA$	-0.0375				
• -	(0.0277)				
Canada*AcquirerBOA	-0.0413				
	(0.0390)				
China*AcquirerBOA	-0.0043				
	(0.0112)				
Denmark*AcauirerBOA	0.0424				
	(0.5075)				
Finland*AcavirerBOA	-0 1937				
	(0.1496)				
	(~·- +00)				

0.0021

(0.0058) 0.0108**

(0.0053)

 $France\,^*\!AcquirerROA$

 $Germany^*AcquirerROA$

Regression (3) of table 4	.6	Regression (4) of table 4.6	Regression (5) of table 4.6
HongKongSARChina*AcquirerROA	-0.0648*		
	(0.0355)		
India*AcquirerROA	-0.0484		
	(0.0374)		
Ireland * Acquirer ROA	-0.0134		
	(0.0453)		
$Italy^*AcquirerROA$	-0.0068		
	(0.0254)		
Japan*AcquirerROA	-0.0642*		
	(0.0346)		
KoreaRep*AcquirerROA	-0.0577*		
	(0.0337)		
Malaysia*AcquirerROA	0.0007		
	(0.0079)		
Mexico*AcquirerROA	-0.0010		
	(0.0081)		
Netherlands*AcquirerROA	-0.0154		
	(0.0531)		
NewZealand * AcquirerROA	0.0195		
	(0.0137)		
$Norway^*A cquirer ROA$	-0.0151		
	(0.0344)		
Poland*AcquirerROA	-0.0394		
	(0.0410)		
RussianFederation * AcquirerROA	-0.0564*		
	(0.0339)		
Singapore*AcquirerROA	-0.0539		
	(0.0349)		
SouthA frica * A cquirer ROA	0.0006		
	(0.0076)		
Spain*AcquirerROA	-0.0365		
	(0.0386)		
Switzer land * A cquirer ROA	0.0027		
	(0.0058)		
United Kingdom * A cquirer ROA	-0.0098		
	(0.0196)		

Note: Table reports supplemental results of regressions (3), (4) and (5) of table 4.6. In particular, the coefficient of the interaction between candidate target country fixed effects with acquirer-specific consolidated financial data (acquirer total assets, acquirer return on assets, acquirer net sales and acquirer earnings before interest, taxes, depreciation and amortization) are shown. In all regressions, the US represent the base category *, **, and *** denote statistical significance at 10%, 5%, and 1% levels, respectively. Robust standard errors are provided in parentheses.

C.3 Descriptives tables on the direction of M&As

Table C.8: Country examples for the four categories of ΔCFC_value									
ΔCFC_{-value}	Examplem: country	CEC mulea?	Tax rate	Passive-to-active-					
of country	Exemplary country	CrC rules:	threshold $>15\%$?	income ratio $<10\%$?					
0	Netherlands	no	n/a	n/a					
1	China (from 2008)	yes (since 2008)	no (12.5%)	no (50%)					
2	Korea, Rep.	yes	yes (15%)	no (50%)					
3	Japan	yes	yes (20%)	yes (no ratio)					
		1 0 1 1 1							

Table C.S. C. . a fo n the f . . .

Note: This table shows some examples for the calculated *CFC_value* in section 4.4 resulting from the various law differences.

Table C.9: Cross-border M&A sample (2002-2014) for analyzing the effect of CFC rules on direction of cross-border M&As (Section 4.4.4).

Country	CFC	Number of	Number of	Country	CFC	Number of	Number of
Country	rule	acquirers	targets		rule	acquirers	targets
Australia	1	43	57	Luxembourg	0	3	6
Austria	0	7	3	Mexico	1	7	5
Belgium	0	21	27	Netherlands	0	41	19
Brazil	1	3	24	New Zealand	1	4	4
Canada	1	70	101	Norway	1	9	24
Chile	0	2	6	Poland	0	1	5
China	1	14	6	Portugal	1	2	1
Denmark	1	7	9	Russian Federation	0	6	2
France	1	64	83	South Africa	1	20	10
Germany	1	55	65	Spain	1	29	40
India	0	32	12	Sweden	1	5	5
Ireland	0	32	14	Switzerland	0	40	18
Israel	1	21	16	United Kingdom	1	156	338
Italy	1	30	21	United States	1	411	260
Japan	1	55	9				
Korea, Rep.	1	9	9	Total		$1,\!199$	1,199

Note: This table shows the number of acquirer ultimate parents and target ultimate parents per country in our crossborder M&A sample to investigate Hypothesis 2. In this context, cross-border M&As are defined as acquirer ultimate parent and target ultimate parent residing in different countries; the direct acquirer and acquirer ultimate parent reside in the same country and also the direct target and target ultimate parent reside in the same country. CFC rule takes the value one if the country has implemented CFC rules in 2014. Each country has at least one acquiring firm and one target firm to ensure that maximum likelihood estimation yields finite likelihood.

Max	с С	1	26.823	1	1.000	1.000	4.314	1.000	0.998	11.742
Min	က္	-1	-26.706	-1	-0.990	-1.000	-22.413	-1.000	-0.997	-13.352
Std. Dev.	1.536	0.466	9.233	0.690	0.301	0.550	0.942	0.783	0.732	2.106
Mean	0.059	-0.069	1.149	-0.008	0.799	0.645	-0.082	0.104	0.089	0.037
Obs.	1,199	1,580	1,199	1,199	1,199	1,199	1,199	1,199	1,199	1,199
Data source	Tax guides	Tax guides	Tax guides	Tax guides	SDC Platinum; Compustat North America; Compustat Global	SDC Platinum	SDC Platinum; Compustat North America; Compustat Global	World Bank	World Bank	World Bank
Definition	Difference in CFC value of the two firms (see Section 4.4.2)	y Difference in CFC rule of the two firms (see Section 4.4.2)	Difference in STRs, including typical local taxes, of the two firms (in $\%$)	Difference in method to avoid double taxation on foreign dividends of two firms where 0 (1) represents the credit (exemption) method	Difference in total assets of the two firms divided by the sum of the firms' total assets	Difference in pre-tax incomes of the two firms divided by the sum of the firms' pre-tax incomes, where non-positive values of pre-tax income are replaced by 0.001 to avoid low values in the denominator	Difference in leverage ratios of the two firms (total liabilities/total assets, in $\%$)	Difference in stock market capitalizations of the two countries divided by the sum of the countries' stock market capitalization volume	Difference in domestic credits to private sector of the two countries divided by the sum of the countries' domestic credit volume	Difference in inflation rates of the two countries (in $\%$)
Variable	ΔCFC_value	ΔCFC_dumm	ΔSTR	ΔDTM	$\Delta Size$	ΔPTI	$\Delta Leverage$	$\Delta StockMrk$	$\Delta C reditMrk$	$\Delta In flation$

Bibliography

- Albertus, James (2018). "The Real Economic Effects of U.S. Tax Arbitrage by Foreign Multinational Firms". In: SSRN working paper. URL: https://ssrn.com/abstract= 2691065.
- Alexander, Anna, Antonio de Vito, and Martin Jacob (2016). "Corporate Tax Reforms and Tax-Motivated Profit Shifting: Evidence from the EU". In: FAccT Center Working Paper (Nr. 23). URL: https://ssrn.com/abstract=2841246.
- Almunia, Miguel and David Lopez-Rodriguez (2018). "Under the Radar: The Effects of Monitoring Firms on Tax Compliance". In: American Economic Journal: Economic Policy 10(1), pp. 1–38.
- Alstadsæter, Annette, Niels Johannesen, and Gabriel Zucman (2019). "Tax Evasion and Inequality". In: American Economic Review 109(6), pp. 2073–2103. DOI: 10.1257/aer. 20172043.
- Altshuler, Rosanne and R. Glenn Hubbard (2003). "The effect of the tax reform act of 1986 on the location of assets in financial services firms". In: *Journal of Public Economics* 87(1), pp. 109–127.
- Altshuler, Rosanne, Stephen Shay, and Eric Toder (2015). Lessons the United States Can Learn from Other Countries' Territorial Systems for Taxing Income of Multinational Corporations. Tax Policy Center - Urban Institute & Brookings Institution.
- Alvarez-Martinez, Maria et al. (2018). How Large is the Corporate Tax Base Erosion and Profit Shifting? A General Equilibrium Approach: CEPR Discussion Papers Nr. 12637.
- AmiNews, Amicorp (2017). The most recent CFC case in China. https://www.amicorp. com/AmiNews/2017/july/the_most_recent_cfc_case_in_china.php. (Visited on 08/18/2018).
- Andries, Kathleen, John Gallemore, and Martin Jacob (2017). "The effect of corporate taxation on bank transparency: Evidence from loan loss provisions". In: *Journal of Accounting* and Economics 63(2-3), pp. 307–328.

- Arulampalam, Wiji, Michael Devereux, and Federica Liberini (2019). "Taxes and the location of targets". In: *Journal of Public Economics* 176, pp. 161–178. DOI: 10.1016/j.jpubeco. 2019.05.005.
- Auerbach, Alan J., Michael Keen, and John Vella (2017). "Destination-Based Cash Flow Taxation". In: Oxford University Centre for Business Taxation WP 17(01).
- Auerbach, Alan and Michael Devereux (2015). Cash Flow Taxes in an International Setting. Saïd Business School. URL: http://ssrn.com/abstract=2556892.
- Avi-yonah, Reuven S. (2017). "Territoriality and the Original Intent of Subpart F". In: SSRN Electronic Journal.
- Avi-yonah, Reuven S. and Oz Halabi (2012). "US Subpart F Legislative Proposals: A Comparative Perspective". In: Law & Economics Working Papers, 69.
- Avi-yonah, Reuven S. and Gianluca Mazzoni (2017). "Tit for Tat: How Will Other Countries React to the Tax Cuts and Jobs Act?" In: *SSRN Electronic Journal*.
- Ayers, Benjamin C., Craig E. Lefanowicz, and John R. Robinson (2003). "Shareholder Taxes in Acquisition Premiums: The Effect of Capital Gains Taxation". In: *The Journal of Finance* 58(6), pp. 2783–2801.
- Ayers, Benjamin, Craig Lefanowicz, and John Robinson (2007). "Capital Gains Taxes and Acquisition Activity: Evidence of the Lock-in Effect". In: Contemporary Accounting Research 24(2), pp. 315–344.
- Badertscher, Brad A., Sharon P. Katz, and Sonja O. Rego (2013). "The separation of ownership and control and corporate tax avoidance". In: *Journal of Accounting and Economics* 56(2-3), pp. 228–250.
- Badertscher, Brad A., John D. Phillips, et al. (2009). "Earnings Management Strategies and the Trade–Off between Tax Benefits and Detection Risk: To Conform or Not to Conform?" In: *The Accounting Review* 84(1), pp. 63–97.
- Barrios, Salvador et al. (2012). "International taxation and multinational firm location decisions". In: *Journal of Public Economics* 96(11-12), pp. 946–958.
- Becker, Johannes and Clemens Fuest (2010). "Taxing foreign profits with international mergers and acquisitions". In: *International Economic Review* 51(1), pp. 171–186.
- Becker, Johannes and Nadine Riedel (2012). "Cross-border tax effects on affiliate investment-Evidence from European multinationals". In: *European Economic Review* 56(3), pp. 436– 450.
- Beer, Sebastian and Jan Loeprick (2015). "Profit shifting: drivers of transfer (mis)pricing and the potential of countermeasures". In: International Tax and Public Finance 22(3), pp. 426–451.

- Beuselinck, Christof, Marc Deloof, and Ann Vanstraelen (2015). "Cross-jurisdictional income shifting and tax enforcement: Evidence from public versus private multinationals". In: *Review of Accounting Studies* 20(2), pp. 710–746.
- Blinder, Alan S. (2006). "Offshoring: The Next Industrial Revolution?" In: Foreign Affairs 85(2), pp. 113–128.
- Blomquist, Sören and Whitney K. Newey (2017). "The bunching estimator cannot identify the taxable income elasticity". In: *Working Paper*, pp. 1–82. URL: http://economics. mit.edu/files/13770.
- Blouin, Jennifer et al. (2014). Thin capitalization rules and multinational firm capital structure. Vol. N.42 - 2014. Taxation papers. Publications Office: Luxembourg. ISBN: 9279354299.
- Bräutigam, Rainer, Christoph Spengel, and Frank Streif (2017). "Decline of Controlled Foreign Company Rules and Rise of Intellectual Property Boxes: How the European Court of Justice Affects Tax Competition and Economic Distortions in Europe". In: *Fiscal Studies* 38(4), pp. 719–745.
- Brown, Jennifer L. and Katharine D. Drake (2013). "Network Ties Among Low-Tax Firms". In: *The Accounting Review* 89(2), pp. 483–510.
- Buettner, Thiess, Michael Overesch, Ulrich Schreiber, et al. (2012). "The impact of thincapitalization rules on the capital structure of multinational firms". In: *Journal of Public Economics* 96(11-12), pp. 930–938.
- Buettner, Thiess, Michael Overesch, and Georg Wamser (2016). "Restricted interest deductibility and multinationals' use of internal debt finance". In: *International Tax and Public Finance* 23(5), pp. 785–797.
- Buettner, Thiess, Michael Overesch, and Georg Wamser (2018). "Anti profit-shifting rules and foreign direct investment". In: *International Tax and Public Finance* 25(3), pp. 553– 580.
- Buettner, Thiess and Martin Ruf (2007). "Tax incentives and the location of FDI: Evidence from a panel of German multinationals". In: International Tax and Public Finance 14(2), pp. 151–164.
- Burgstahler, David and Ilia Dichev (1997). "Earnings management to avoid earnings decreases and losses". In: *Journal of Accounting and Economics* 24(1), pp. 99–126.
- Castro, Leonardo F. M. (2013). "Brazilian CFC Rules and Indirect Controlled Foreign Companies: Recent Case Laws and Consequences of Supreme Court Precedent". In: International Tax Journal 39(3), pp. 23–47.

- Chetty, Raj et al. (2011). "Adjustment Costs, Firm Responses, and Micro vs. Macro Labor Supply Elasticities: Evidence from Danish Tax Records". In: *The Quarterly Journal of Economics* 126(2), pp. 749–804.
- Clausing, Kimberly (2015). "Beyond Territorial and Worldwide Systems of International Taxation". In: Journal of International Finance and Economics 15(2), pp. 43–58. DOI: 10.18374/jife-15-2.5.
- Clifford, Sarah (2017). "Taxing multinationals beyond borders: financial and locational responses to CFC rules". In: *CBT Oxford Working Paper* WP 17/18.
- Collins, Julie H. and Douglas A. Shackelford (1997). "Global organizations and taxes: An analysis of the dividend, interest, royalty, and management fee payments between U.S. multinationals' foreign affiliates". In: *Journal of Accounting and Economics* 24(2), pp. 151–173.
- Cristea, Anca D. and Daniel X. Nguyen (2016). "Transfer Pricing by Multinational Firms: New Evidence from Foreign Firm Ownerships". In: American Economic Journal: Economic Policy 8(3), pp. 170–202.
- Davies, Ronald B. et al. (2018). "Knocking on Tax Haven's Door: Multinational Firms and Transfer Pricing". In: *The Review of Economics and Statistics*.
- Deloitte (2015). Guide to Controlled Foreign Company Regimes. https://www2.deloitte. com/al/en/pages/tax/articles/guide-to-controlled-foreign-companyregimes.html. (Visited on 06/08/2018).
- Desai, Mihir A., C. Fritz Foley, and James R. Hines (2006). "The demand for tax haven operations". In: *Journal of Public Economics* 90(3), pp. 513–531.
- Devereux, Michael and Rachel Griffith (2003). "Evaluating Tax Policy for Location Decisions". In: International Tax and Public Finance 10(2), pp. 107–126.
- Devereux, Michael, Li Liu, and Simon Loretz (2014). "The elasticity of corporate taxable income: New evidence from UK tax records". In: American Economic Journal: Economic Policy 6(2), pp. 19–53.
- Dharmapala, Dhammika (2014). "What Do We Know about Base Erosion and Profit Shifting? A Review of the Empirical Literature". In: *Fiscal Studies* 35(4), pp. 421–448.
- Dharmapala, Dhammika and Nadine Riedel (2013). "Earnings shocks and tax-motivated income-shifting: Evidence from European multinationals". In: *Journal of Public Economics* 97, pp. 95–107.
- Di Giovanni, Julian (2005). "What drives capital flows? The case of cross-border M&A activity and financial deepening". In: *Journal of International Economics* 65(1), pp. 127–149.

- DLA piper (2018). Canada announces new international tax measures: Global Tax Alert. https://www.dlapiper.com/en/us/insights/publications/2018/03/2018federal-budget-international-tax-measures/. (Visited on 08/04/2018).
- Dowd, Tim, Paul Landefeld, and Anne Moore (2017). "Profit shifting of U.S. multinationals".In: Journal of Public Economics 148(Supplement C), pp. 1–13.
- Dyreng, Scott D., Michelle Hanlon, et al. (2017). "Changes in corporate effective tax rates over the past 25 years". In: *Journal of Financial Economics* 124(3), pp. 441–463.
- Dyreng, Scott D., Bradley P. Lindsey, et al. (2015). "The effect of tax and nontax country characteristics on the global equity supply chains of U.S. multinationals". In: *Journal of Accounting and Economics* 59(2), pp. 182–202.
- Edwards, Alexander, Todd Kravet, and Ryan Wilson (2016). "Trapped Cash and the Profitability of Foreign Acquisitions". In: *Contemporary Accounting Research* 33(1), pp. 44– 77.
- Egger, Peter, Valeria Merlo, et al. (2015). "Consequences of the New UK Tax Exemption System: Evidence from Micro-level Data". In: *The Economic Journal* 125(589), pp. 1764– 1789.
- Egger, Peter and Georg Wamser (2015). "The impact of controlled foreign company legislation on real investments abroad. A multi-dimensional regression discontinuity design". In: Journal of Public Economics 129, pp. 77–91.
- Ernst & Young (2004-2016). Worldwide Corporate Tax Guides. http://www.ey.com/gl/en/services/tax/global-tax-guide-archive. (Visited on 06/08/2018).
- European Court of Justice (2006). Cadbury Schweppes plc & Cadbury Schweppes Overseas Ltd v Commissioners of Inland Revenue: Cadbury-Schweppes Case C-196/04.
- EY (2018a). Canada Corporate Group taxation. http://taxsummaries.pwc.com/ID/ Canada-Corporate-Group-taxation. (Visited on 08/04/2018).
- EY (2018b). Tax Alert Canada Federal budget 2018-19: Equality + Growth A Strong Middle Class. https://www.ey.com/Publication/vwLUAssets/Tax_Alert_2018_No_07/ \$FILE/TaxAlert2018No07.pdf. (Visited on 08/04/2018).
- Faulkender, Michael and Jason M. Smith (2016). "Taxes and leverage at multinational corporations". In: Journal of Financial Economics 122(1), pp. 1–20.
- Feld, Lars and Jost Heckemeyer (2011). "Fdi and taxation: A meta-study". In: Journal of Economic Surveys 25(2), pp. 233–272.
- Feld, Lars, Martin Ruf, et al. (2016). "Repatriation taxes and outbound M&As". In: Journal of Public Economics 139, pp. 13–27.
- Field, Heather M. (2008-2009). "Checking in on Check-the-Box". In: Loyola of Los Angeles Law Review 42(2), pp. 451–528.

- Fuest, Clemens et al. (2013). "Profit Shifting and 'Aggressive' Tax Planning by Multinational Firms: Issues and Options for Reform". In: World Tax Journal 5(3), pp. 307–324.
- Goldman, Lisa (2018). First Look at the Tax Cuts and Jobs Act The CPA Journal. https: //www.cpajournal.com/2018/05/16/first-look-at-the-tax-cuts-and-jobs-act-3/. (Visited on 08/13/2018).
- Gordon, Roger H. (1986). "Taxation of Investment and Savings in a World Economy". In: *The American Economic Review* 76(5), pp. 1086–1102.
- Gravelle, Jane G. (2009). "Tax Havens: International Tax Avoidance and Evasion". In: *National Tax Journal* 62(4), pp. 727–753.
- Griffith, Rachel, Helen Miller, and Martin O'Connell (2014). "Ownership of intellectual property and corporate taxation". In: *Journal of Public Economics* 112, pp. 12–23.
- Grubert, Harry (2012). "Foreign Taxes and the Growing Share of U.S. Multinational Company Income Abroad: Profits, not Sales, are being globalized". In: National Tax Journal 65(2), pp. 247–282.
- Grubert, Harry and Rosanne Altshuler (2006). Corporate Taxes In The World Economy -Reforming the taxation of cross-border income. URL: ftp://www-snde.rutgers.edu/ Rutgers/wp/2006-26.pdf.
- Grubert, Harry and John Mutti (1995). "Taxing multinationals in a world with portfolio flows and R&D: Is capital export neutrality obsolete?" In: International Tax and Public Finance 2(3), pp. 439–457.
- Guenther, David A., Steven R. Matsunaga, and Brian M. Williams (2017). "Is Tax Avoidance Related to Firm Risk?" In: *The Accounting Review* 92(1), pp. 115–136.
- Gumpert, Anna, James R. Hines, and Monika Schnitzer (2016). "Multinational firms and tax havens". In: *Review of Economics and Statistics* 98(4), pp. 713–727.
- Haberly, D. and D. Wojcik (2015). "Tax havens and the production of offshore FDI: an empirical analysis". In: *Journal of Economic Geography* 15, pp. 75–101.
- Hagen, Dominik von and Fabian Nicolas Pönnighaus (2017). "International Taxation and M&A Prices". In: SSRN Working Paper.
- Hanlon, Michelle, Rebecca Lester, and Rodrigo Verdi (2015). "The effect of repatriation tax costs on U.S. multinational investment". In: *Journal of Financial Economics* 116(1), pp. 179–196.
- Haufler, Andreas, Mohammed Mardan, and Dirk Schindler (2018). "Double tax discrimination to attract FDI and fight profit shifting: The role of CFC rules". In: Journal of International Economics 114, pp. 25–43.

- Haufler, Andreas and Marco Runkel (2012). "Firms' financial choices and thin capitalization rules under corporate tax competition". In: *European Economic Review* 56(6), pp. 1087– 1103.
- Hebous, Shafik, Martin Ruf, and Alfons Weichenrieder (2011). "The Effects of Taxation on the Location Decision of Multinational Firms: M&A vs. Greenfield Investments". In: *National Tax Journal* 64(3), pp. 817–838.
- Heckemeyer, Jost H. and Michael Overesch (2017). "Multinationals' profit response to tax differentials: Effect size and shifting channels". In: *Canadian Journal of Economics* 50(4), pp. 965–994.
- Helpman, Elhanan (2014). "Foreign trade and investment: Firm-level perspectives". In: Economica 81(321), pp. 1–14.
- Hentschel, Sven and Till Moser (2017). "The Provisions of the EU Anti-Tax Avoidance Directive regarding Controlled Foreign Company Rules: A Critical Review Based on the Experience with the German CFC Legislation". In: Intertax 45(10), pp. 606–623.
- Herger, Nils, Christos Kotsogiannis, and Steve McCorriston (2016). "Multiple taxes and alternative forms of FDI: Evidence from cross-border acquisitions". In: *International Tax* and Public Finance 23(1), pp. 82–113.
- Hines, James R. (1994). "Credit and Deferral as International Investment Incentive". In: Journal of policy analysis and management : [the journal of the Association for Public Policy Analysis and Management] 55, pp. 323–347.
- Hines, James R. (1999). "Lessons from Behavioral Responses to International Taxation". In: National Tax Journal 52, pp. 305–322.
- Huizinga, Harry and Luc Laeven (2008). "International profit shifting within multinationals: A multi-country perspective". In: *Journal of Public Economics* 92(5-6), pp. 1164–1182.
- Huizinga, Harry, Luc Laeven, and Gaetan Nicodeme (2008). "Capital structure and international debt shifting". In: *Journal of Financial Economics* 88(1), pp. 80–118.
- Huizinga, Harry and Johannes Voget (2009). "International Taxation and the Direction and Volume of Cross-Border M&As". In: *The Journal of Finance* 64(3), pp. 1217–1249.
- Huizinga, Harry, Johannes Voget, and Wolf Wagner (2012). "Who bears the burden of international taxation? Evidence from cross-border M&As". In: Journal of International Economics 88(1), pp. 186–197.
- Huizinga, Harry, Johannes Voget, and Wolf Wagner (2018). "Capital gains taxation and the cost of capital: Evidence from unanticipated cross-border transfers of tax base". In: *Journal of Financial Economics* 129(2), pp. 306–328. DOI: 10.1016/j.jfineco.2018. 04.014.

- IBFD, ed. (2002-2016). *European Tax Handbook*. European Tax Handbook. International Bureau of Fiscal Documentation: Amsterdam (Netherlands).
- Jalan, Akanksha and R. Vaidyanathan (2017). "Tax havens: conduits for corporate tax malfeasance". In: Journal of Financial Regulation and Compliance 25(1), pp. 86–104.
- Kane, Mitchell (2013). "Milking Versus Parking: Transfer Pricing and CFC Rules under the Code". In: Tax Law Review 1(66), pp. 487–506.
- Keller, Sara and Deborah Schanz (2013). "Measuring tax attractiveness across countries". In: Arqus-Diskussionsbeiträge zur quantitativen Steuerlehre 143(143), pp. 1–58.
- Khan, Mozaffar, Suraj Srinivasan, and Liang Tan (2017). "Institutional Ownership and Corporate Tax Avoidance: New Evidence". In: *The Accounting Review* 92(2), pp. 101–122.
- Kleven, Henrik J. (2016). "Bunching". In: Annual Review of Economics 8(1), pp. 435–464.
- Kleven, Henrik J. and Mazhar Waseem (2013). "Using Notches to Uncover Optimization Frictions and Structural Elasticities: Theory and Evidence from Pakistan". In: *The Quarterly Journal of Economics* 128(2), pp. 669–723.
- Koerver, Schmidt Peter (2016). "Taxation of Controlled Foreign Companies in Context of the OECD/G20 Project on Base Erosion and Profit Shifting as well as the EU Proposal for the Anti-Tax Avoidance Directive – An Interim Nordic Assessment". In: Nordic Tax Journal(2), pp. 87–112.
- KPMG (2003-2018). Corporate tax rates table. https://home.kpmg.com/xx/en/home/ services/tax/tax-tools-and-resources/tax-rates-online/corporate-taxrates-table.html. (Visited on 06/08/2018).
- Krautheim, Sebastian and Tim Schmidt-Eisenlohr (2011). "Heterogeneous firms, 'profit shifting' FDI and international tax competition". In: *Journal of Public Economics* 95(1-2), pp. 122–133.
- Kuźniacki, Błażej (2015). "The Need to Avoid Double Economic Taxation Triggered by CFC Rules under Tax Treaties, and the Way to Achieve It". In: *Intertax* 43(12), pp. 758–772.
- Laffitte, Sébastien and Farid Toubal (2018). "Firms, Trade and Profit Shifting: Evidence from Aggregate Data". In: *CESifo Working Paper*(7171).
- Langenmayr, Dominika and Rebecca Lester (2018). "Taxation and Corporate Risk-Taking". In: The Accounting Review 93(3), pp. 237–266.
- Lewellen, Katharina and Leslie Robinson (2013). "Internal Ownership Structures of U.S. Multinational Firms". In: SSRN Electronic Journal.
- Lohse, Theresa and Nadine Riedel (2013). "Do Transfer Pricing Laws Limit International Income Shifting? Evidence from European Multinationals". In: CESifo Working Paper Series (4404).

- Loudermilk, Margaret S. (2007). "Estimation of Fractional Dependent Variables in Dynamic Panel Data Models with an Application to Firm Dividend Policy". In: Journal of Business & Economic Statistics 25(4), pp. 462–472.
- Mardan, Mohammed (2017). "Why countries differ in thin capitalization rules: The role of financial development". In: *European Economic Review* 91, pp. 1–14.
- Markle, Kevin and Leslie Robinson (2012). "Tax Haven Use Across International Tax Regimes". In: University of Iowa and Dartmouth College Working Paper. URL: http://faculty.tuck.dartmouth.edu/images/uploads/faculty/leslierobinson/marklerobinson.pdf.
- Marques, Mário and Carlos Pinho (2014). "Tax-Treaty Effects on Foreign Investment: Evidence from European Multinationals". In: *FinanzArchiv: Public Finance Analysis* 70(4), pp. 527–555.
- Marques, Mário and Carlos Pinho (2016). "Is transfer pricing strictness deterring profit shifting within multinationals? Empirical evidence from Europe". In: Accounting and Business Research 46(7), pp. 703–730.
- Maydew, Edward L. (1997). "Tax-Induced Earnings Management by Firms with Net Operating Losses". In: Journal of Accounting Research 35(1), p. 83.
- Mazars (2015). Blog: BEPS- CFC Let's Talk Tax. http://blogs.mazars.com/ letstalktax/2015/04/beps-cfc/. (Visited on 08/01/2018).
- McClure, Ross et al. (2018). "The impact of dividend imputation on corporate tax avoidance: The case of shareholder value". In: *Journal of Corporate Finance* 48, pp. 492–514.
- McGee, Robert W. (2012). The Ethics of Tax Evasion: Perspectives in Theory and Practice. Springer Science+Business Media, LLC: New York, NY.
- McGuire, Sean T., Dechun Wang, and Ryan J. Wilson (2014). "Dual Class Ownership and Tax Avoidance". In: *The Accounting Review* 89(4), pp. 1487–1516.
- Merlo, Valeria, Nadine Riedel, and Georg Wamser (2015). "The Impact of Thin Capitalization Rules on the Location of Multinational Firms' Foreign Affiliates". In: CESifo Working Paper(5449).
- Merz, Julia, Michael Overesch, and Georg Wamser (2017). "The location of financial sector FDI: Tax and regulation policy". In: *Journal of Banking & Finance* 78(Supplement C), pp. 14–26.
- Mitchell, Mark L. and J.Harold Mulherin (1996). "The impact of industry shocks on takeover and restructuring activity". In: *Journal of Financial Economics* 41(2), pp. 193–229.
- Musgrave, Peggy B. (1969). United States taxation of foreign investment income: Issues and arguments. Harvard Univ Harvard Law School.

- Mutti, John and Harry Grubert (2004). "Empirical asymmetries in foreign direct investment and taxation". In: *Journal of International Economics* 62(2), pp. 337–358.
- Mutti, John and Harry Grubert (2009). "The Effect of Taxes on Royalties and the Migration of Intangible Assets Abroad". In: International Trade in Services and Intangibles in the Era of Globalization. University of Chicago Press, pp. 111–137.
- Nessa, Michelle L. (2017). "Repatriation Tax Costs and U.S. Multinational Companies" Shareholder Payouts". In: *The Accounting Review* 92(4), pp. 217–241.
- Nicodème, Gaëtan (2009). Corporate income tax and economic distortions. Vol. 15. Taxation papers. Working paper. EUR-OP: Luxembourg. ISBN: 978-92-79-12058-9.
- OECD (2013). Addressing Base Erosion and Profit Shifting. Ed. by OECD. OECD Publishing. ISBN: 9789264192744.
- OECD (2015a). Designing Effective Controlled Foreign Company Rules, Action 3 2015 Final Report. OECD Publishing. ISBN: 9789264241145.
- OECD (2015b). Public Consultation: Strengthening CFC Rules, 12. May 2015. http://www. oecd.org/tax/aggressive/public-consultation-beps-action-3-strengtheningcfc-rules.htm. OECD Conference Centre, Paris. (Visited on 08/01/2018).
- Oguttu, Annet Wanyana (2009). "Resolving the conflict between 'controlled foreign company' legislation and tax treaties: A South African perspective". In: *The Comparative and International Law Journal of Southern Africa*.
- Ohrn, Eric and Nathan Seegert (2019). "The impact of investor-level taxation on mergers and acquisitions". In: *Journal of Public Economics* 177, p. 104038. DOI: 10.1016/j. jpubeco.2019.06.006.
- Overesch, Michael and Johannes Rincke (2011). "What Drives Corporate Tax Rates Down? A Reassessment of Globalization, Tax Competition, and Dynamic Adjustment to Shocks*". In: Scandinavian Journal of Economics 18, no-no.
- Picciotto, Sol (2017). The G20 and the Base Erosion and Profit Shifting (BEPS) Project. Vol. 18/2017. Discussion paper / Deutsches Institut f
 ür Entwicklungspolitik. Bonn. ISBN: 3960210418.
- Piketty, Thomas (2014). Capital in the twenty first century: Thomas Piketty ; translated by Arthur Goldhammer. London: Cambridge (Massachusetts). ISBN: 067443000X.
- Pinto, Dale (2009). "A proposal to reform income anti-tax-deferal regimes". In: Journal of Australian Taxation 12, pp. 41–75.
- PwC (2015). How the amended CFC rules will work. https://www.pwc.ru/en/taxconsulting-services/assets/legislation/tax-flash-report-2015-20-eng.pdf.

- Qio, Dongmei (Doreen) (2017). Controlled Foreign Company (CFC) rules and the latest development in the PRC; Xiamen University. https://www.law.hku.hk/aiifl/wp-content/uploads/ppt/TLRP-DoreenQiu-ppt.pdf. (Visited on 08/18/2018).
- Redmiles, Melissa and Jason Wenrich (2007). "A History of Controlled Foreign Corporations and the Foreign Tax Credit". In: *Statistics of Income. SOI Bulletin* 27(1), pp. 129–135.
- Rego, Sonja Olhoft (2003). "Tax-Avoidance Activities of U.S. Multinational Corporations".In: Contemporary Accounting Research 20(4), pp. 805–833.
- Richman, Peggy Brewer (1963). Taxation of foreign investment income: An economic analysis. Johns Hopkins Press.
- Riedel, Nadine (2018). "Quantifying International Tax Avoidance: A Review of the Academic Literature". In: *Review of Economics* 69(2), pp. 169–181.
- Ruf, Martin and Dirk Schindler (2015). "Debt Shifting and Thin-Capitalization Rules German Experience and Alternative Approaches". In: Nordic Tax Journal (1), pp. 17– 33.
- Ruf, Martin and Alfons Weichenrieder (2012). "The Taxation of Passive Foreign Investment: lessons from German experience". In: *Canadian Journal of Economics* 45(4), pp. 1504– 1528.
- Ruf, Martin and Alfons Weichenrieder (2013). "CFC Legislation, Passive Assets and the Impact of the ECJ's Cadbury-Schweppes Decision". In: *CESifo Working Paper*(4461).
- Rust, Alexander (2008). "CFC Legislation and EC Law". In: Intertax 36(11), pp. 492–501.
- Saez, Emmanuel (2010). "Do Taxpayers Bunch at Kink Points?" In: American Economic Journal: Economic Policy 2(3), pp. 180–212.
- Saunders-Scott, Molly J. (2015). "Substitution Across Methods of Profit Shifting". In: National Tax Journal 68(4), pp. 1099–1120.
- Schäfer, Anne and Christoph Spengel (2010). "International Tax Planning in the Age of ICT". In: ZEW Discussion Paper 04-27.
- Schneider, Michaela (2004). "Characteristics of foreign- controlled enterprises in focus". In: Statistics in focus 4(21), pp. 1–12.
- Simone, Lisa de, Kenneth J. Klassen, and Jeri K. Seidman (2017). "Unprofitable Affiliates and Income Shifting Behavior". In: *The Accounting Review* 92(3), pp. 113–136.
- Simone, Lisa de and Richard C. Sansing (2018). "Income Shifting Using a Cost Sharing Arrangement". In: *The Journal of the American Taxation Association*.
- Slemrod, Joel (1990). "Optimal Taxation and Optimal Tax Systems". In: Journal of Economic Perspectives 4(1), pp. 157–178.
- Slemrod, Joel (2007). "Cheating Ourselves: The Economics of Tax Evasion". In: Journal of Economic Perspectives 21(1), pp. 25–48.

- Slemrod, Joel (2013). "Buenas notches: lines and notches in tax system design". In: *eJournal of Tax Research* 11(3), pp. 259–283.
- The Economist (2013). The price isn't right. https://www.economist.com/specialreport/2013/02/16/the-price-isnt-right. (Visited on 09/12/2018).
- Todtenhaupt, Maximilian et al. (2020). "Taxing away M&A: Capital gains taxation and acquisition activity". In: *European Economic Review* 128, p. 103505. DOI: 10.1016/j.euroecorev.2020.103505.
- UNCTAD (2017). "World Investment Report Investment and the Digital Economy". In: United Nations Publications.
- Voget, Johannes (2011). "Relocation of headquarters and international taxation". In: Journal of Public Economics 95(9-10), pp. 1067–1081.
- Wamser, Georg (2014). "The Impact of Thin-Capitalization Rules on External Debt Usage A Propensity Score Matching Approach". In: Oxford Bulletin of Economics and Statistics 76(5), pp. 764–781.
- Weichenrieder, Alfons (1996). "Anti-tax-avoidance provisions and the size of foreign direct investment". In: *International Tax and Public Finance* 3(1), pp. 67–81.
- Weichenrieder, Alfons (2009). "Profit shifting in the EU: Evidence from Germany". In: *International Tax and Public Finance* 16(3), pp. 281–297.
- White, James (2013). Corporate Income Tax Effective Tax Rates Can Differ Significantly from the Statutory Rate. GAO-13-520 Washington, DC.
- Zucman, Gabriel (2015). The Hidden Wealth of Nations: The Scourge of Tax Havens. University of Chicago Press. ISBN: 9780226245423.
- Zwick, Eric and James Mahon (2017). "Tax Policy and Heterogeneous Investment Behavior".
 In: American Economic Review 107(1), pp. 217–248. (Visited on 11/01/2017).